## Final Proposed Filing - Coversheet

#### **Instructions:**

In accordance with Title 3 Chapter 25 of the Vermont Statutes Annotated and the "Rule on Rulemaking" adopted by the Office of the Secretary of State, this filing will be considered complete upon filing and acceptance of these forms with the Office of the Secretary of State, and the Legislative Committee on Administrative Rules.

All forms shall be submitted at the Office of the Secretary of State, no later than 3:30 pm on the last scheduled day of the work week.

The data provided in text areas of these forms will be used to generate a notice of rulemaking in the portal of "Proposed Rule Postings" online, and the newspapers of record if the rule is marked for publication. Publication of notices will be charged back to the promulgating agency.

# PLEASE REMOVE ANY COVERSHEET OR FORM NOT REQUIRED WITH THE CURRENT FILING BEFORE DELIVERY!

Certification Statement: As the adopting Authority of this rule (see 3 V.S.A. § 801 (b) (11) for a definition), I approve the contents of this filing entitled:

Nursing Home Reimbursement Rule Simplificat	cion			
/s/ Todd W. Daloz	, on 3/15/24			
(signature)	(date)			
Printed Name and Title: Todd Daloz				
Deputy Secretary, Agency of Human Services				
	RECEIVED BY:			
<ul> <li>□ Coversheet</li> <li>□ Adopting Page</li> </ul>				
☐ Economic Impact Analysis				
☐ Environmental Impact Analysis				
☐ Strategy for Maximizing Public Input				
Scientific Information Statement (if applicable)				
	Incorporated by Reference Statement (if applicable)			
☐ Clean text of the rule (Amended text without annotation)				
<ul> <li>□ Annotated text (Clearly marking changes from previous rule)</li> <li>□ ICAR Minutes</li> </ul>				
☐ Copy of Comments				

☐ Responsiveness Summary



OFFICE OF THE SECRETARY TEL: (802) 241-0440 FAX: (802) 241-0450

JENNEY SAMUELSON SECRETARY

TODD W. DALOZ DEPUTY SECRETARY

# STATE OF VERMONT AGENCY OF HUMAN SERVICES

#### **MEMORANDUM**

TO: Sarah Copeland Hanzas, Secretary of State

FROM: Jenney Samuelson, Secretary, Agency of Human Services

**DATE:** January 31, 2023

**SUBJECT:** Signatory Authority for Purposes of Authorizing Administrative Rules

I hereby designate Deputy Secretary of Human Services Todd W. Daloz as signatory to fulfill the duties of the Secretary of the Agency of Human Services as the adopting authority for administrative rules as required by Vermont's Administrative Procedure Act, 3. V.S.A § 801 et seq.

Cc: Todd W. Daloz

#### 1. TITLE OF RULE FILING:

#### Nursing Home Reimbursement Rule Simplification

# 2. PROPOSED NUMBER ASSIGNED BY THE SECRETARY OF STATE 24P006

#### 3. ADOPTING AGENCY:

Agency of Human Services

#### 4. PRIMARY CONTACT PERSON:

(A PERSON WHO IS ABLE TO ANSWER QUESTIONS ABOUT THE CONTENT OF THE RULE).

Name: James LaRock

Agency: Department of Vermont Health Access

Mailing Address: NOB 1 South, 280 State Drive, Waterbury,

VT 05671

Telephone: 802-241-0251 Fax: 802-241-0260

E-Mail: james.larock@vermont.gov

Web URL (WHERE THE RULE WILL BE POSTED):

https://humanservices.vermont.gov/rules-

policies/health-care-rules/health-care-

administrative-rules-hcar

#### 5. SECONDARY CONTACT PERSON:

(A SPECIFIC PERSON FROM WHOM COPIES OF FILINGS MAY BE REQUESTED OR WHO MAY ANSWER QUESTIONS ABOUT FORMS SUBMITTED FOR FILING IF DIFFERENT FROM THE PRIMARY CONTACT PERSON).

Name: Jaime Mooney

Agency: Department of Vermont Health Access

Mailing Address: NOB 1 South, 280 State Drive, Waterbury,

VT 05671

Telephone: 802-798-2144 Fax: 802-241-0260

E-Mail: jaime.mooney@vermont.gov

#### 6. RECORDS EXEMPTION INCLUDED WITHIN RULE:

(DOES THE RULE CONTAIN ANY PROVISION DESIGNATING INFORMATION AS CONFIDENTIAL; LIMITING ITS PUBLIC RELEASE; OR OTHERWISE, EXEMPTING IT FROM INSPECTION AND COPYING?) Yes

#### IF YES, CITE THE STATUTORY AUTHORITY FOR THE EXEMPTION:

33 V.S.A. § 908(a)

#### PLEASE SUMMARIZE THE REASON FOR THE EXEMPTION:

Salaries and wages of individual employees are nonpublic information under state statute.

#### 7. LEGAL AUTHORITY / ENABLING LEGISLATION:

(The specific statutory or legal citation from session law indicating who the adopting Entity is and thus who the signatory should be. THIS SHOULD BE A SPECIFIC CITATION NOT A CHAPTER CITATION).

- 33 V.S.A. § 1901(a), 33 V.S.A. § 904(a)
- 8. EXPLANATION OF HOW THE RULE IS WITHIN THE AUTHORITY OF THE AGENCY:
  - 33 V.S.A. § 1901(a) allows the Secretary to adopt rules needed to administer the Medicaid program. 33 V.S.A. § 904(a) requires the Director of the Division of Rate Setting to adopt rules establishing "procedures for determining payment rates" for nursing home care. The Division's rules establish a process for applying for, receiving, and appealing per diem payment rates for nursing home providers and coordinate the expenditures of the public payers responsible for reimbursing these providers.
- 9. THE FILING HAS NOT CHANGED SINCE THE FILING OF THE PROPOSED RULE.
- 10. THE AGENCY HAS NOT INCLUDED WITH THIS FILING A LETTER EXPLAINING IN DETAIL WHAT CHANGES WERE MADE, CITING CHAPTER AND SECTION WHERE APPLICABLE.
- 11. SUBSTANTIAL ARGUMENTS AND CONSIDERATIONS WERE NOT RAISED FOR OR AGAINST THE ORIGINAL PROPOSAL.
- 12. THE AGENCY HAS INCLUDED COPIES OF ALL WRITTEN SUBMISSIONS AND SYNOPSES OF ORAL COMMENTS RECEIVED.
- 13. THE AGENCY HAS INCLUDED A LETTER EXPLAINING IN DETAIL THE REASONS FOR THE AGENCY'S DECISION TO REJECT OR ADOPT THEM.
- 14. CONCISE SUMMARY (150 words or Less):

These rules strike all existing Division of Rate Setting rules and replace them in the Agency of Human Services's Health Care Administrative Rules. Some material, particularly language regarding which costs are allowable and how the Division applies various bonuses or penalties, is moved into a new manual. The remaining material primarily sets out the administrative process for applying for, receiving, and appealing per diem rates set by the Division.

#### 15. EXPLANATION OF WHY THE RULE IS NECESSARY:

Because the existing rate-setting process is prescribed by rule, it takes several months to amend the finer details of the process in response to changing conditions in the industry or to push increased appropriations to providers. Further, Vermont Medicaid has adopted a notice and comment process for making changes to its reimbursement policies (the Global Commitment Registry), which is redundant with the rulemaking process. Moving rate-setting details and processes into a new manual will align the nursing home rate-setting process with other Vermont Medicaid rate-setting processes. The rule retains necessary material regarding applying for, receiving, and appealing per diem rates and setting out the broad principles of the rate setting process.

#### 16. EXPLANATION OF HOW THE RULE IS NOT ARBITRARY:

Factual basis for the rule: Vermont Medicaid is currently required to undergo the rulemaking process to make almost any changes to the nursing home ratesetting process, regardless of their fiscal impact or the scale of the change.

Rational connection to factual basis: This rule would allow Vermont Medicaid to increase or change reimbursements to nursing home providers without undergoing the full rulemaking process. Fundamental changes to the process would still require rulemaking. Changes with a fiscal impact would still require approval by the Legislature during the budget process, but would not require approval by the Legislature in the budget as well as in the rulemaking process.

Reasonableness: Vermont Medicaid's experience with nearly all other provider types, which have rates set by Vermont Medicaid manuals and policy decisions, shows that such a system is a reasonable way to calculate and set provider rates.

# 17. LIST OF PEOPLE, ENTERPRISES AND GOVERNMENT ENTITIES AFFECTED BY THIS RULE:

- -Nursing home residents
- -Nursing home providers, including their trade association, the Vermont Health Care Association

- -Other providers of residential care or facilities for the aging and disabled, who may be indirectly impacted by Vermont Medicaid reimbursement policy for nursing homes
- -The Department of Disabilities, Aging, and Independent Living, which manages the Choices for Care budget and licenses and inspects nursing homes
- -The Division of Rate Setting, which calculates and adjusts each nursing home's Medicaid rate and recommends changes to the rate-setting system.
- 18. BRIEF SUMMARY OF ECONOMIC IMPACT (150 words or Less):

The proposed change will not have a fiscal impact. Vermont Medicaid is simultaneously proposing a new manual that increases caps on cost growth, lowers minimum occupancy requirements, and transitions to a new resident classification system that will moderately impact the Vermotn Medicaid Choices for Care budget managed by the Department of Disabilities, Aging, and Independent Living. The state is seeking legislative approval for these impacts.

19. A HEARING WAS NOT HELD.

#### 20. HEARING INFORMATION

(THE FIRST HEARING SHALL BE NO SOONER THAN 30 DAYS FOLLOWING THE POSTING OF NOTICES ONLINE).

IF THIS FORM IS INSUI	FFICIENT TO LIST THE INFORMATION FOR EACH HEARING, PLEAS
ATTACH A SEPARATE	SHEET TO COMPLETE THE HEARING INFORMATION.
Date:	
Time:	AM
Street Address:	
Zip Code:	
URL for Virtual:	
	**************************************
Date:	
Time:	AM
Street Address:	
Zip Code:	
URL for Virtual:	

Administrative Procedures
Final Proposed Filing – Coversheet

Date:
Time: AM
Street Address:
Zip Code:
URL for Virtual:

Date:

Time:

AM

Street Address:

Zip Code:

URL for Virtual:

21. DEADLINE FOR COMMENT (NO EARLIER THAN 7 DAYS FOLLOWING LAST HEARING):

KEYWORDS (PLEASE PROVIDE AT LEAST 3 KEYWORDS OR PHRASES TO AID IN THE SEARCHABILITY OF THE RULE NOTICE ONLINE).

Nursing home

Medicaid

Per diem rate

Residential

## Adopting Page

#### **Instructions:**

This form must accompany each filing made during the rulemaking process:

Note: To satisfy the requirement for an annotated text, an agency must submit the entire rule in annotated form with proposed and final proposed filings. Filing an annotated paragraph or page of a larger rule is not sufficient. Annotation must clearly show the changes to the rule.

When possible, the agency shall file the annotated text, using the appropriate page or pages from the Code of Vermont Rules as a basis for the annotated version. New rules need not be accompanied by an annotated text.

- 1. TITLE OF RULE FILING:
  Nursing Home Reimbursement Rule Simplification
- 2. ADOPTING AGENCY: Agency of Human Services
- 3. TYPE OF FILING (PLEASE CHOOSE THE TYPE OF FILING FROM THE DROPDOWN MENU BASED ON THE DEFINITIONS PROVIDED BELOW):
  - **AMENDMENT** Any change to an already existing rule, even if it is a complete rewrite of the rule, it is considered an amendment if the rule is replaced with other text.
  - **NEW RULE** A rule that did not previously exist even under a different name.
  - **REPEAL** The removal of a rule in its entirety, without replacing it with other text.

This filing is AN AMENDMENT OF AN EXISTING RULE

4. LAST ADOPTED (PLEASE PROVIDE THE SOS LOG#, TITLE AND EFFECTIVE DATE OF THE LAST ADOPTION FOR THE EXISTING RULE):



State of Vermont
Agency of Administration
109 State Street
Montpelier, VT 05609-0201
www.aoa.vermont.gov

[phone] 802-828-3322

#### INTERAGENCY COMMITTEE ON ADMINISTRATIVE RULES (ICAR) MINUTES

Meeting Date/Location: January 8, 2024, virtually via Microsoft Teams

Members Present: Chair Sean Brown, Jared Adler, Jennifer Mojo, Diane Sherman, Michael

Obuchowski, and Nicole Dubuque

Members Absent: John Kessler

Minutes By: Melissa Mazza-Paquette

- 2:00 p.m. meeting called to order, welcome and introductions.
- Review and approval of <u>minutes</u> from the November 13, 2023 meeting.
- No additions/deletions to agenda. Agenda approved as drafted.
- · No public comments made.
- Presentation of Proposed Rules on pages 2-12 to follow.
  - 1. Administration of Nonemergency Involuntary Psychiatric Medications, Department of Mental Health, page 2
  - 2. Vital Records Rule, Department of Health, page 3
  - 3. VPharm Coverage, Department of Vermont Health Access, page 4
  - 4. Rules Governing the Importation of Domestic Animals, Including Livestock and Poultry, Vermont Agency of Agriculture, Food & Markets, page 5
  - 5. Antidegradation Implementation Rule, Agency of Natural Resources, page 6
  - 6. Reach Up Eligibility Rules, Department for Children and Families, page 7
  - 7. Reach Up Services Rules, Department for Children and Families, page 8
  - 8. Reach First Rules, Department for Children and Families, page 9
  - 9. Postsecondary Education Program Rules, Department for Children and Families, page 10
  - 10. Private Nonmedical Institution Rules Simplification, Department of Vermont Health Access, page 11
  - 11. Nursing Home Reimbursement Rule Simplification, Department of Vermont Health Access, page 12
- Next scheduled meeting is Monday, February 23, 2024 at 1:00 p.m.
- 3:38 p.m. meeting adjourned.



**Proposed Rule:** Nursing Home Reimbursement Rule Simplification, Department of Vermont Health Access

Presented By: James LaRock

Motion made to accept the rule by Sean Brown, seconded by Nicole Dubuque, and passed unanimously with the following recommendations:

- 1. Proposed Filing Coversheet:
  - a. #7: Provide a detailed explanation of where the authority is to adopt this without rulemaking,
  - b. #8: Include more detail; a summary of substantive changes. Clarify why the rule is necessary. Define what is being kept in the rule and what is going into the manual.
- 2. Throughout the proposed rule filing: Clarify where necessary that you're retaining part of the rule.

## **Economic Impact Analysis**

#### **Instructions:**

In completing the economic impact analysis, an agency analyzes and evaluates the anticipated costs and benefits to be expected from adoption of the rule; estimates the costs and benefits for each category of people enterprises and government entities affected by the rule; compares alternatives to adopting the rule; and explains their analysis concluding that rulemaking is the most appropriate method of achieving the regulatory purpose. If no impacts are anticipated, please specify "No impact anticipated" in the field.

Rules affecting or regulating schools or school districts must include cost implications to local school districts and taxpayers in the impact statement, a clear statement of associated costs, and consideration of alternatives to the rule to reduce or ameliorate costs to local school districts while still achieving the objectives of the rule (see 3 V.S.A. § 832b for details).

Rules affecting small businesses (excluding impacts incidental to the purchase and payment of goods and services by the State or an agency thereof), must include ways that a business can reduce the cost or burden of compliance or an explanation of why the agency determines that such evaluation isn't appropriate, and an evaluation of creative, innovative or flexible methods of compliance that would not significantly impair the effectiveness of the rule or increase the risk to the health, safety, or welfare of the public or those affected by the rule.

#### 1. TITLE OF RULE FILING:

Nursing Home Reimbursement Rule Simplification

#### 2. ADOPTING AGENCY:

Agency of Human Services

#### 3. CATEGORY OF AFFECTED PARTIES:

LIST CATEGORIES OF PEOPLE, ENTERPRISES, AND GOVERNMENTAL ENTITIES POTENTIALLY AFFECTED BY THE ADOPTION OF THIS RULE AND THE ESTIMATED COSTS AND BENEFITS ANTICIPATED:

No impact anticipated.

#### 4. IMPACT ON SCHOOLS:

INDICATE ANY IMPACT THAT THE RULE WILL HAVE ON PUBLIC EDUCATION, PUBLIC SCHOOLS, LOCAL SCHOOL DISTRICTS AND/OR TAXPAYERS CLEARLY STATING ANY ASSOCIATED COSTS:

No impact anticipated.

5. ALTERNATIVES: Consideration of alternatives to the rule to reduce or ameliorate costs to local school districts while still achieving the objective of the rule.

No impact anticipated.

#### 6. IMPACT ON SMALL BUSINESSES:

INDICATE ANY IMPACT THAT THE RULE WILL HAVE ON SMALL BUSINESSES (EXCLUDING IMPACTS INCIDENTAL TO THE PURCHASE AND PAYMENT OF GOODS AND SERVICES BY THE STATE OR AN AGENCY THEREOF):

No impact anticipated.

7. SMALL BUSINESS COMPLIANCE: EXPLAIN WAYS A BUSINESS CAN REDUCE THE COST/BURDEN OF COMPLIANCE OR AN EXPLANATION OF WHY THE AGENCY DETERMINES THAT SUCH EVALUATION ISN'T APPROPRIATE.

No impact anticipated.

#### 8. COMPARISON:

COMPARE THE IMPACT OF THE RULE WITH THE ECONOMIC IMPACT OF OTHER ALTERNATIVES TO THE RULE, INCLUDING NO RULE ON THE SUBJECT OR A RULE HAVING SEPARATE REQUIREMENTS FOR SMALL BUSINESS:

No impact anticipated.

9. SUFFICIENCY: DESCRIBE HOW THE ANALYSIS WAS CONDUCTED, IDENTIFYING RELEVANT INTERNAL AND/OR EXTERNAL SOURCES OF INFORMATION USED.

The change in this rule will not have any direct economic impact. Vermont Medicaid intends to make future changes using the more liberal reimbursement adjustment process this rule allows which are likely to have economic impacts. Vermont Medicaid will assess the economic impacts of those changes using the Global Commitment Register process and in future budget requests from the Legislature.

## **Environmental Impact Analysis**

#### **Instructions:**

In completing the environmental impact analysis, an agency analyzes and evaluates the anticipated environmental impacts (positive or negative) to be expected from adoption of the rule; compares alternatives to adopting the rule; explains the sufficiency of the environmental impact analysis. If no impacts are anticipated, please specify "No impact anticipated" in the field.

Examples of Environmental Impacts include but are not limited to:

- Impacts on the emission of greenhouse gases
- Impacts on the discharge of pollutants to water
- Impacts on the arability of land
- Impacts on the climate
- Impacts on the flow of water
- Impacts on recreation
- Or other environmental impacts
- 1. TITLE OF RULE FILING:

Nursing Home Reimbursement Rule Simplification

2. ADOPTING AGENCY:

Agency of Human Services

- 3. GREENHOUSE GAS: EXPLAIN HOW THE RULE IMPACTS THE EMISSION OF GREENHOUSE GASES (E.G. TRANSPORTATION OF PEOPLE OR GOODS; BUILDING INFRASTRUCTURE; LAND USE AND DEVELOPMENT, WASTE GENERATION, ETC.):

  No impact anticipated.
- 4. WATER: EXPLAIN HOW THE RULE IMPACTS WATER (E.G. DISCHARGE / ELIMINATION OF POLLUTION INTO VERMONT WATERS, THE FLOW OF WATER IN THE STATE, WATER QUALITY ETC.):

No impact anticipated.

5. LAND: EXPLAIN HOW THE RULE IMPACTS LAND (E.G. IMPACTS ON FORESTRY, AGRICULTURE ETC.):

No impact anticipated.

6. RECREATION: EXPLAIN HOW THE RULE IMPACTS RECREATION IN THE STATE: No impact anticipated.

- 7. CLIMATE: EXPLAIN HOW THE RULE IMPACTS THE CLIMATE IN THE STATE: No impact anticipated.
- 8. OTHER: EXPLAIN HOW THE RULE IMPACT OTHER ASPECTS OF VERMONT'S ENVIRONMENT:
  No impact anticipated.
- 9. SUFFICIENCY: DESCRIBE HOW THE ANALYSIS WAS CONDUCTED, IDENTIFYING RELEVANT INTERNAL AND/OR EXTERNAL SOURCES OF INFORMATION USED. The Division's environmental analysis was cursory because the rule will not affect the operations, inputs, or outputs of Vermont nursing homes.

## Public Input Maximization Plan

#### **Instructions:**

Agencies are encouraged to hold hearings as part of their strategy to maximize the involvement of the public in the development of rules. Please complete the form below by describing the agency's strategy for maximizing public input (what it did do, or will do to maximize the involvement of the public).

This form must accompany each filing made during the rulemaking process:

1. TITLE OF RULE FILING:

Nursing Home Reimbursement Rule Simplification

2. ADOPTING AGENCY:

Agency of Human Services

3. PLEASE DESCRIBE THE AGENCY'S STRATEGY TO MAXIMIZE PUBLIC INVOLVEMENT IN THE DEVELOPMENT OF THE PROPOSED RULE, LISTING THE STEPS THAT HAVE BEEN OR WILL BE TAKEN TO COMPLY WITH THAT STRATEGY:

The Division of Rate Setting has regularly met with its state government partners to discuss improvements to the nursing home reimbursement process. The Division further regularly conducts outreach with nursing home providers. Vermont Medicaid has regularly met with VHCA to discuss the required change to resident acuity classification that will be addressed in the accompanying manual and has also discussed other changes to the reimbursement process that would stabilize the nursing home system for all providers. Vermont Medicaid is committed to engaging with VHCA into the future to roll out these changes.

Further, when AHS files a rule with the Office of the Secretary of State, AHS provides notice and access to the rule through the Global Commitment Register (GCR). The GCR provides notification of policy changes and clarifications of existing Medicaid policy, including rulemaking, under Vermont's 1115 Global Commitment to Health waiver. Anyone can subscribe to the GCR.

#### Public Input

Proposed, final proposed, and adopted rules, including all public comments and responses to rulemaking, are posted to the GCR. Subscribers receive email notification of rule filings including hyperlinks to posted documents and an explanation of how to provide comment and be involved in the rulemaking.

- 4. BEYOND GENERAL ADVERTISEMENTS, PLEASE LIST THE PEOPLE AND ORGANIZATIONS THAT HAVE BEEN OR WILL BE INVOLVED IN THE DEVELOPMENT OF THE PROPOSED RULE:
  - -The Department of Disabiltiies, Aging, and Independent Living
  - -The Agency of Human Services
  - -The Department of Vermont Health Access, Division of Rate Setting
  - -The Vermont Health Care Association

## Incorporation by Reference

# THIS FORM IS ONLY REQUIRED WHEN INCORPORATING MATERIALS BY REFERENCE. PLEASE REMOVE PRIOR TO DELIVERY IF IT DOES NOT APPLY TO THIS RULE FILING:

#### **Instructions:**

In completing the incorporation by reference statement, an agency describes any materials that are incorporated into the rule by reference and how to obtain copies.

This form is only required when a rule incorporates materials by referencing another source without reproducing the text within the rule itself (e.g., federal or national standards, or regulations).

Incorporated materials will be maintained and available for inspection by the Agency.

1. TITLE OF RULE FILING:

Nursing Home Reimbursement Rule Simplification

2. ADOPTING AGENCY:

Agency of Human Services

3. DESCRIPTION (DESCRIBE THE MATERIALS INCORPORATED BY REFERENCE):

Section 4.1 of the Division's nursing home rate-setting rules incorporates Medicare Provider Reimbursement Manual (CMS Publication 15, formerly known as HCFA-15) by reference. This rule filing does not create or amend Section 4.1 of the Division's rules.

- 4. FORMAL CITATION OF MATERIALS INCORPORATED BY REFERENCE:
  - Centers for Medicare and Medicaid Services, The Provider Reimbursement Manual, Publication #15-1, available at https://www.cms.gov/Regulations-and-Guidance/Guidance/Manuals/Paper-Based-Manuals-Items/CMS021929.
- 5. OBTAINING COPIES: (EXPLAIN WHERE THE PUBLIC MAY OBTAIN THE MATERIAL(S) IN WRITTEN OR ELECTRONIC FORM, AND AT WHAT COST):

The public may obtain a copy of the entire Provider Reimbursement Manual in electronic form for free at

https://www.cms.gov/Regulations-and-Guidance/Guidance/Manuals/Paper-Based-Manuals-Items/CMS021929

6. MODIFICATIONS (PLEASE EXPLAIN ANY MODIFICATION TO THE INCORPORATED MATERIALS E.G., WHETHER ONLY PART OF THE MATERIAL IS ADOPTED AND IF SO, WHICH PART(S) ARE MODIFIED):

The Provider Reimbursement Manual is incorporated by reference to the extent that the PNMI regulations, or the proposed PNMI manual, do not already address whether a particular cost is allowable or reasonable or how to treat a particular reimbursement issue.

Run Spell Check



State of Vermont
Agency of Human Services
280 State Drive
Waterbury, VT 05671-1000
www.humanservices.vermont.gov

[phone] 802-241-0440 [fax] 802-241-0450 Jenney Samuelson, Secretary Todd Daloz, Deputy Secretary

Date: March 15, 2024

**Re:** Response to Public Comment on proposed Health Care Administrative Rule (HCAR) Methods, Standards, and Principles for Establishing Medicaid Payment Rates for Long-Term Care Facilities (5.101).

The Department received one comment on the proposed rule from the Vermont Health Care Association (VHCA), a trade association for Vermont long-term care facilities including nursing homes. A summary of comments received is included below along with responses to comments from the Agency of Human Services (AHS).

#### Response to Comments on Rule #24P006

The comment raised a few general themes and made specific suggestions about the text of the rule.

#### 1. General themes

In general, the comment supported the main goal of this proposed rule: simplifying the Division of Rate Setting's Rules and transferring the details of the rate-setting process to a new reimbursement manual to allow for greater flexibility to respond to changing circumstances. The Division appreciates this positive comment.

VHCA also commented generally that it is "concerned that more substantive changes to the underlying rate structure are also needed" and that "the reimbursement structure, as proposed, does not fulfill" Vermont state policy regarding nursing home reimbursement. This proposed rule does not alter the basic reimbursement structure for Vermont nursing homes. The Division of Rate Setting will continue to set per diem rates for nursing homes based on each home's historically allowable costs. This comment therefore does not address issues raised in this rulemaking.

To the extent that this comment suggests significantly increasing reimbursement at Vermont nursing homes, this rule change is not the appropriate forum for doing so. The main goal of this proposed rule is to move the details of the rate-setting process to a new reimbursement manual. VHCA is currently advocating for increased reimbursements to nursing homes and other reforms to the rate setting process at the Legislature. That is the appropriate forum for this discussion.

The Division of Rate Setting, as well as partners across the Agency of Human Services, is currently working with VHCA to plan for an eventual transition to a new system for assessing resident acuity. The state is happy to continue working with VHCA to identify appropriate reforms Vermont's nursing home rate setting process and ensure that any proposed changes are implemented with VHCA's full

knowledge and opportunity to comment, as well as the Legislature's fiscal support. The state is committed to transparency in the reimbursement process.

#### 2. Specific comments

VHCA made numerous comments recommending specific changes to the Division's rate setting system and structure. The majority of these comments are not comments on the proposed rule because they deal with content in the draft reimbursement manual. VHCA will have the opportunity to comment on the draft reimbursement manual when it is actually proposed. Before then, VHCA will have the opportunity to advocate that the Agency's Choices for Care Budget be increased to fund its proposals, as it is currently doing at the Legislature.

Accordingly, the state has no response to VHCA's general comments advocating that the state reduce the number of years for costs to be fully rebased, that the state change how certain costs are included in existing cost categories, that the state reduce the minimum occupancy requirement to 80% from 90%, or that the state clarify the availability of financial relief or special rates. These matters are outside the scope of this rulemaking. If this rule is adopted as proposed, the state will be more easily able to accommodate any changes in these areas that are approved and funded by the Legislature in future rate-setting cycles.

VHCA also addressed comments to specific provisions of the Division's existing and proposed rules.

#### a. Comment on Rule 5.101.5.4: Interim Rates

In this rule, the state proposed transitioning rule 5.4 of the Division of Rate Setting's existing rules unchanged to Rule 5.101.5.4 of the Agency of Human Services' unified rule chapters. These rules allow the Division to set an interim rate for facilities that have not yet received a final rate during the rate setting process.

In response to this proposed rule, VHCA argued that the state could use the interim rate process to increase facilities' rates while they await the next rebase for a given cost center. This comment does not propose a change to this rule; it merely recommends that the state interpret its existing rules differently.

Further, this comment misunderstands the purpose of interim rates. The Division sets an interim rate if it has not completed its review of a nursing home's cost reports and set a final rate. The Division may not set an interim rate for a facility that already has a final rate. The Division's goal each year is to quickly and efficiently review each facility's cost reports and set an accurate final rate as soon as possible. The state has made no change in response to this comment.

#### b. Comment on Rule 5.101.5.7: Occupancy Level

In this rule, the state proposed removing the 90% minimum occupancy requirement, which penalizes nursing homes that fail to maintain that target occupancy, and transitioning the target amount to its proposed reimbursement manual. Reducing the minimum occupancy requirement has a large fiscal impact. Putting the target in a reimbursement manual would allow the state to change this requirement more easily if the state has fiscal support to do so or if the Legislature orders the state to increase the requirement.

In response to this proposed rule, VHCA suggested granting the Director of the Division of Rate Setting greater authority to waive this requirement. The state did not propose expanding the Division's authority to waive this requirement. Further, waiving this requirement for homes outside the narrow situations where it is already waived could weaken its force, may be inequitable, and would require the state to incur a large cost in between budgeting cycles that the nursing home rate setting process is intended to avoid. The state has made no change in response to this comment.

#### c. Comment on 5.101.8.2: Change in Law

In this rule, the state proposed transitioning rule 8.2 of the Division of Rate Setting's existing rules unchanged to Rule 5.101.8.2 of the Agency of Human Services' unified rule chapters. The existing rule allows nursing homes to apply for a rate increase "for additional costs that are a necessary result of complying with changes in applicable federal and state laws and regulations, or the orders of a state agency that specifically requires an increase in staff or other expenditures."

In response to this proposed rule, VHCA argued that the state should remove the "specifically requires" clause to increase the state's ability to adjust rates in response to changes in state or federal law. VHCA believes that this rule is only applicable if any change in law "specifically requires an increase in staff or other expenditures." This comment misinterprets the existing rule. The existing rule sets out two justifications for applying for a rate increase: 1) to address costs "that are a necessary result" of changes to "federal and state laws," or 2) to address costs "that are a necessary result of complying with ... the orders of a state agency that specifically requires an increase in staff." Because the state interprets the rule in accordance with what VHCA already requests, the state has made no change in response to this comment.

VHCA further comments that if a "sweeping change causes a sector-wide impact," requiring facilities to apply for a rate increase on a case-by-case basis could "produce unequal results." The state disagrees. First, the basic purpose of the nursing home rate setting system is to set specific rates based on specific costs incurred by each facility on a case-by-case basis. Under bedrock federal Medicaid reimbursement principles, the state's per diem rates must be based on specific and allowable costs. Second, it would be more inequitable to adopt a one-size-fits-all rate adjustment for all facilities when any change, even a "sweeping change" with a "sector-wide impact," will fall on different facilities across the state differently according to their unique circumstances. The state has made no change in response to this comment.

#### d. Comment on 5.101.10.1: Extraordinary Financial Relief

In this rule, the state proposed transitioning rule 10 of the Division of Rate Setting's existing rules unchanged to Rule 5.101.10 of the Agency of Human Services' unified rule chapters. These rules allow facilities that are in danger of immediate failure to apply for extraordinary financial relief (EFR). EFR allows the state to stabilize a facility in the short term, protect the residents, and plan for the long-term future of the facility. It is funded generally out of unspent dollars in the Choices for Care budget.

In response to this proposed rule, VHCA comments that it generally supports the availability of EFR but hopes to return EFR to being an exception that is rarely used, as it was prior to the

COVID-19 pandemic and attendant economic consequences for nursing homes. The state agrees with this sentiment, but this is not a comment on the text of the proposed rule.

Further, VHCA specifically recommends adding language to set specific timelines for when facilities should submit EFR applications, how long the state will take to respond, and options for "interim decisions." The state disagrees with this comment. Each facility's fiscal situation, and potential for failure, are unique, so EFR is necessarily a case-by-case analysis. There is no one-size-fits-all timeline for applying for, or receiving, EFR. The rule already grants the state broad discretion to grant interim relief, including Medicaid advances, as it frequently does if the facility is in danger of failing before the state can complete its review. Accordingly, the state has made no change in response to this comment.

e. Comment on 5.101.6: Base Year Cost Categories for Nursing Facilities, and related comments on sections of the draft Reimbursement Manual

In this rule, the state proposed removing specific details about which costs are included in the cost categories that it uses to construct the components of each facility's rate and replacing this information in its new reimbursement manual. The state "rebases" the cost categories on a regular cycle of "base years." In between base years, the growth of costs in each category is tightly controlled, but facilities also receive an annual inflation adjustment under state law. All costs are rebased at least every four years. Nursing costs, which are the largest category by far, are rebased every two years.

In response to this proposed rule, VHCA generally complains that the rebase cycle is too long and that costs rise higher than the pace of the inflation adjustments that all nursing homes receive. Further, VHCA suggests changing language in the proposed reimbursement manual to change which categories correspond to certain classes of employees at nursing homes, including changes to how the resident care, indirect, and director of nursing cost categories are calculated. These are not comments on the text of the proposed rule, but on the draft reimbursement manual. Adjusting the rebase cycle for any cost category, particularly for nursing costs, would have a substantial fiscal impact. Adjusting how the state calculates cost categories may also have a fiscal impact. VHCA will have the opportunity to comment on the draft reimbursement manual once it is actually proposed. Before then, VHCA will have the opportunity to advocate that the Agency's Choices for Care Budget be increased to fund its proposals, as it is currently doing at the Legislature.

f. Comment on 5.101.7: Calculation of Costs, Limits and Rate Components for Nursing Facilities

In this rule, the state proposed removing the specific details about how the Division of Rate Setting calculates overall costs in each category and any limits that may apply and replacing this information in its new reimbursement manual.

In response to this proposed rule, VHCA noted that it generally supported moving the details of calculations to a reimbursement manual, but suggested that this rule could include "guidance on the frequency of rebasing." Specifically, VHCA suggested that the rules should require the state to mandate costs be rebased on their current cycle or more frequently.

The state disagrees with this comment. First, the overall goal of this rule change is to move details about the rate setting process such as this to the reimbursement manual to allow for greater flexibility. This change is value-neutral – it is not intended to help or harm nursing homes in general. Just as the state has proposed moving details about the minimum occupancy requirement or cost caps to a reimbursement manual, the state is also proposing that the pace of the rebase cycle be moved to a reimbursement manual. Second, the pace of the rebase cycle has significant fiscal implications. At the moment, the state does not intend to propose that rebases happen more or less frequently than their current cycle. If the state were to do so, it would do so first by proposing the change in the Legislature, which is best suited to balance the competing considerations implicit in this policy question. Because the state would, by necessity, seek fiscal authority from the Legislature to make any change to the rebase cycle, whether the "guidelines" VHCA suggests would exist in the rules or in a reimbursement manual is immaterial. Accordingly, the state has made no change in response to this comment.

#### 3. Conclusion

The state welcomes VHCA's thoughtful comments on proposed rule #24P006 and VHCA's further engagement in discussions on changes to the rate setting process. The state notes that VHCA generally supports its proposal to transition many details of the nursing home rate setting process to a reimbursement manual. The state has made no changes to its proposed rule in response to this comment.

Annotated

# STATE OF VERMONT AGENCY OF HUMAN SERVICES DIVISION OF RATE SETTING



# METHODS, STANDARDS AND PRINCIPLES FOR ESTABLISHING MEDICAID PAYMENT RATES FOR LONG-TERM CARE FACILITIES

**MARCH 2015** 

	-			

#### **TABLE-OF-CONTENTS**

Cite as Vermont Division of Rate Setting Rules (V.D.R.S.R.)

1	GENERAL PROVISIONS	1
	1.1 Purpose	
	1.2 Scope	
	1.3 Authority	
	1.4 General Description of the Rate Setting System	
	1.5—Requirements for Participation in Medicaid Program	
	1.6 Responsibilities of Owners.	
	1.7 Duties of the Owner	
	1.8 Powers and Duties of the Division and the Director	2
	1.9 Powers and Duties of the Department of Disabilities, Aging and Independent Living's Division	
	Licensing and Protection as Regards Reimbursement	3
	1.10 Computation of and Enlargement of Time; Filing and Service of Documents	
	1.11 Representation in All Matters before the Division	
	1.12 Severability	
	1.13 Effective Date	4
2	ACCOUNTING REQUIREMENTS	
	2.1—Accounting Principles	4
	2.2 Procurement Standards	5
	2.3 Cost Allocation Plans and Changes in Accounting Principles	5
	2.4—Substance Over Form	6
	2.5 Record Keeping and Retention of Records	6
	• •	
3	FINANCIAL REPORTING	6
	3.1 [Repealed]	6
	3.2 Uniform Cost Reports	
	3.3—Adequacy and Timeliness of Filing	
	3.4—Review of Cost Reports by Division	
	3.5 Settlement of Cost Reports.	
	19 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	
4	DETERMINATION OF ALLOWABLE COSTS FOR NURSING FACILITIES	9
	4.1—Provider Reimbursement Manual and GAAP	9
	4.2 General Cost Principles	
	4.3 Non-Recurring Costs	
	4.4 Interest Expense	
	4.5 Basis of Property, Plant and Equipment.	
	4.6 Depreciation and Amortization of Property, Plant and Equipment	
	4.7 Change in Ownership of Depreciable Assets Sales of Facilities.	
	4.8 [Repealed]	
	4.9 Leasing Arrangements for Property, Plant and Equipment	
	4.10 Funding of Depreciation	
	4.11 Adjustments for Large Asset Acquisitions and Changes of Ownership	
	4.12 [Repealed]	17
	4.13 Advertising Expenses 4.14 Barber and Beauty Service Costs	
	4.14 Barber and Beauty Service Costs 4.15 Bad Debt, Charity and Courtesy Allowances	
	4.15 Bad Debt, Charity and Courtesy Allowances 4.16 Child Day Care	
	4.17 Community Service Activities 4.18 Dental Services	
	4.19 Legal Costs	17 17
	4 /U LINGHIOR SNA ACHIEMENT LOSIS	

	4.21	Motor Vehicle Allowance	18
	4.22	Non-Competition Agreement Costs	
	4.23	Compensation of Owners, Operators, or their Relatives	18
		Management Fees and Home Office Costs	
		Membership Dues	
	4.26	Post-Retirement Benefits.	19
	4.27	Public Relations	19
	4.28	Related Party	19
		Revenues	
	4.30	Travel/Entertainment Costs	19
		Transportation Costs	
		Services Directly Billable	
_	n	PIMPUDGEMENT CTANDADDC	10
<b>3</b> -		REIMBURSEMENT STANDARDS	
	5.1	Prospective Case Mix Reimbursement System	19
	5.2	Retroactive Adjustments to Prospective Rates	20
		Lower of Rate or Charges	
		-Interim Rates	
		Upper Payment Limits	
		Base Year	
		Occupancy Level	
	5.8	Inflation Factors	22
		Costs for New Facilities	
	5.10	Costs for Terminating Facilities	23
6-	—В	SASE YEAR COST CATEGORIES FOR NURSING FACILITIES	23
	61	-General	23
		Nursing Care Costs	
		Resident Care Costs	
		Indirect Costs	
		Director of Nursing	
	6-6-	Property and Related	24
		Ancillaries	
7-		CALCULATION OF COSTS, LIMITS AND RATE COMPONENTS FOR NURSE	
	I	FACILITIES	25
	7.1	-Calculation of Per Diem Costs	25
		- Curculation of Fet Diefit Costs - Nursing Care Component.	
		Resident Care Base Year Rate	
		Indirect Base Year Rate	
		Director of Nursing Base Year Rate	
		Ancillary Services Rate	
		Property and Related Per Diem	
		-Limits Final	
8-	A	ADJUSTMENTS TO RATES	29
	0 1	Change in Services	20
		Change in Services	
		Change in Law	
		Facilities in Receivership	
		Efficiency Measures	
		Interest Rates	
		Emergencies and Unforeseeable Circumstances	
		Procedures and Requirements for Rate Adjustments	
		Limitation on Availability of Rate Adjustments	
9	P	RIVATE NURSING FACILITY AND STATE NURSING FACILITY RATES	31

#### MEDICAID PAYMENT RATES FOR LONG-TERM CARE FACILITIES

9.1 Nursing Facility Rate Components	31
9.2 Calculation of the Total Rate	31
9.3 Updating Rates for a Change in the Average Case Mix Score	31
9.4 State Nursing Facilities	31
9.5 Quality Incentives	32
10—EXTRAORDINARY FINANCIAL RELIEF	33
10.1 Objective	33
10.2 Nature of the Relief	
10.3 Criteria to be Considered by the Division	33
10.4 Procedure for Application	34
11 PAYMENT FOR OUT-OF-STATE PROVIDERS	34
11.1 Long Term Care Facilities Other Than Rehabilitation Centers	34
11.2 Rehabilitation Centers	34
11.3 Pediatric Care	
12—RATES FOR ICF/MRS	34
12.1 Reasonable Cost Reimbursement	24
12.2 Application of these Rules to ICF/MRS	3/1
13 RATES FOR SWING BEDS AND OTHER LONG-TERM CARE SERVICES IN	•
HOSPITALS	34
14—SPECIAL RATES FOR CERTAIN INDIVIDUAL RESIDENTS	35
14.1 Availability of Special Rates for Individuals with Unique Physical Conditions	35
14.2 Special Rates for Certain Former Patients of the Vermont State Hospital	36
14.3 Special Rates for Medicaid Eligible Furloughees of the Department of Corrections	36
15 ADMINISTRATIVE REVIEW AND APPEALS	36
15.1 Draft Findings and Decisions	36
15.2 Request for an Informal Conference on Draft Findings and Decisions	36
15.3 Request for Reconsideration	37
15.4 Appeals from Final Orders of the Division	38
15.5 Request for Administrative Review to the Secretary of Human Services pursuant to 33	V.S.A.
§909(a)(3)	
15.6 Appeal to Vermont Supreme Court pursuant to 33 V.S.A. §909(a)(1)	
15.7 Appeal to Superior Court pursuant to 33 V.S.A. §909(a)(2)	
15.8 Settlement Agreements	40
16 DEFINITIONS AND TERMS	
17—TRANSITIONAL PROVISIONS	43
17.1 Commenter Alde Dilet Decient	5074152500E

#### 1 GENERAL PROVISIONS

#### 1.1 Purpose

The purpose of these rules is to implement state and federal reimbursement policy with respect to nursing facilities providing services to Medicaid eligible persons. The methods, standards, and principles of rate setting established herein reflect the objectives set out in 33 V.S.A. §901 and balance the competing policy objectives of access, quality, cost containment and administrative feasibility. Rates set under this payment system are consistent with the efficiency, economy, and quality of care necessary to provide services in conformity with state and federal laws, regulations, quality and safety standards, and meet the requirements of 42 U.S.C. §1396a(a)(13)(A).

#### 1.2 Scope

These rules apply to all privately owned nursing facilities and state nursing facilities providing services to Medicaid residents. Long-term care services in swing-bed hospitals, and Intermediate Care Facilities for the Mentally Retarded are reimbursed under different methods and standards. Swing bed hospitals are reimbursed pursuant to 42 U.S.C. §1396l(b)(1). Intermediate Care Facilities for the Mentally Retarded are reimbursed pursuant to the Regulations Governing the Operation of Intermediate Care Facilities for the Mentally Retarded adopted by the Agency and are subject to the Division's Accounting Requirements (Section 2) and Financial Reporting (Section 3).

#### 1.3 Authority

These rules are promulgated pursuant to 33 V.S.A. §§904(a) and 908(e) to meet the requirements of 33 V.S.A. Chapter 9, 42 U.S.C. §§1396a(a)(13)(A) and §1396a(a)(30).

# 1.4 General Description of the Rate Setting System

A prospective case mix payment system for nursing facilities is established by these rules in which the payment rate for services is set in advance of the actual provision of those services. A per diem rate is set for each facility based on the historic allowable costs of that facility. The costs are divided into certain designated cost categories, some of which are subject to limits. The basis for reimbursement within the Nursing Care cost category is a resident classification system that groups residents into classes according to their assessed conditions and the resources required to care for them. The costs in some categories are adjusted to reflect economic trends and conditions, and the payment rate for each facility is based on the per diem costs for each category.

# 1.5 Requirements for Participation in Medicaid Program

- (a) Nursing facilities must satisfy all of the following prerequisites in order to participate in the Medicaid program:
  - (1) be licensed by the Agency, pursuant to 33 V.S.A. §7103(b),
  - (2) be certified by the Secretary of Health and Human Services pursuant to 42 C.F.R. Part 442, Subpart C, and
  - (3) have executed a Provider Agreement with the Agency, as required by 42 C.F.R. Part 442, Subpart B.
- (b) To the extent economically and operationally feasible, providers are encouraged, but not required, to be certified for participation in the Medicare program, pursuant to 42 C.F.R. §488.3.
- (c) Medicaid payments shall not be made to any facility that fails to meet all the requirements of Subsection 1.5(a).

#### 1.6 Responsibilities of Owners

The owner of a nursing facility shall prudently manage and operate a residential health care program of adequate quality to meet its residents' needs. Neither the issuance of a per diem rate, nor final orders made by the Director or a duly authorized representative shall in any way relieve the owner of a nursing facility from full responsibility for compliance with the requirements and standards of the Agency of Human Services.

#### 1.7 Duties of the Owner

The owner of a nursing facility, or a duly authorized representative shall:

- (a) Comply with the provisions of Subsections 1.5 and 1.6 setting forth the requirements for participation in the Medicaid Program.
- (b) Submit cost reports in accordance with the provisions of subsections 3.2 and 3.3 of these rules.
- (e) Maintain adequate financial and statistical records—and—make—them—available—at reasonable—times—for—inspection—by—an authorized representative of the Division, the state, or the federal government.
- (d) Assure that an annual audit is performed in conformance with Generally Accepted Auditing Standards (GAAS).
- (e) Assure that the construction of buildings and the maintenance and operation of premises and programs comply with all applicable health and safety standards.
- (f) Notwithstanding any other provision of these rules, any provider that fails to make a complete cost report filing within the time prescribed in subsection 3.3(a) or fails to file any other materials requested by the Division within the time prescribed shall receive no increase to its Medicaid rate until the first day of the calendar quarter after a complete

eost report or the requested materials are filed, unless within an extension of time previously approved by the Division.

# 1.8 Powers and Duties of the Division and the Director

- (a) The Division shall establish and certify to the Department of Vermont Health Access per diem rates for payment to providers of nursing facility services on behalf of residents eligible for assistance under Title XIX of the Social Security Act.
- (b) The Division may request any nursing facility or related party or organization to file such relevant and appropriate data, statistics, schedules or information as the Division finds necessary to enable it to carry out its function.
- (e) The Division may examine books and accounts of any nursing facility and related parties or organizations, subpoena witnesses and documents, administer oaths to witnesses and examine them on all matters over which the Division has jurisdiction.
- (d) From time to time, the Director may issue notices of practices and procedures employed by the Division in carrying out its functions under these rules.
- (e) The Director shall prescribe the forms required by these rules and instructions for their completion.
- (f) Copies of each notice of practice and procedure, form, or set of instructions shall be sent to each nursing facility participating in the Medicaid program at the time it is issued. A compilation of all such documents currently in force shall be maintained at the Division, pursuant to 3 V.S.A. §835, and shall be available to the public.
- (g) Neither the issuance of final per diem rates nor Final Orders of the Division which fail, in any one or more instances, to enforce the performance of any of the terms or conditions of these rules shall be construed as

a waiver of the Division's future performance of the right. The obligations of the provider with respect to performance shall continue, and the Division shall not be estopped from requiring such future performance.

# 1.9 Powers and Duties of the Department of Disabilities, Aging and Independent Living's Division of Licensing and Protection as Regards Reimbursement

(a) The Division of Licensing and Protection of the Department of Disabilities, Aging and Independent Living shall receive from providers resident assessments on forms it specifies. The Department of Disabilities, Aging and Independent Living shall process this information and shall periodically, but no less frequently than quarterly, provide the Division of Rate Setting with the average case mix scores of each facility based upon the federal RUG IV classification system (48 group version). This score will be used in the quarterly determination of the Nursing Care portion of the rate.

(b) The management of the resident assessment process used in the determination of case mix scores shall be the duty of the Division of Licensing and Protection of the Department of Disabilities, Aging and Independent Living. Any disagreements between the facility's assessment of a resident and the assessment of that same resident by the audit staff of Licensing and Protection shall be resolved with the Division of Licensing and Protection and shall not involve the Division of Rate Setting. As the final rates are prospective and adjusted on a quarterly basis to reflect the most current data, the Division of Rate Setting will not make retroactive rate adjustments as a result of audits or successfully appealed individual case-mix scores

# 1.10 Computation of and Enlargement of Time; Filing and Service of Documents

(a) In computing any period of time prescribed or allowed by these rules, the day of the act or event from which the designated

period of time begins to run shall not be included. The last day of the period so computed shall be included, unless it is a Saturday, a Sunday, or a state or federal legal holiday, in which event the period runs until the end of the next day which is not a Saturday, a Sunday, or a state or federal legal holiday.

(b) For the purposes of any provision of these rules in which time is computed from the receipt of a notice or other document issued by the Division or other relevant administrative officer, the addressee of the notice shall be rebuttably presumed to have received the notice or other document three days after the date on the document.

(e) When by these rules or by a notice given thereunder, an act is required or allowed to be done at or within a specified time, the relevant administrative officer, for just cause shown, may at any time in her or his discretion, with or without motion or notice, order the period enlarged. This subsection shall not apply to the time limits for appeals to the Vermont Supreme Court or Superior Court from Final Orders of the Division or Final Determinations of the Secretary, which are governed by the Vermont Rules of Appellate Procedure and the Vermont Rules of Civil Procedure respectively.

(d) Filing shall be deemed to have occurred when a document is received and datestamped as received at the office of the Division or in the case of a document directed to be filed under this rule other than at the office of the Division, when it is received and stamped as received at the appropriate office. Filings with the Division may be made by telefacsimile (FAX), but the sender bears the risk of a communications failure from any cause. Filings with the Division may also be made electronically, but the sender bears the risk of a communications failure from any cause, including, but not limited to, filings blocked due to size.

(e) Service of any document required to be served by this rule shall be made by delivering a copy of the document to the person or entity required to be served or to his or her representative or by sending a copy by prepaid first class mail to the official service address. Service by mail is complete upon mailing.

# 1.11 Representation in All Matters before the

(a) A facility may be represented in any matter under this rule by the owner (in the ease of a corporation, partnership, trust, or other entity created by law, through a duly authorized agent), the nursing facility administrator, or by a licensed attorney or an independent public accountant.

(b) The provider shall file written notification of the name and address of its representative for—each matter before the Division. Thereafter, on that matter, all correspondence from the Division will be addressed to that representative. The representative of a provider failing to so file shall not be entitled to notice or service of any document in connection with such matter, whether required to be made by the Division or any other person, but instead service shall be made directly on the provider.

#### 1.12 Severability

If any part of these rules or their application is held invalid, the invalidity does not affect other provisions or applications which can be given effect without the invalid provision or application, and to this end the provisions of these rules are severable.

#### **1.13 Effective Date**

(a) These rules are effective from January 29, 1992, (as amended June 18, 1993, July 1, 1994, January 4, 1995, January 1, 1996, January 1, 1997, July 1, 1998, May 1, 1999, July 1, 1999, August 1, 1999, July 1, 2001, November 1, 2002, May 1, 2004, July 1, 2004, July 1, 2004, July 1, 2005, October 29, 2007,

August 25, 2008, April 1, 2011, September 17, 2012, September 9, 2013, and March 6, 2015).

- (b) Application of Rule: Amended provisions of this rule shall apply to:
  - (1) all cost reports draft findings issued on or after the effective date of the most recent amendment, and
  - (2) all rates set on or after the effective date of the most recent amendment.
- (e) With respect to any administrative proceeding pending on the effective date of the most recent amendment the Director or the Secretary may apply any provision of such prior rules where the failure to do so would work an injustice or substantial inconvenience.

#### **2 ACCOUNTING REQUIREMENTS**

#### 2.1 Accounting Principles

- (a) All financial and statistical reports shall be prepared in accordance with Generally Accepted Accounting Principles (GAAP), consistently applied, unless these rules authorize specific variations in such principles.
- (b) The provider shall establish and maintain a financial management system which provides for adequate internal control assuring the accuracy of financial data, safeguarding of assets and operational efficiency.
- (e) The provider shall report on an accrual basis. The provider whose records are not maintained on an accrual basis shall develop accrual data for reports on the basis of an analysis of the available documentation. In such a case, the provider's accounting process shall provide sufficient information to compile data to satisfy the accrued expenditure reporting requirements and to demonstrate the link between the accrual data reports and the non-accrual fiscal accounts.

The provider shall retain all such documentation for audit purposes.

#### 2.2 Procurement Standards

- (a) Providers shall establish and maintain a code of standards to govern the performance of its employees engaged in purchasing goods and services. Such standards shall provide, to the maximum extent practical, open and free competition among vendors. Providers should participate in group purchasing plans when feasible.
- (b) If a provider pays more than a competitive bid for a good or service, any amount over the lower bid which cannot be demonstrated to be a reasonable and necessary expenditure that satisfies the prudent buyer principle is a nonallowable cost.

# 2.3 Cost Allocation Plans and Changes in Accounting Principles

With respect to the allocation of costs to the nursing facility and within the nursing facility, the following rules shall apply:

#### (a) [Repealed]

- (b) Providers that have costs allocated from related entities included in their cost reports shall include, as a part of their cost report submission, a summary of the allocated costs, including a reconciliation of the allocated costs to the entity's financial statements, which must also be submitted with the Medicaid cost report. In the case of a home—office—or—related—management company, this would include a completed Home Office Cost Statement. The provider shall—submit—this—reconciliation—with—the Medicaid cost report.
- (c) The Division reserves the right not to recognize changes in accounting principles or methods or basis of cost allocation made for the purpose or having the likely effect of increasing a facility's Medicaid payments.

- (d) [Repealed]
- (e) [Repealed]
- (f) Each provider shall notify the Division of changes in statistical allocations or record keeping required by the Medicare Intermediary.
- (g) Preferred statistical methods of allocation are as follows:
  - (1) Nursing salaries and supplies direct cost.
  - (2) Plant operations—square footage,
  - (3) Utilities square footage,
  - (4) Laundry pounds of laundry,
  - (5) Dietary resident days,
  - (6) Administrative and General accumulated costs.
  - (7) [Repealed]
  - (8) Property and Related square footage,
  - (9) Fringe Benefits direct allocation/gross salaries.
- (h) Food costs included in allocated dietary costs are calculated by dividing the facility's allocated dietary costs by total organization dietary costs, both of which include allocated overhead, and multiplying the result by the total organization food costs.
- (i) Utility costs included in allocated plant operation and maintenance costs are calculated by dividing the facility's plant operation and maintenance costs by total organization plant operation and maintenance cost, both of which include allocated overhead, and multiplying the result by the total organization utility costs.
- (j) All administrative and general costs, including home office and management

eompany costs, allocated to a facility shall be included in the Indirect Cost category.

(k) The capital component of goods or services purchased or allocated from a related or unrelated party, such as plant operation and maintenance, utilities, dietary, laundry, housekeeping, and all others, whether or not acquired from a related party, shall be considered as costs for that particular good or service and not classified as Property and Related costs of the nursing facility.

(1) Costs allocated to the nursing facility shall be reasonable, as determined by the Division pursuant to these rules.

#### 2.4 Substance Over Form

The cost effect of transactions that have the effect of circumventing the intention of these rules may be adjusted by the Division on the principle that the substance of the transaction shall prevail over the form.

# 2.5 Record Keeping and Retention of Records

(a) Each provider must maintain complete documentation, including accurate financial and statistical records, to substantiate the data reported on the uniform financial and statistical report (cost report), and must, upon request, make these records available to the Division of Rate Setting, or the U. S. Department of Health and Human Services, and the authorized representatives of both agencies.

(b) Complete documentation means clear and compelling evidence of all of the financial transactions of the provider and affiliated entities, including but not limited to census data, ledgers, books, invoices, bank statements, canceled checks, payroll records, copies of governmental filings, time records, time cards, purchase requisitions, purchase orders, inventory records, basis of apportioning costs, matters of provider ownership and organization, resident service schedule and amounts of income received by

service, or any other record which is necessary to provide the Director with the highest degree of confidence in the reliability of the claim for reimbursement. For purposes of this definition, affiliated entities shall extend to realty, management and other entities for which any reimbursement is directly or indirectly claimed whether or not they fall within the definition of related parties.

(c) The provider shall maintain all such records for at least six years from the date of filing, or the date upon which the fiscal and statistical records were to be filed, whichever is the later. The Division shall keep all cost reports, supporting documentation submitted by the provider, correspondence, workpapers and other analyses supporting Summaries of Findings for six years. In the event of litigation or appeal involving rates established under these regulations, the provider and Division shall retain all records which are in any way related to such legal proceeding until the proceeding has terminated and any applicable appeal period has lapsed.

(d) Pursuant to 33 V.S.A. §908(a), all documents and other materials filed with the Division are public information, except for individually identifiable health information protected by law or the policies, practices, and procedures of the Agency of Human Services. With the exception of the administrator's salary, the salaries and wages of individual employees shall not be made public.

#### 3 FINANCIAL REPORTING

#### 3.1 [Repealed]

#### 3.2 Uniform Cost Reports

(a) Each long term care facility participating in the Vermont Medicaid program shall annually submit a uniform financial and statistical report (cost report) on forms prescribed by the Division. The inclusive dates of the reporting year shall be the 12

month period of each provider's fiscal year, unless advance authorization to submit a report for a greater or lesser period has been granted by the Division.

- (1) The Division may require providers to file special cost reports for periods other than a facility's fiscal year.
- (2) The Division may require providers to file budget cost reports. Such cost reports may be used inter alia as the basis for new facilities' rates or for rate adjustments.
- (b) The cost report must include the certification page signed by the owner, or its representative, if authorized in writing by the owner.
- (e) The original and one copy of the cost report must be submitted to the Division. All documents must bear original signatures.
- (d) The following supporting documentation is required to be submitted with the cost report:
  - (1) Audited financial statements (except that at the discretion of the Director, this requirement may be waived).
  - (2) Most recently filed Medicare Cost Report with the required supplemental data on CMS Form 339 (if a participant in the Medicare Program), which for hospital-based nursing homes shall be the Medicare cost report for the same fiscal year as the Medicaid cost report,
  - (3) Independent auditor's adjusting entries and reconciliation of the audited financial statements to the cost report.
- (e) A provider must also submit, upon request during the desk review or audit process, such data, statistics, schedules or other information which the Division requires in order to carry out its function. If, before the draft findings are issued, the facility has been specifically requested to provide certain information or materials and

has failed to do so, such information or materials will not be admissible in any subsequent appeal taken pursuant to Section 15, provided the Division has notified the provider of such failure and afforded the provider a final opportunity to cure.

(f) Providers shall follow the cost report instructions prescribed by the Director in completing the cost report. The chart of accounts prescribed by the Director, shall be used as a guideline providing the titles, and description for type of transactions recorded in each asset, liability, equity, income, and expense account.

#### 3.3 Adequacy and Timeliness of Filing

- (a) With the exception of hospital based nursing homes, an acceptable cost report filing shall be made on or before the last day of the fifth month following the close of the period covered by the report.
  - (1) Hospital based nursing homes shall file their Medicaid cost reports within five days after filing their Medicare cost report for the same cost reporting period with CMS.
  - (2) If a hospital based Medicaid nursing home's cost report is not filed on or before June 30 following the end of the facility's fiscal year, the Division may require the facility to provide certain data or to file a draft cost report.
- (b) The Division may reject any filing which does not comply with these regulations and/or the cost reporting instructions. In such ease, the report shall be deemed not filed, until refiled and in compliance.
- (c) Extensions for filing of the cost report beyond the prescribed deadline must be requested as follows:
  - (1) All Requests for Extension of Time to File Cost Report must be in writing, on a form prescribed by the Director, and must be received by the Division of Rate Setting prior to the due date. The provider must

elearly explain the reason for the request and specify the date on which the Division will receive the report.

- (2) Notwithstanding any previous practice, the Division will not grant automatic extensions. Such extensions will be granted for good cause only, at the Director's sole discretion, based on the merits of each request. A "good cause" is one that supplies a substantial reason, one that affords a legal excuse for the delay or an intervening action beyond the provider's control. The following are not considered "good cause": ignorance of the rule, inconvenience, or a cost report preparer engaged in other work.
- (d) Notwithstanding any other provision of these rules, any provider that fails to make a complete cost report filing within the time prescribed in subsection 3.3(a) or within an extension of time approved by the Division, shall be subject to the provisions of subsection 1.7(f).

#### 3.4 Review of Cost Reports by Division

- (a) Uniform Desk Review
  - (1) The Division shall perform a uniform desk review on each cost report submitted.
- (2) The uniform desk review is an analysis of the provider's cost report to determine the adequacy and completeness of the report, accuracy and reasonableness of the data recorded thereon, allowable costs and a summary of the results of the review for the purpose of either settling the cost report without an on site audit or determining the extent to which an on site audit verification is required.
- (3) Uniform desk reviews shall be completed within an average of 18 months after receipt of an acceptable cost report filing, except in unusual situations, including but not limited to, delays in obtaining necessary information from a provider. Notwithstanding this subdivision, the Division shall have an additional six

months to complete its review or audits of facilities' base year cost reports.

(4) Unless the Division schedules an on-site audit, it shall issue a written summary report of its findings and adjustments upon completion of the uniform desk review.

#### (b) On site Audit

- (1) The Division will perform on site audits, as considered appropriate, of the provider's financial and statistical records and systems in accordance with the relevant provisions of the Medicare Intermediary Manual Audits Reimbursement Program Administration, CMS Publication 13-2 (CMS 13).
- (2) The Division will base its selection of a facility for an on site audit on factors such as length of time since last audit, changes in facility ownership, management, or organizational structure, evidence or official complaints of financial irregularities, questions raised in the uniform desk review, failure to file a timely cost report without a satisfactory explanation, and prior experience.
- (3) The audit scope will be limited so as to avoid duplication of work performed by an independent public accountant, provided such work is adequate to meet the Division's audit requirements.
- (4) Upon completion of an audit, the Division shall review its draft findings and adjustments with the provider and issue a written summary report of such findings.
- (e) The procedure for issuing and reviewing Summaries of Findings is set out in Subsections 15.1, 15.2 and 15.3.

#### 3.5 Settlement of Cost Reports

(a) A cost report is settled if there is no request for reconsideration of the Division's findings or, if such request was made, the

Division has issued a final order pursuant to Subsection 15.3 of these rules.

- (b) Cost report determinations and decisions, otherwise final, may be reopened and corrected when the specific requirements set out below are met. The Division's decision to reopen will be based on new and material evidence submitted by the provider, evidence of a clear and obvious material error, or a determination by the Secretary or a court of competent jurisdiction that the determination is inconsistent with applicable law, regulations and rulings, or general instructions.
- (c) Reopening means an affirmative action taken by the Division to re-examine the correctness of a determination or decision otherwise final. Such action may be taken:
  - (1) On the initiative of appropriate authority within the applicable time period set out in paragraph (f), or
  - (2) In response to a written request of the provider or other relevant entity, filed with the Division within the applicable time period set out in subsection (f), and
  - (3) When the reopening has a material effect (more than one percent) on the provider's Medicaid rate payments.
- (d) A correction is a revision (adjustment) in the Division's determination or Secretary's decision, otherwise final, which is made after a proper re-opening.
- (e) A correction may be made by the Division, or the provider may be required to file an amended cost report. If the cost report is reopened by an order of the Secretary or a court of competent jurisdiction, the correction shall be made by the Division.
- (f) A determination or decision may be reopened within three years from the date of the notice containing the Division's determination, or the date of a decision by the Secretary or a court.

(g) The Division may also require or allow an amended cost report to correct material errors detected subsequent to the filing of the original cost report or to comply with applicable standards and regulations. Once a cost report is filed, the provider is bound by its elections. The Division shall not accept an amended cost report to avail the provider of an option it did not originally elect.

# 4 DETERMINATION OF ALLOWABLE COSTS FOR NURSING FACILITIES

# 4.1 Provider Reimbursement Manual and GAAP

In determining the allowability or reasonableness of costs or treatment of any reimbursement issue, not addressed in these rules, the Division shall apply the appropriate provisions of the Medicare Provider Reimbursement Manual (CMS 15, formerly known as HCFA or HIM 15). If neither these regulations nor CMS-15 specifically addresses a particular issue, the determination of allowability will be made in accordance with Generally Accepted Accounting Principles (GAAP). The Division reserves the right, consistent with applicable law, to determine the allowability and reasonableness of costs in any case not specifically covered in the sources referenced in this subsection

#### 4.2 General Cost Principles

For rate setting purposes, a cost must satisfy criteria, including, but not limited to, the following:

- (a) The cost must be ordinary, reasonable, necessary, related to the care of residents, and actually incurred.
- (b) The cost adheres to the prudent buyer principle.
- (c) The cost is related to goods and/or services actually provided in the nursing facility.

#### 4.3 Non-Recurring Costs

- (a) Non-recurring costs shall include:
  - (1) any reasonable and resident-related cost that exceeds \$10,000, which is not expected to recur on an annual basis in the ordinary operation of the facility, may be designated by the Division as a "Non Recurring Cost" subject to any limits on the cost category into which the type of cost would otherwise be assigned,
  - (2) litigation expenses of \$10,000 or more, recognized pursuant to subsection 4.20.
  - (3) allowable lump-sum-costs of \$2,000 or more per cost reporting period for recruitment and legal fees or similar expenses associated with the hiring of registered nurses from countries outside the United States on condition that such fees or expenses shall be allowable only in respect of such nurses who are paid at least the prevailing salary/wage and benefits for employed nurses of similar qualifications and experience in the geographic area in which the facility is located or tuition expenses for nurse aide training reimbursed pursuant to 42 C.F.R. §483.152(e)(2).
- (b) A non-recurring cost shall be capitalized and amortized and carried as an on-going adjustment beginning with the first quarterly rate change after the settlement of the cost report for a period of three years.

#### 4.4 Interest Expense

- (a) Necessary and proper interest is an allowable cost.
- (b) "Necessary requires that:
  - (1) The interest be incurred on a loan made to satisfy a financial need of the provider.
  - (2) A financial need does not exist if the provider has cash and/or cash equivalents of more than 60 days cash needs.

- (3) Cash and eash equivalents include:
  - (i) monetary investments, including unrestricted grants and gifts,
  - (ii) non-monetary investments not related to resident care that can readily be converted to cash not of any related liability,
  - (iii) receivables from (net of any payables to) officers, owners, partners, parent organizations, or other related parties, excluding education loans to employees.
  - (iv) receivables that result from transactions not related to resident care.
- (4) Cash and cash equivalents exclude:
  - (i) funded depreciation recognized by the Division.
  - (ii) restricted grants and gifts.
- (5) Interest income offset.
  - (i) Interest expense shall be reduced by realized investment income, except where such income is from:
    - (A) funded depreciation recognized by the Division pursuant to CMS-15,
    - (B) grants and gifts, whether restricted or unrestricted.
  - (ii) Only working capital interest expense shall be offset by interest income derived from working capital.
- (6) The provider must have a legal obligation to pay the interest.
- (c) "Proper" requires that:
  - (1) Interest be incurred at a rate not in excess of what a prudent buyer would have

had to pay in the money market existing at the time the loan was made.

- (2) Interest must be paid to a lender that is not a related party of the borrowing organization except as provided in paragraph (k).
- (d) Interest expense shall be included in property costs if the interest is necessary and proper and if it is incurred as a result of financing the acquisition of fixed assets related to resident care.
- (e) The date of such financing must be within 60 days of the date the asset is put in use, except for assets approved through the Certificate of Need process or approved by the Division under Subsection 4.11 of this rule. Allowable interest, on loans financed more than 60 days before or after the asset is put in use, will be included in Indirect Costs for the entire term of the loan.
- (f) Borrowings to finance asset additions cannot exceed the sum of the basis of the asset(s), determined in accordance with Subsections 4.5 and 4.7, and other costs allowed pursuant to paragraph (g) related to the borrowing. The limit on borrowings related to fixed assets is determined as follows:

Basis of the assets recognized by the Division, plus a proportionate share of other costs allowed pursuant to paragraph (g), or

the principal amount of the loan, whichever is the lower:

Less: The provider's cash and cash equivalents in excess of 60 days needs, per subparagraph (b)(2) of this subsection.

Equals: The limits on borrowings related to fixed assets.

(g) Other costs related to the acquisition of the assets may be included in loans where the interest is recognized by the Division. These

- eosts include bank finance charges, points and costs for legal and accounting fees, and discounts on debentures and letters of credit.
- (h) Necessary and proper interest expense on debt incurred other than for the acquisition of assets shall be recognized as working capital interest expense and included in Indirect Costs.
- (i) Application of Principal Payments.
  - (1) For loans entered into before a facility's 1998 fiscal year, principal payments shall be applied first to loan balances on allowable borrowings and second to non-allowable loan balances.
  - (2) For loans entered into during or after a facility's 1998 fiscal year, principal payments shall be applied to allowable and non-allowable loan balances on the ratio of each to the total amount of the loan.
- (j) Refinancing of indebtedness.
  - (1) The provider must demonstrate to the Division that the costs of refinancing will be less than the allowable costs of the current financing.
  - (2) Costs of refinancing must include accounting fees, legal fees and debt acquisition costs related to the refinancing.
  - (3) Material interest expense related to the original loan's unpaid interest charges, to the extent that it is included in the refinanced loan's principal, shall not be allowed.
  - (4) A principal balance in excess of the sum of the principal balance of the previous financing plus accounting fees, legal fees and debt acquisition costs shall be considered a working capital loan, subject to the cash needs test in subsection 4.4(b)(2), unless the provider demonstrates to the Division that the excess was for the acquisition of assets as set forth in (a) through (g).

- (k) Interest expense incurred as a result of transactions with a related party (or related parties) will be recognized if the expense would otherwise be allowable and if the following conditions are met:
  - (1) The interest expense relates to a first and/or second mortgage or to assets leased from a related party where the costs to the related party are recognized in lieu of rent.
- (2) The interest rate is no higher than the rate charged by lending institutions at the inception of the loan.
- (1) Interest is not allowable with respect to any capital expenditure in property, plant and equipment related to resident care which requires approval, if the necessary approval has not been granted.
- (m) Interest on loans that do not include reasonable and ordinary principle repayments in the debt service payments shall not be allowable except to the extent that it would have been incurred pursuant to a standard amortization schedule for a term equivalent to the useful life of the asset.

#### 4.5 Basis of Property, Plant and Equipment

- (a) The basis of a donated asset is the fair market value.
- (b) The basis of other assets that are owned by a provider and used in providing resident eare shall generally be the lower of cost or fair market value. Specific exceptions are addressed elsewhere in this rule. Cost includes:
- (1) purchase price,
- (2) sales tax,
- (3) costs to prepare the asset for its intended use, such as, but not limited to, costs of shipping, handling, installation, architectural fees, consulting and legal fees.

- (e) The basis of assets constructed by the provider to provide resident care shall be determined from the construction costs which include:
  - (1) all direct costs, including, but not limited to, salaries and wages, the related payroll taxes and fringe benefits, purchase price of materials, sales tax, costs of shipping, handling and installation, costs for permits, architectural fees, consulting fees and legal fees.
  - (2) indirect costs related to the construction of the asset.
  - (3) interest costs related to capital indebtedness used to finance the construction of the asset and prepare it for its intended use.
- (d) The basis of betterments or improvements, if they extend the useful life of an asset two or more years or significantly increase the productivity of an asset are costs as set forth in paragraphs (b) and (c) above.
- (e) Any asset that has a basis of \$2,000 or more and an estimated useful life of two or more years must be capitalized and depreciated in accordance with Subsection 4.6. Groups of assets with the majority of assets in the group valued at \$300 or more and a useful life of two years or more must also be capitalized and depreciated in accordance with Subsection 4.6. Assets or groups of assets with a basis lower than \$2,000 may be expensed or depreciated at the provider's election.
- (f) The gain on a transfer of an asset to a related party shall be calculated as follows: the fair market value of the asset, less the net book value will be the gain irrespective of the of the amount paid to the facility for the asset. This gain will be offset against property and related costs.

# 4.6 Depreciation and Amortization of Property, Plant and Equipment

- (a) Costs for depreciation and amortization must be based on property records sufficient in detail to identify specific assets.
- (b) Depreciation and amortization must be computed on the straight-line method.
- (c) The depreciable basis of an asset shall be the basis established according to Subsections 4.5 and 4.7, not of any salvage value.
- (d) The estimated useful life of an asset shall be determined by the Division as follows:
  - (1) The recommended useful life is the number of years listed in the most recent edition of Estimated Useful Lives of Depreciable Hospital Assets, published by the American Hospital Association.
  - (2) Leasehold improvements may be amortized over the term of an arms length lease, including renewal period, if such a lease term is shorter than the estimated useful life of the asset.

# 4.7 Change in Ownership of Depreciable Assets - Sales of Facilities

- (a) A change of ownership will be recognized when the following criteria have been met:
  - (1) The change of ownership did not occur between related parties, except for transactions that meet the criteria in subparagraph (2).
  - (2) The transaction takes place between family members and meets the following conditions:
    - (i) The Division shall be notified at least two years before the sale. The notice shall include a description of the terms and conditions of the sale and be accompanied by a current appraisal of the facility being sold.

- (ii) The buyer shall demonstrate the capacity to manage and/or administer the facility; or if the buyer is to be an absentee owner, the buyer shall demonstrate that there will be sufficient capable staff to operate the facility according to standards prescribed by state and federal law.
- (iii) The seller shall not maintain full time employment with the facility, except for a transition period which shall not be longer than one year during which the seller may provide reasonable consultation to assure a smooth transition.
- (iv) A sale of the facility shall not have occurred between any members of the same family within the previous 12 years.
- (v) For the purposes of this subsection, family members shall include spouses, parents, grandparents, ehildren, grandehildren, brothers, sisters, spouses of parents, grandparents, ehildren, grandehildren, brothers and sisters, aunts, uncles, nieces and nephews, or such other familial relationships as the Director may reasonably approve in the circumstances of the transaction.
- (3) The change of ownership was made for reasonable consideration.
- (4) The change of ownership was a bona fide transfer of all the powers and indicia of ownership.
- (5) The change in ownership is in substance the sale of the assets or stock of the facility and not a method of financing.
- (i) If the transferor and the transferee enter into a financing agreement, the agreement must be constructed to effect a complete change of ownership. The Division shall determine if the agreement does in substance effect a complete change of ownership and the Division shall monitor the compliance with the agreement.

- (ii) Where, subsequent to a change of ownership, the transferor forgives or reduces the debt of the transferee, the amount of the forgiveness or reduction shall be retroactively applied to the acquisition or basis of the asset as determined by the Division.
- (6) The buyer shall demonstrate to the satisfaction of the Division that all obligations to the State of Vermont arising out of the transaction have been satisfied.
- (7) For rate setting purposes, the transfer of stock or shares shall not be recognized as a change in ownership in the following eircumstances:
  - (i) the transferred stock or shares are those of a publicly traded corporation.
  - (ii) the transfer was made solely as a method of financing (not as a method of transferring management or control) and the number of shares transferred does not exceed 25 percent of the total number of shares in any one class of stock.
- (b) Where the Division recognizes the change in ownership of an asset, the basis of the assets for the new owner shall be determined as follows:
  - (1) If the seller did not own the assets during the entire twelve year period immediately preceding the change in ownership or if the seller's facility did not receive Vermont Medicaid reimbursement during the entire twelve year period immediately preceding the change in ownership, the depreciable cost basis of the transferred asset for the new owner shall be the lowest of:
    - (i) the fair market value of the assets,
    - (ii) the acquisition cost of the asset to the buyer,

- (iii) the original basis of the asset to the seller as recognized by the Division, less accumulated depreciation.
- (2) If the seller owned the assets during the entire twelve year period immediately preceding the change in ownership and if the seller's facility received Vermont Medicaid reimbursement during the entire twelve year period immediately preceding the change in ownership, the depreciable cost basis of the transferred fixed equipment and building improvements for individual assets having an original useful life of at least 20 years in agreement with the useful life assigned in the American Hospital Association guidelines, the depreciable cost basis of land improvements, the depreciable cost basis of buildings and the cost basis of land for the new owner shall be the lowest of:
  - (i) the fair market value of the assets,
  - (ii) the acquisition cost of the asset to the buyer,
  - (iii) the amount determined by the revaluation of the asset. An asset is revalued by increasing the original basis of the asset to the seller, as recognized by the Division, by an annual percentage rate. The annual percentage rate will be limited to the lower of:
    - (A) One half the percentage increase in the Consumer Price Index (CPI) for All Urban consumers (United States City Average).
    - (B) One half the percentage change in an appropriate construction cost index as determined by the Division of Rate Setting, which change shall not be greater than one half of the percentage increase in the Dodge Construction index (or a reasonable proxy therefor) for the same period.

(3) If the seller owned the assets during the entire twelve year period immediately preceding the change in ownership and if the seller's facility received Vermont Medicaid reimbursement during the entire twelve year period immediately preceding the change in ownership, the depreciable cost basis of individual assets categorized as building improvements and fixed equipment with an original useful life of less than 20 years, in agreement with the useful life assigned in the American Hospital Association guidelines, shall be the seller's net book value and shall be depreciated over a useful life of seven years.

(4) If the seller owned the assets during the entire twelve year period immediately preceding the change in ownership and if the seller's facility received Vermont Medicaid reimbursement during the entire twelve year period immediately preceding the change in ownership, the depreciable cost basis of moveable equipment and vehicles shall be the seller's net book value and shall be depreciated over a useful life of ten years.

#### 4.8 [Repealed]

# 4.9 Leasing Arrangements for Property, Plant and Equipment

Leasing arrangements for property, plant and equipment must meet the following conditions:

- (a) Rent expense on facilities and equipment leased from a related organization will be limited to the Medicaid allowable interest, depreciation, insurance and taxes incurred for the year under review, or the price of comparable services or facilities purchased elsewhere, whichever is lower.
- (b) Rental or leasing charges, including sale and leaseback agreements for property, plant and equipment to be included in allowable costs cannot exceed the amount which the provider would have included in allowable

costs had it purchased or retained legal title to the asset, such as interest on mortgage, taxes, insurance and depreciation.

#### 4.10 Funding of Depreciation

- (a) Funding of depreciation is not required, but it is strongly recommended that providers use this mechanism as a means of conserving funds for replacement of depreciable assets, and coordinate their planning of capital expenditures with area wide planning of community and state agencies. As an incentive for funding, investment income on funded depreciation will not be treated as a reduction of allowable interest expense.
- (b) To the extent that the provider fails to retain sufficient working capital or sufficient resources to support operations, before making deposits in a funded depreciation account, the deposits will not be recognized as funded depreciation.
- (e) To the extent that funded depreciation in the cost reporting period under consideration is used for purposes other than nursing facility asset acquisition, interest income on those sums will be offset against interest expense not only in the current period, but the Division may reopen settled cost reports for previous periods to revise funded depreciation and allowable interest expense. However, with the prior approval of the Division, under appropriate conditions, some or all of a provider's funded depreciation may be used as follows without triggering an interest income offset:
  - (1) to convert existing nursing home beds to residential care or assisted living, or
  - (2) when more economic, for new construction of residential care or assisted living units with a reduction in licensed nursing home beds.
- (d) All relevant provisions of CMS 15 shall be followed, except as noted below:

- (1) Replacement reserves. Some lending institutions require funds to be set aside periodically for replacement of fixed assets. The periodic amounts set aside for this purpose are not allowable costs in the period expended, but will be allowed when withdrawn and utilized either through depreciation or expense after considering the usage of these funds. Since the replacement reserves are essentially the same as funded depreciation the same regulations regarding interest will apply.
- (2) If a facility is leased from an unrelated party and the ownership of the reserve rests with the lessor, then the replacement reserve payment becomes part of the lease payment and is considered an allowable cost in the year expended. If the lessee is allowed to use this replacement reserve for the replacement of the lessee's assets, lessee shall not be allowed to depreciate the assets purchased.
- (e) The provider must maintain appropriate documentation to support the funded depreciation account and income earned thereon to be eligible for relief from the investment income offset.

#### 4.11 Adjustments for Large Asset Acquisitions and Changes of Ownership

#### (a) Large Asset Acquisitions

- (1) A provider may apply to the Division for an adjustment to the property and related component of the rate for *individual* eapital expenditures determined to be necessary and reasonable. No application for a rate adjustment should be made if the change to the rate would be smaller than one half of one percent of the facility's rate in effect at the time the application is made. Interest expense related to these assets, provided it is necessary and reasonable, shall be included in calculating the adjustment.
- (2) In the event that approval is granted by the Division, the adjustment will be made

effective from the first day of the quarter after the filing date of the written notice, following the date of the final order on the application, or following the date the asset is actually put into service, whichever is the latest.

#### (b) Changes of Ownership

- (1) Application shall also be made under this subsection, no later than 30 days after the execution of a purchase and sale agreement or other binding contract, or the receipt of a Certificate of Need pursuant to 18 V.S.A. §9434, for changes in basis resulting from a change in ownership of depreciable assets recognized by the Division pursuant to Subsection 4.7. The Division may make related adjustments to the Property and Related rate component.
- (2) Adjustments to the Property and Related rate component resulting from a change in ownership of depreciable assets shall be effective from the first day of the month following the date of sale.
- (e) Except in circumstances determined by the Division to constitute an emergency precluding a 60 day notice period, a provider applying for an adjustment pursuant to this subsection is required to give 60 days written notice to the Division prior to the purchase of the asset. Such applications shall be exempt from the materiality test set out in subsection 8.7(b), but are subject to the other provisions of subsection 8.7. The burden is on the provider to document all information applicable to this adjustment and to demonstrate that any costs to be incurred are necessary and reasonable. When applicable, such documentation shall include the Certificate of Need application and all supporting financial information. The Division shall review the application and issue draft findings approving. denying, or proposing modifications to the adjustment applied for within 60 days of receipt of all information required.

#### 4.12 [Repealed]

#### 4.13 Advertising Expenses

The reasonable and necessary expense of newspaper or other public media advertisement for the purpose of securing necessary employees is an allowable cost. No other advertising expenses are allowed.

#### 4.14 Barber and Beauty Service Costs

The direct costs of barber and beauty services are not allowable for purposes of Medicaid reimbursement. However, the fixed costs for space and equipment related to providing these services and overhead associated with billing for these services are allowable.

### 4.15 Bad Debt, Charity and Courtesy Allowances

Bad debts, charity and courtesy allowances are deductions from revenues and are not to be included in allowable costs.

#### 4.16 Child Day Care

Reasonable and necessary costs incurred for the provision of day care services to children of employees performing resident related functions will be allowable. Costs will be adjusted by any revenues received for the provision of care provided to employees' children. The direct and indirect expenses related to providing these services to non-employee children are not an allowable expense. Costs must be accumulated in a separate cost center. Revenues earned from providing day care must be identified for employees and non-employees in a separate account.

#### **4.17 Community Service Activities**

As an incentive for nursing home providers to furnish needed services (i.e., meals on wheels, adult day and certain respite eare, etc.) to local communities, with the prior permission of the Division, only direct identifiable incremental costs will be adjusted (i.e., food, direct labor and fringe benefits, transportation). Overhead costs will

not be apportioned for adjustment unless there is a significant expansion to a program resulting from community service involvement. The provider must maintain auditable records for all incremental direct costs associated with providing a community service.

#### 4.18 Dental Services

Costs incurred for services performed in connection with the care, treatment, filling, removal, or replacement of teeth or structures directly supporting teeth will not be allowed for the purposes of calculating the per diem rate. Dental services for Medicaid eligible individuals are covered pursuant to the Medicaid Covered Services Rules. However, the fixed costs for space and equipment related to providing these services and overhead associated with billing for these services may be allowable.

#### 4.19 Legal Costs

Necessary, ordinary, and reasonable legal fees incurred for resident-related activities will be allowable.

#### **4.20 Litigation and Settlement Costs**

#### (a) Civil and criminal litigation -

- (1) General Rule. Attorney fees and other expenses incurred in conjunction with litigation will be recognized only to the extent that the costs are related to resident care, that the provider prevails, and that the costs are not covered by insurance.
- (2) Settlements. In instances, where a matter is settled before judgment (whether or not a lawsuit has been commenced), one half the costs, including attorney fees, settlement award, and other expenses, relating to the matter will be recognized to the extent that the costs are related to resident care and are not covered by insurance.

- (3) Costs related to criminal or professional practice matters are not allowable.
- (b) Challenges to decisions of the Division Attorney fees and other expenses incurred by a provider in challenging decisions of the Division will be allowed based on the extent to which the provider prevails as determined on the ratio of total dollars at issue in the case to the total dollars awarded to the provider.
- (c) All costs recognized pursuant to this subsection shall be subject to the non-recurring costs provision in subsection 4.3(a)(2) or subsection 6.4.

#### **4.21 Motor Vehicle Allowance**

Cost of operation of a motor vehicle necessary to meet the facility needs is an allowable cost. Where the vehicle is used for personal and business purposes, the portion of vehicle costs associated with personal use will not be allowed. If the provider does not document personal use and business use under a pre approved method, DRS reserves the right to disallow all vehicle costs in question. All costs in excess of the cost of a similar size mid-price vehicle are not allowable.

#### 4.22 Non Competition Agreement Costs

Amounts paid to the seller of an on going facility by the purchaser for an agreement not to compete are considered capital expenditures. The amortized costs for such agreements are not allowable.

## 4.23 Compensation of Owners, Operators, or their Relatives

(a) Facilities which have a full time (40 hours per week minimum) administrator and/or assistant administrator, will not be allowed compensation for owners, operators, or their relatives who claim to provide some or all of the administrative functions required to operate the facility efficiently except in limited and special circumstances such as

- those listed in paragraph (b) of this subsection.
- (b) The factors to be evaluated by the Division in determining the amount allowable for owner's compensation shall include, but not limited to the following:
  - (1) All applicable Medicare policies identified in CMS-15.
  - (2) The unduplicated functions actually performed, as described by the provider on the Medicaid cost report.
  - (3) The hours actually worked and the number of employees supervised, as reported on the cost report.
- (c) For any facility fiscal year, the maximum allowable salary for an owner administrator shall be equal to 110 percent of the average of all reported administrator salaries for Vermont nursing facilities participating in the Medicaid program for that facility fiscal year.

### 4.24 Management Fees and Home Office Costs

- (a) Management fees, home office costs and other costs incurred by a nursing facility for similar services provided by other entities shall be included in the Indirect Cost category. These costs are subject to the provisions for allowable costs, allocation of costs and related party transactions contained in these rules and shall include property and related costs incurred for the management company. These costs are allowable only if such costs would be allowable if a nursing facility provided the services for itself.
- (b) Allowable costs shall be limited to five percent of the total net allowable costs less reported management fees, home office, or other costs, as defined in this subsection.

#### 4.25 Membership Dues

Reasonable and necessary membership dues, including any portions used for lobbying

activities, shall be considered Medicaid allowable costs, provided the organization's function and purpose are directly related to providing resident care.

#### 4.26 Post-Retirement Benefits

The allowability of costs of certain benefits which may be available to retired personnel shall be governed by CMS 15, except that all such costs shall be included in fringe benefits and shall be allocated accordingly.

#### 4.27 Public Relations

Costs incurred for services, activities and events that are determined by the Division to be for public relations purposes will not be allowed.

#### 4.28 Related Party

Expenses otherwise allowable shall not be included for purposes of determining a prospective rate where such expenses are paid to a related party unless the provider identifies any such related party and the expenses attributable to it and demonstrates that such expenses do not exceed the lower of the cost to the related party or the price of comparable services, facilities or supplies that could be purchased elsewhere. The Division may request either the provider or the related party, or both, to submit information, books and records relating to such expenses for the purpose of determining their allowability.

#### 4.29 Revenues

Where a facility reports operating and nonoperating revenues related to goods or services, the costs to which the revenues correspond are not allowable. If the specific costs cannot be identified, the revenues shall be deducted from the most appropriate costs. If the revenues are more than such costs, the deduction shall be equal to such costs.

#### 4.30 Travel/Entertainment Costs

Only reasonable and necessary costs of meals, lodging, transportation and incidentals incurred for purposes related to resident care will be allowed. All costs determined to be for the pleasure and convenience of the provider or providers' representatives will not be allowed.

#### 4.31 Transportation Costs

(a) Costs of transportation incurred, other than ambulance services for emergency transportation or transportation home from a nursing facility covered pursuant to the Medicaid Covered Services Rules, that are necessary and reasonable for the care of residents are allowable. Such costs shall include depreciation of utility vehicles, mileage reimbursement to employees for the use of their vehicles to provide transportation for residents, and any contractual arrangements for providing such transportation. Such costs shall not be separately billed for individual residents.

(b) Transportation costs related to residents receiving kidney dialysis shall be reported in the Ancillary cost category, pursuant to subsection 6.7(a)(5).

#### **4.32 Services Directly Billable**

Allowable costs shall not include the cost of services to individual residents which are ordinarily billable directly to Medicaid irrespective of whether such costs are payable by Medicaid.

#### 5 REIMBURSEMENT STANDARDS

# 5.1 Prospective Case-Mix Reimbursement System

(a) In general, these rules set out incentives to control costs and Medicaid outlays, while promoting access to services and quality of care.

- (b) Case mix reimbursement takes into account the fact that some residents are more costly to care for than others. Thus the system requires:
  - (1) the assessment of residents on a form prescribed by the Director of the Division of Licensing and Protection;
  - (2) a means to classify residents into groups which are similar in costs, known as RUG IV (48 group version) and
  - (3) a weighting system which quantifies the relative costliness of caring for different classes of residents to determine the average case mix score.
- (e) Per diem rates shall be prospectively determined for the rate year based on the allowable operating costs of a facility in a Base Year, plus property and related and ancillary costs from the most recently settled cost report, calculated as described in Subsection 9.2.

### 5.2 Retroactive Adjustments to Prospective Rates

- (a) In general, a final rate may not be adjusted retroactively.
- (b) The Division may retroactively revise a final rate under the following conditions:
  - (1) as an adjustment pursuant to Sections 8 and 10;
  - (2) in response to a decision by the Secretary pursuant to Subsection 15.5 or to an order of a court of competent jurisdiction, whether or not that order is the result of a decision on the merits, or as the result of a settlement pursuant to Subsection 15.8;
  - (3) for mechanical computation or typographical errors;

- (4) for a terminating facility or a facility in receivership, pursuant to Subsections 5.10, 8.3, and 10.2;
- (5) as a result of revised findings resulting from the reopening of a settled cost report pursuant to Subsection 3.5;
- (6) in those cases where a rate includes payment for Ancillary services and the provider subsequently arranges for another Medicaid provider to provide and bill directly for these services;
- (7) recovery of overpayments, or other adjustments as required by law or duly promulgated regulation;
- (8) when a special rate is revised pursuant to subsection 14.1(e)(2) or
- (9) when revisions of final rates are necessary to pass the upper limits test in 42 C.F.R. §447.272.

#### 5.3 Lower of Rate or Charges

- (a) At no time shall a facility's Medicaid per diem rate exceed the provider's average customary charges to the general public for nursing facility services in semi-private rooms at the beginning of the calendar quarter. In this subsection, "charges" shall mean the amount actually required to be paid by or on behalf of a resident (other than by Medicaid, Medicare Part A or the Department of Veterans Affairs) and shall take into account any discounts or contractual allowances.
- (b) It is the duty of the provider to notify the Division within 10 days of any change in its charges.
- (c) Rates limited pursuant to paragraph (a) shall be revised to reflect changes in the provider's average customary charges to the general public effective on the latest of the following:

- (1) the first day of the month in which the change to the provider's charges is made if the changes is effective on the first day of the month.
- (2) the first day of the quarter after the effective date of the change to the provider's charges if the change to the provider's charges is not effective on the first day of the quarter, or
- (3) the first day of the following quarter after the receipt by the Division of notification of the change pursuant to paragraph (b).

#### 5.4 Interim Rates

- (a) The Division may set interim rates for any or all facilities. The notice of an interim rate is not a final order of the Division and is not subject to review or appeal pursuant to any provision of these rules or 33 V.S.A. §909.
- (b) Any overpayments or underpayments resulting from the difference between the interim and final rates will be either refunded by the provider or paid to the provider.

#### 5.5 Upper Payment Limits

- (a) Aggregate payments to nursing facilities pursuant to these rules may not exceed the limits established for such payment in 42 C.F.R. §447.272.
- (b) If the Division projects that Medicaid payments to nursing facilities in the aggregate will exceed the Medicare upper limit, the Division shall adopt a rule limiting some or all of the payments to providers to the level that would reduce the aggregate payments to the Medicare upper limit.

#### 5.6 Base Year

(a) A Base Year shall be a calendar year, January through December.

- (b) All costs shall be rebased on July 1, 2007. Subsequent rebasing for Nursing Care costs shall occur two years after the last rebase of such costs. All costs shall be rebased no less frequently than every four years.
- (e) For the purposes of rebasing, the Director may require individual facilities to file special cost reports covering the calendar year when this is not the facility's fiscal year or the Division may use the facility's fiscal year cost report adjusted by the inflation factors in subsection 5.8 to the Base Year. The Director may require audited financial statements for the special cost reporting period. The costs of preparing the special cost report and audited financial statements are the responsibility of the provider, without special reimbursement; however, for reporting purposes, these costs are allowable.
- (d) The determination of a Base Year shall be subject of a notice of practices and procedures pursuant to Subsection 1.8(d) of these rules.

#### 5.7 Occupancy Level

- (a) A facility should maintain an annual average level of occupancy at a minimum of 90 percent of the licensed bed capacity.
- (b) For facilities with less than 90 percent occupancy, the number of total resident days at 90 percent of licensed capacity shall be used, pursuant to section 7, in determining the per diem rate for all categories except the Nursing Care and Ancillary categories.
- (c) The 90 percent minimum occupancy provision in paragraph (b) shall be waived for facilities with 20 or fewer beds or terminating facilities pursuant to Subsection 5.10, and when appropriate, for facilities operating under a receivership pursuant to Subsection 8.3.
- (d) Decreasing the Number of Licensed Beds
  —For any facility that operated at less than 90
  percent occupancy during the period used as
  the cost basis for any rate component subject

to subsection (b) which subsequently reduces the number of licensed beds, the minimum occupancy shall be calculated based on the number of the facility's licensed beds on the first day of the quarter after the facility notifies the Division of such reduction.

#### **5.8 Inflation Factors**

The Director shall use the most recent publication of the Health Care Cost Service available June 1 in the calculation of inflation factors, whether for rebase inflation calculations or annual inflation calculations. Different inflation factors are used to adjust different rate components. Subcomponents of each inflation factor are weighted in proportion to the percentage of actual allowable costs incurred by Vermont facilities for specific subcomponents of the relevant cost component. For example, if a cost in the Nursing Care cost component is 83.4 percent attributable to salaries and wages and 16.6 percent attributable to employee benefits, the weights for the two subcomponents of the Nursing Care inflation factor shall be 0.834 and 0.166 respectively. The weights for each inflation factor shall be recalculated no less frequently than each time the relevant cost category is rebased.

(a) The Nursing Care rate component shall be adjusted by an inflation factor that uses two price indexes to account for estimated economic trends with respect to two subcomponents of nursing costs: wages and salaries, and benefits. The price indexes for each subcomponent are the wages and salaries portion of the Health Care Cost Service NHMB, and the employee benefits portion of the NHMB, respectively. An additional adjustment of one percentage point shall be made for every 12 month period, prorated for fractions thereof, from the midpoint of the base year to the midpoint of the rate year.

(b) The Resident Care Rate Component shall be adjusted by an inflation factor that uses four price indexes to account for estimated economic trends with respect to the subcomponents of Resident Care costs: wages and salaries, employee benefits, utilities, and food and all other Resident care costs. The price indexes for each subcomponent are: the wages and salaries portion of the Health Care Cost Service NHMB, the employee benefits portion of the NHMB, and the food portion of the NHMB respectively.

(e) The Indirect rate component shall be adjusted by an inflation factor that uses three price indexes to account for estimated economic trends with respect to three subcomponents of Indirect costs: wages and salaries, employee benefits, and all other indirect costs. The price indexes for each subcomponent are: the wages and salaries portion of the Health-Care Cost Service NHMB, the employee benefits portion of the NHMB and the NECPI-U (all items), respectively.

(d) The Director of Nursing rate component shall be adjusted by an inflation factor that uses two price indexes to account for estimated economic trends with respect to two subcomponents of Director of Nursing costs: wages and salaries and employee benefits. The price indexes for each subcomponent are: the wages and salaries portion of the Health Care Cost Service NHMB, and the employee benefits portion of the NHMB, respectively.

(e) Pursuant to Subsection 1.8(d), the Division shall issue a description of the practices and procedures used to calculate and apply the Inflation Factors.

#### 5.9 Costs for New Facilities

(a) For facilities that are newly constructed, newly operated as nursing facilities, or new to the Medicaid program, the prospective ease mix rate shall be determined based on budget cost reports submitted to the Division and the greater of the estimated resident days for the rate year or the resident days equal to 90 percent occupancy of all beds used or

intended to be used for resident care at any time within the budget cost reporting period. This rate shall remain in effect no longer than one year from the effective date of the new rate. The principles on allowability of costs and existing limits in Sections 4 and 7 shall apply.

- (b) The costs reported in the budget cost report shall not exceed reasonable budget projections (adjusted for inflation and changes in interest rates as necessary) submitted in connection with the Certificate of Need.
- (c) Property and related costs included in the rate shall be consistent with the property and related costs in the approved Certificate of Need.
- (d) At the end of the first year of operation, the prospective case mix rate shall be revised based on the provider's actual allowable costs as reported in its annual cost report filed pursuant to subsection 3.2 for its first full fiscal year of operation.

#### 5.10 Costs for Terminating Facilities

- (a) When a nursing facility plans to discontinue all or part of its operation, the Division may adjust its rate so as to ensure the protection of the residents of the facility.
- (b) A facility applying for an adjustment to its rate pursuant to this subsection must have a transfer plan approved by the Department of Disabilities, Aging and Independent Living, a copy of which shall be supplied to the Division.
- (c) An application under this subsection shall be made on a form prescribed by the Director and shall be accompanied by a financial plan demonstrating how the provider will meet its obligations set out in the approved transfer plan.
- (d) In approving such an application the Division may waive the minimum occupancy requirements in Subsection 5.7, the

limitations on costs in Section 7, or make such other reasonable adjustments to the facility's reimbursement rate as shall be appropriate in the circumstances. The adjustments made under this subsection shall remain in effect for a period not to exceed six months.

#### 6 BASE YEAR COST CATEGORIES FOR NURSING FACILITIES

#### 6.1 General

In the case mix system of reimbursement, allowable costs are grouped into cost categories. The accounts to be used for each cost category shall be prescribed by the Director. The Base Year costs shall be grouped into the following cost categories:

#### **6.2 Nursing Care Costs**

- (a) Allowable costs for the Nursing Care component of the rate shall include actual costs of licensed personnel providing direct resident care, which are required to meet federal and state laws as follows:
  - (1) registered nurses,
  - (2) licensed practical nurses,
  - (3) certified or licensed nurse aides, including wages related to initial and ongoing nurse aide training as required by OBRA,
  - (4) contract nursing,
  - (5) the MDS coordinator,
  - (6) fringe benefits, including child day care.
- (b) Costs of bedmakers, geriatric aides, transportation aides, paid feeding/dining assistants, ward clerks, medical records librarians and other unlicensed staff will not be considered nursing costs. The salary and related benefits of the position of Director of Nursing shall be excluded from the calculation of allowable nursing costs and shall be reimbursed separately.

#### 6.3 Resident Care Costs

Allowable costs for the Resident Care component of the rate shall include reasonable costs associated with expenses related to direct care. The following are Resident Care costs:

- (a) food, vitamins and food supplements,
- (b) utilities, including heat, electricity, sewer and water, garbage and liquid propane gas,
- (e) activities personnel, including recreational therapy and direct activity supplies,
- (d) Medical Director, Pharmacy Consultant, Geriatric Consultant, and Psychological/psychiatric Consultant,
- (e) counseling personnel, chaplains, art therapists and volunteer stipends,
- (f) social service worker,
- (g) employee physicals,
- (h) wages for paid feeding/dining assistants only for those hours that they are actually engaged in assisting residents with eating,
- (i) fringe benefits, including child day care,
- (j) such other items as the Director may prescribe by a practice and procedure issued pursuant to subsection 1.8(d).

#### **6.4 Indirect Costs**

- (a) Allowable costs for the indirect component of the rate shall include costs reported in the following functional cost centers on the facility's cost report, including those extracted from a facility's cost report or the cost report of an affiliated hospital or institution.
  - (1) fiscal services,
  - (2) administrative services and professional fees,
  - (3) plant operation and maintenance,
  - (4) grounds,
  - (5) security,
  - (6) laundry and linen,
  - (7) housekeeping,
  - (8) medical records.
  - (9) cafeteria,
  - (10) seminars, conferences and other inservice training (except tuition for college

- eredit in a discipline related to the individual staff member's employment or costs of obtaining a GED which shall be treated as fringe benefits).
- (11) dietary excluding food,
- (12) motor vehicle,
- (13) clerical, including ward clerks,
- (14) transportation (excluding depreciation),
- (15) insurances (director and officer liability, comprehensive liability, bond indemnity, malpractice, premise liability, motor vehicle, and any other costs of insurance incurred or required in the care of residents that has not been specifically addressed elsewhere),
- (16) office supplies/telephone,
- (17) conventions and meetings,
- (18) EDP bookkeeping/payroll,
- (19) fringe benefits including child day care.
- (b) All expenses not specified for inclusion in another cost category pursuant to these rules shall be included in the Indirect Costs category, unless the Director at her/his discretion specifies otherwise in the instructions to the cost report, the chart of accounts, or by the issuance of a practice and procedure. For nursing facility rate setting, the costs of prescription drugs are not allowable.

#### 6.5 Director of Nursing

Allowable costs associated with the position of Director of Nursing shall include reasonable salary for one position and associated fringe benefits, including child day care.

#### 6.6 Property and Related

- (a) The following are Property and Related costs:
- (1) depreciation on buildings and fixed equipment, major movable equipment, minor equipment, computers, motor vehicle, land improvements, and

amortization of leasehold improvements and capital leases.

- (2) interest on capital indebtedness,
- (3) real estate leases and rents,
- (4) real estate/property taxes,
- (5) all equipment irrespective of whether it is capitalized, expensed, or rented,
- (6) fire and casualty insurance,
- (7) amortization of mortgage acquisition costs.
- (b) For a change in services, facility, or a new health care project with projected property and related costs of \$250,000 or more, providers shall give written notice to the Division no less than 60 days before the commencement of the project. Such notice shall include a detailed description of the project and detailed estimates of the costs.

#### 6.7 Ancillaries

- (a) The following are ancillary costs:
  - (1) All physical, speech, occupational, respiratory, and IV therapy services and therapy supplies (excluding oxygen) shall be considered ancillaries. Medicaid allowable costs shall be based on the cost-to-charge ratio for these services. These therapy services shall not be allowable for Medicaid reimbursement pursuant to this subsection unless:
    - (i) the services are provided pursuant to a physician's order,
    - (ii) the services are provided by a licensed therapist or other State certified or registered therapy assistant, or qualified IV professional, or other therapy aides,
    - (iii) the services are not reimbursable by the Medicare program, and
    - (iv) the provider records charges by payor class for all units of these services.
  - (2) Medical supplies, whether or not the provider customarily records charges.

- (i) Medical supplies shall include, but are not limited to: oxygen, disposable eatheters, catheters, colostomy bags, drainage equipment, trays and tubing.
- (ii) Medical supplies shall not include rented or purchased equipment, with the exception of rented or purchased oxygen concentrators, which shall be included in medical supplies.
- (3) Over the counter drugs. All drug costs may be disallowed for providers commingling the costs of prescription drugs (which are not allowable) and over the counter drugs.
- (4) Incontinent Supplies and Personal Care Items: including adult diapers, chux and other disposable pads, personal care items, such as toothpaste, shampoo, body powder, combs, brushes, etc.
- (5) Dialysis Transportation. The costs of transportation for Medicaid residents receiving kidney dialysis shall be included in the ancillary cost category. Allowable costs may include contract or other costs, but shall not include employee salaries or wages or cost associated with the use of provider owned vehicles.
- (6) Overhead costs related to ancillary services and supplies are included in ancillary costs.

#### (b) [Repealed]

# 7 CALCULATION OF COSTS, LIMITS AND RATE—COMPONENTS—FOR—NURSING FACILITIES

Base year costs, rates, and category limits are calculated pursuant to this section. The Medicaid per diem payment rate for each facility is calculated pursuant to Section 9.

#### 7.1 Calculation of Per Diem Costs

Per diem costs for each cost category, excluding the Nursing Care and Ancillary

eost categories, are calculated by dividing allowable costs for each case mix category by the greater of actual bed days of service rendered, including revenue generating hold/reserve days, or the number of resident days—computed—using—the—minimum occupancy at 90 percent of the licensed bed capacity during the cost period under review calculated pursuant to subsection 5.7.

#### 7.2 Nursing Care Component

#### (a) Case Mix Weights.

There are 48 case mix resident classes. Each case mix class has a specific case mix weight as follows:

Group	Case Mix	Description
Code	Weight	
ES3	3.00	Extensive Services
ES2	2.23	Extensive Services
ES1	2.22	Extensive Services
RAE	1.65	Rehabilitation
RAD	1.58	Rehabilitation
RAC	1.36	Rehabilitation
RAB	1.10	Rehabilitation
RAA	0.82	Rehabilitation
HE2	1.88	Special Care High
HE1	1.47	Special Care High
HD2	1.69	Special Care High
HDI	1.33	Special Care High
HC2	1.57	Special Care High
HC1	1.23	Special Care High
HB2	1.55	Special Care High
HB1	1.22	Special Care High
LE2	1.61	Special Care Low
LE1	1.26	Special Care Low
LD2	1.54	Special Care Low
LD1	1.21	Special Care Low
LC2	1.30	Special Care Low
LC1	1.02	Special Care Low
LB2	1.21	Special Care Low
LB1	0.95	Special Care Low
CE2	1.39	Clinically Complex
CE1	1.25	Clinically Complex
CD2	1.29	Clinically Complex
CD1	1.15	Clinically Complex
CC2	1.08	Clinically Complex
CC1	0.96	Clinically Complex
CB2	0.95	Clinically Complex
CB1	0.85	Clinically Complex
CA2	0.73	Clinically Complex

CA1	0.65	Clinically Complex
BB2	0.81	Behavioral Symptoms
		Plus Cognitive
		Performance
<del>BB1</del>	0.75	Behavioral Symptoms
		Plus Cognitive
		Performance
BA2	0.58	Behavioral Symptoms
		Plus Cognitive
		Performance
BAI	0.53	Behavioral Symptoms
		Plus Cognitive
		Performance
PE2	1.25	Reduced Physical
		Function
PE1	1.17	Reduced Physical
		<b>Function</b>
PD2	1.15	Reduced Physical
		Function .
PD1	1.06	Reduced Physical
		Function .
PC2	0.91	Reduced Physical
		<del>Function</del>
PC1	0.85	Reduced Physical
		<del>Function</del>
PB2	0.70	Reduced Physical
		<del>Function</del>
PB1	0.65	Reduced Physical
		Function
PA2	0.49	Reduced Physical
		Function
PA1	0.45	Reduced Physical
		Function

#### (b) Average case mix score

The Department of Disabilities, Aging and Independent Living's Division of Licensing and Protection shall compute each facility's average case mix score.

(1) The Division of Licensing and Protection shall periodically, but no less frequently than quarterly, certify to the Division of Rate Setting the average casemix score for those residents of each facility whose room and board (excluding resident share) is paid for solely by the Medicaid program.

(2) For the Base Year, the Division of Licensing and Protection shall certify the average case mix score for all residents.

- (e) Nursing Care cost per case mix point. Each facility's Nursing Care cost per case mix point will be calculated as follows:
  - (1) Using each facility's Base Year cost report, the total allowable Nursing Care costs shall be determined in accordance with Subsection 6.2.
  - (2) Each facility's Standardized Resident Days shall be computed by multiplying total Base Year resident days by that facility's average case mix score for all residents for the four quarters of the cost reporting period under review.
  - (3) The per diem nursing care cost per casemix point shall be computed by dividing total Nursing Care costs by the Base Year Standardized Resident Days for that Base Year.
- (d) Per diem limits on the Base Year allowable Nursing Care rate per case mix point:
  - (1) The Division shall array all nursing eare facilities' allowable Base Year per diem Nursing Care costs per case-mix point, excluding those for state nursing facilities and nursing facilities that are no longer in the Medicaid program at the time the limits are set, from low to high. These costs shall be limited to the cost at the ninetieth percentile, calculated using the percentile spreadsheet function.
  - (2) Each facility's Base Year Nursing Care rate per case mix point shall be the lesser of the limit in subparagraph (1) or the facility's allowable Nursing Care cost per case mix point.

#### 7.3 Resident Care Base Year Rate

Resident Care Base Year rates shall be computed as follows:

(a) Using each facility's Base Year cost report, the provider's Base Year total allowable Resident Care costs shall be

- determined in accordance with Subsection 6.3.
- (b) The Base Year per diem allowable Resident Care costs for each facility shall be calculated by dividing the Base Year total allowable Resident Care costs by total Base Year resident days.
- (e) The Division shall array all nursing facilities' Base Year per diem allowable Resident Care costs, excluding those for state nursing facilities and nursing facilities that are no longer in the Medicaid program at the time the limits are set, from low to high and identify the median.
- (d) The per diem limit shall be the median plus five percent.
- (e) Each facility's Base Year Resident Care per diem rate shall be the lesser of the limit set in paragraph (d) or the facility's Base Year per diem allowable Resident Care costs.

#### 7.4 Indirect Base Year Rate

Indirect Base Year rates shall be computed as follows:

- (a) Using each facility's Base Year cost report, each provider's Base Year total allowable Indirect costs shall be determined in accordance with Subsection 6.4.
- (b) The Base Year per diem allowable Indirect costs for each facility shall be calculated by dividing the Base Year total allowable Indirect costs by total Base Year resident days.
- (c) The Division shall array all nursing facilities' Base Year per diem allowable Indirect costs, excluding those for state nursing facilities and nursing facilities that are no longer in the Medicaid program at the time the limits are set, from low to high and identify the median.
- (d) The per diem limit shall be set as follows:

- (1) For special hospital-based nursing facilities, the limit shall be 137 percent of the median.
- (2) For all other privately owned nursing facilities, the limit shall be the median plus five percent.
- (e) Each provider's Base Year Indirect per diem rate shall be the lesser of the limit in paragraph (d) or the facility's Base Year per diem allowable Indirect costs.

#### 7.5 Director of Nursing Base Year Rate

The Director of Nursing Base Year per diem rates shall be computed as follows:

- (a) Using each facility's Base Year cost report, total allowable Base Year Director of Nursing costs shall be determined in accordance with Subsection 6.5.
- (b) Each facility's Base Year per diem allowable Director of Nursing costs shall be calculated by dividing the Base Year total allowable Director of Nursing costs by total Base Year resident days.
- (e) The Director of Nursing per diem rate shall be the facility's Base Year per diem allowable Director of Nursing costs calculated pursuant to this subsection.

#### 7.6 Ancillary Services Rate

- (a) The Ancillary per diem rate shall be computed as follows:
  - (1) Medicaid Ancillary costs shall be determined in accordance with subsection 6.7.
  - (2) Using each facility's most recently settled cost report, the per diem Ancillary rate shall be the sum of the following per diem costs calculated as follows:
    - (i) Costs for therapy services per diem, including IV therapy, shall be calculated by dividing allowable Medicaid costs by

- the number of related Medicaid resident days less Medicaid hold days.
- (ii) Dialysis transportation costs per diem shall be calculated by dividing the allowable costs for Vermont Medicaid residents by the number of Vermont Medicaid resident days less Vermont Medicaid hold days.
- (iii) Costs for medical supplies, over the eounter drugs, and incontinent supplies and personal care items per diem shall be ealculated by dividing allowable costs, by total resident days less hold days.
- (b) Any change to the Ancillary per diem rate shall be implemented at the time of the first quarterly case mix rate recalculation after the cost report is settled.

#### 7.7 Property and Related Per Diem

The Property and Related per diem rate shall be computed as follows:

- (a) Using each facility's most recently settled annual cost report, total allowable Property and Related costs shall be determined in accordance with Subsection 6.6.
- (b) Using each facility's most recently settled cost report, the per diem property and related costs shall be calculated by dividing allowable property and related costs by total resident days. Any change to the property and related per diem rate shall be implemented at the time of the first quarterly case mix rate recalculation after the cost report is settled.

#### 7.8 Limits Final

Once a final order has been issued for all facilities' Base Year cost reports, notwithstanding any subsequent changes to the cost report findings, resulting from a reopening, appeal, or other reason, the limits set pursuant to subsections 7.2(d)(2), 7.3(d), and 7.4(d) will not change until nursing home costs are rebased pursuant to 5.6(b),

except for annual adjustment by the inflation factors or a change in law necessitating such a change.

#### 8 ADJUSTMENTS TO RATES

#### 8.1 Change in Services

The Division, on application by a provider, may make an adjustment to the prospective case mix rate for additional costs which are directly related to:

- (a) a new health care project previously approved under the provisions of 18 V.S.A. §9434. Costs greater than those approved in the Certificate of Need (as adjusted for inflation) will not be considered when ealeulating such an adjustment,
- (b) a change in services, facility, or new health care project not covered under the provisions of 18 V.S.A. §9434, if such a change has previously been approved by the Division, or
- (c) with the prior approval of the Division, a reduction in the number of licensed beds.

#### 8.2 Change in Law

The Division may make or a provider may apply for an adjustment to a facility's prospective case mix rate for additional costs that are a necessary result of complying with changes in applicable federal and state laws, and regulations, or the orders of a State agency that specifically requires an increase in staff or other expenditures.

#### 8.3 Facilities in Receivership

(a) The Division, on application by a receiver appointed pursuant to state or federal law, may make an adjustment to the prospective case mix rate of a facility in receivership for the reasonable and necessary additional costs to the facility incurred during the receivership.

(b) On the termination of the receivership, the Division shall recalculate the prospective ease mix rate to eliminate this adjustment.

#### **8.4 Efficiency Measures**

The Division, on application by a provider, may make an adjustment to a prospective case-mix rate for additional costs which are directly related to the installation of energy conservation devices or the implementation of other efficiency measures, if they have been previously approved by the Division.

#### 8.5 Interest Rates

- (a) A provider may apply for an adjustment to the Property and Related rate, or the Division may initiate an adjustment if there are cumulative interest rate increases or decreases of more than one half of one percentage point because of existing financing agreements with a balloon payment or a refinancing clause that forces a mortgage to be refinanced at a different interest rate, or because of a variable rate of adjustable rate mortgages.
- (b) A provider with an interest rate adjustment shall notify the Division of any change in the interest rate within 10 days of its receipt of notice of that change. The Division may reseind all interest rate adjustments of any facility failing to file a timely notification pursuant to this subsection for a period of up to two years.

### 8.6 Emergencies and Unforeseeable Circumstances

- (a) The Division, on application by a provider, may make an adjustment to the prospective case mix rate under emergencies and unforeseeable circumstances, such as damage from fire or flood.
- (b) Providers must carry sufficient insurance to address adequately such circumstances.

# 8.7 Procedures and Requirements for Rate Adjustments

- (a) Application for a rate adjustment pursuant to this section should be made as follows. Approval of any application for a rate adjustment under this section is at the sole discretion of the Director.
- (b) Except for applications made pursuant to subsection 4.11, no application for a rate adjustment should be made if the change to the rate would be smaller than one percent of the rate in effect at the time.
- (c) Application for a Rate Adjustment shall be made on a form prescribed by the Director and filed with the Division and shall be accompanied by all documents and proofs determined necessary for the Division to make a decision.
- (d) The burden of proof is at all times on the provider to show that the costs for which the adjustment has been requested are reasonable, necessary and related to resident care.
- (e) The Division may grant or deny the Application, or make an adjustment modifying the provider's proposal. If the materials filed by the provider are inadequate to serve as a basis for a reasonable decision, the Division shall deny the Application, unless additional proofs are submitted.
- (f) The Division shall not be bound in considering other Applications, or in determining the allowability of reported costs, by any prior decision made on any Application under this section. Such decisions shall have no precedential value either for the applicant facility or for any other facility. Principles and decisions of general applicability shall be issued as a Division practice or procedure, pursuant to Section 1.8(d).
- (g) For adjustments requiring prior approval of the Division, such approval should be sought before the provider makes any

- commitment to expenditures. An Application for Prior Approval is subject to the same requirements as an Application for a Rate Adjustment under this section.
- (h) Rate adjustments made under this section shall be effective from the first day of the quarter following the date of the final order on the application or following the date the assets are actually put into service, whichever is the later, and may be continued, at the discretion of the Division, notwithstanding a general rebase of costs. Costs which are the basis for a continuing rate adjustment shall not be included in the cost categories used as the basis for the other rate components.
- (i) The Division may require an applicant for a rate adjustment under this section or under subsection 4.11 to file a budget cost report in support of its application.
- (j) When determined to be appropriate by the Division, a budget rate may be set for the facility according to the procedures in and subject to the provisions of subsection 5.9. Appropriate cases may include, but are not limited to, changes in the number of beds, an addition to the facility, or the replacement of existing property.
- (k) In calculating an adjustment under this section and subsection 4.11, the Division may take into account the effect of such changes on all the cost categories of the facility.
- (1) A revision may be made prospectively to a rate adjustment under this section and subsection 4.11 based on a "look back" which will be computed based on a provider's actual allowable costs.
- (m) In this subsection "additional costs" means the incremental costs of providing resident care directly and proximately caused by one of the events listed in this section or subsection 4.11. Increases in costs resulting from other causes will not be recognized. It is not intended that this section be used to effect a general rebase in a facility's costs.

# 8.8 Limitation on Availability of Rate Adjustments

Providers may not apply for a rate adjustment under this section for the sole reason that actual costs incurred by the facility exceed the rate of payment.

# 9 PRIVATE NURSING FACILITY AND STATE NURSING FACILITY RATES

The Medicaid per diem payment rates for nursing home services are calculated according to this section as follows:

#### 9.1 Nursing Facility Rate Components

The per diem rate of reimbursement consists of the following rate components:

- (a) Nursing Care
- (b) Resident Care
- (c) Indirect
- (d) Director of Nursing
- (e) Property and Related
- (f) Ancillaries
- (g) Adjustments (if any)

#### 9.2 Calculation of the Total Rate

The total per diem rate in effect for any nursing facility shall be the sum of the rates calculated for the components listed in Subsection 9.1, adjusted in accordance with the Inflation Factors, as described in Subsection 5.8.

# 9.3 Updating Rates for a Change in the Average Case-Mix Score

- (a) The Nursing Care rate component shall be updated quarterly, on the first day of January, April, July and October, for changes in the average case mix score of the facility's Medicaid residents.
- (b) The Nursing Care rate component and any part of a Section 8 adjustment that

reimburses nursing costs are updated for a change in the average case mix score for the facility's Medicaid residents. The update is calculated as follows:

- (1) The Nursing Care rate component (or rate adjustment) in the current rate of reimbursement for a facility is divided by the average case mix score used to determine the current Nursing Care rate component. This quotient is the current Nursing Care rate per case mix point.
- (2) The current Nursing Care rate component (or rate adjustment) per case-mix point is multiplied by the new average case mix score. This product is the new Nursing Care rate component (or rate adjustment).

#### 9.4 State Nursing Facilities

- (a) Notwithstanding any other provisions of these rules, payment rates for state nursing facilities shall be determined retrospectively by the Division based on the reasonable and necessary costs of providing those services as determined using the cost reporting and cost finding principles set out in sections 3 and 4 of these rules.
- (b) Until such time as the cost report is settled, the Division shall set an interim rate based on an estimate of the facility's costs and census for the rate year.
- (e) After reviewing the facility's cost report, the Division shall set a final rate for the fiscal year based on the facility's allowable costs. If there has been an under payment for the period the difference shall be paid to the facility. If there has been an overpayment the excess payments shall be recouped.
- (d) At no time shall the final rates paid to State nursing facilities exceed the upper limits established in 42 C.F.R. §447.272.

#### 9.5 Quality Incentives

Certain awards shall be made annually to facilities that provide a superior quality of care in an efficient and effective manner.

- (a) These payments will be based on:
  - (1) objective standards of quality, which shall include resident satisfaction surveys, to be determined by the Department of Disabilities, Aging and Independent Living, and
  - (2) objective standards of cost efficiency determined by the Division.
- (b) Supplemental payments will not be available under this subsection for any facility that does not participate in the statewide resident satisfaction survey program.
- (e) Supplemental payments shall be expended by the provider to enhance the quality of care provided to Medicaid eligible residents. In determining the nature of these expenditures, the provider shall consult with the facility's Resident Council.
- (d) The amount and method of distribution of the quality incentive payments shall be as follows:
  - (1) The quality incentive payments shall be made from a pool. The annual size of the pool shall be based on the amount of \$25,000 times the number of facilities meeting the award criteria, up to a maximum of five.
  - (2) The pool shall be distributed among the qualifying facilities, awarding each qualifying facility a share of the pool based on the ratio of its Medicaid days to the total Medicaid days for all the qualifying facilities.
- (e) Award Criteria

The following criteria will be applied to facility data up to March 31 each year to determine eligibility for the award to be presented in May. In order to be eligible for

the award, a facility must participate in the Vermont Medicaid program and meet all of the following criteria. All eligible facilities will be ranked according to their quality of care by the Department of Disabilities, Aging and Independent Living based on these basic quality criteria. The five facilities with the highest quality of care will receive an award. If, based on the basic criteria, there are ties which would cause more than five facilities to be equally qualified, the tied facilities will be ranked according to the efficiency criteria set out below in paragraph (6), to determine those facilities that will receive an award.

- (1) The most recent health survey report resulted in a score of five or less, no deficiency with a scope and severity greater than "D" level, with no more than two "D" level deficiencies in the general categories of Quality of Care, Quality of Life, or Resident Rights.
- (2) No substantiated complaints since the most recent survey and prior full health survey related to quality of care, quality of life, or residents' rights.
- (3) Participation in Advancing Excellence in America's Nursing Homes campaign.
- (4) Resident satisfaction survey results above the statewide average.
  - (5) Fire Safety deficiency score of 5 or less with scope and severity less than "E" in the most recent full survey.
- (6) The efficiency rankings shall be based upon the allowable costs per day from each facility's most recently settled Medicaid cost report. Cost per day will be calculated using actual resident days for the same fiscal period.

#### **10 EXTRAORDINARY FINANCIAL RELIEF**

10.1 Objective

In order to protect Medicaid recipients from the closing of a nursing facility in which they reside, this section establishes a process by which nursing homes that are in immediate danger of failure may seek extraordinary financial relief. This process does not create any entitlement to rates in excess of those required by 33 V.S.A. Chapter 9 or to any other form of relief.

#### 10.2 Nature of the Relief

- (a) Based on the individual circumstances of each case, the Director may recommend any of the following on such financial, managerial, quality, operational or other conditions as she or he shall find appropriate: a rate adjustment, an advance of Medicaid payments, other relief appropriate to the circumstances of the applicant, or no relief.
- (b) The Director's Recommendation shall be in writing and shall state the reasons for the Recommendation. The Recommendation shall be a public record.
- (c) The Recommendation shall be reviewed by the Secretary who shall make a Final Decision, which shall not be subject to administrative or judicial review.
- (d) In those cases where the Division determines that financial relief may be appropriate, such relief may be implemented on an interim basis pending a Final Decision by the Secretary. The interim financial relief shall be taken into account in the Division's Recommendation to the Secretary and in the Secretary's Final Decision.

# 10.3 Criteria to be Considered by the Division

(a) Before a provider may apply for extraordinary financial relief, its financial condition must be such that there is a substantial likelihood that it will be unable to continue in existence in the immediate future.

- (b) The following factors will be considered by the Director in making the Recommendation to the Secretary:
  - (1) the likelihood of the facility's closing without financial assistance,
  - (2) the inability of the applicant to pay bona fide debts.
  - (3) the potential availability of funds from related parties, parent corporations, or any other source.
  - (4) the ability to borrow funds on reasonable terms.
  - (5) the existence of payments or transfers for less than adequate consideration.
  - (6) the extent to which the applicant's financial distress is beyond the applicant's control.
  - (7) the extent to which the applicant can demonstrate that assistance would prevent, not merely postpone the closing of the facility.
  - (8) the extent to which the applicant's financial distress has been caused by a related party or organization,
  - (9) the quality of care provided at the facility,
  - (10) the continuing need for the facility's beds.
  - (11) the age and condition of the facility,
  - (12) other factors found by the Director to be material to the particular circumstances of the facility, and
  - (13) the ratio of individuals receiving care in a nursing facility to individuals receiving home and community based services in the county in which the facility is located.

#### 10.4 Procedure for Application

- (a) An Application for Extraordinary Financial Relief shall be filed with the Division according to procedures to be prescribed by the Director.
- (b) The Application shall be in writing and shall be accompanied by such documentation and proofs as the Director may prescribe. The burden of proof is at all times on the provider. If the materials filed by the provider are inadequate to serve as a basis for a reasoned recommendation, the Division shall deny the Application, unless additional proofs are submitted.
- (c) The Secretary shall not be bound in eonsidering other Applications by any prior decision made on any Application under this section. Such decisions shall have no precedential value either for the applicant facility or for any other facility.

### 11 PAYMENT FOR OUT-OF-STATE PROVIDERS

# 11.1 Long-Term Care Facilities Other Than Rehabilitation Centers

Payment for services, other than Rehabilitation Center services, provided to Vermont Medicaid residents in long term care facilities in another state shall be at the per diem rate established for Medicaid payment by the appropriate agency in that state. Payment of the per diem rate shall constitute full and final payment, and no retroactive settlements will be made.

#### 11.2 Rehabilitation Centers

- (a) Payment for prior authorized Rehabilitation Center services provided in nursing facilities located outside Vermont for the severely disabled, such as head injured or ventilator dependent people, will be made at the lowest of:
- (1) the amount charged; or

- (2) the Medicaid rate, including ancillaries as paid by at least one other state agency in CMS Region I.
- (b) Payment for Rehabilitation Center services which have not been prior authorized by the Commissioner of the Department of Vermont Health Access or a designee will be made according to Subsection 11.1.

#### 11.3 Pediatric Care

No Medicaid payments will be made for services provided to Vermont pediatric residents in out of state long term care facilities without the prior authorization of the Commissioner of the Department of Vermont Health Access.

#### 12 RATES FOR ICF/MRS

#### 12.1 Reasonable Cost Reimbursement

Intermediate Care Facilities for the Mentally Retarded (ICF/MRs) are paid according to Medicaid principles of reimbursement, pursuant to the Regulations Governing the Operation of Intermediate Care Facilities for the Mentally Retarded adopted by the Agency.

#### 12.2 Application of these Rules to ICF/MRS

The Division's Accounting Requirements (Section 2) and Financial Reporting (Section 3) shall apply to this program.

# 13 RATES FOR SWING BEDS AND OTHER LONG TERM CARE SERVICES IN HOSPITALS

Payment for swing bed and other long term care services provided by hospitals, pursuant to 42 U.S.C. §1396l(a), shall be made at a rate equal to the average rate per diem during the previous calendar year under the State Plan to nursing facilities located in the State of Vermont. Supplemental payments made pursuant to section 14 and subsection 9.5

shall not be included in the calculation of swing bed rates.

### 14 SPECIAL RATES FOR CERTAIN INDIVIDUAL RESIDENTS

#### 14.1 Availability of Special Rates for Individuals with Unique Physical Conditions

- (a) In rare and exceptional circumstances, a special rate shall be available for the care of an individual eligible for the Vermont Medicaid program whose unique physical conditions makes it otherwise extremely difficult to obtain appropriate long term care.
- (b) A special rate under this subsection is available subject to the conditions set out below:
- (e) Required Findings. Before a rate is payable under this section:
  - (1) the Commissioner of the Department of Vermont Health Access, in consultation with the Department's Medical Director, and the Director of Adult Services Division, must make a written finding that the individual's care needs meet the requirements of this section and that the proposed placement is appropriate for that individual's needs; and
  - (2) the Division of Rate Setting, in consultation with the Commissioner of the Department of Vermont Health Access and the Commissioner of the Department of Disabilities, Aging and Independent Living, must determine that the special rate, calculated pursuant to paragraph (e) of this subsection, is reasonable for the services provided.

#### (d) Plan of Care:

(1) Before an individual can be placed with any facility and a rate established, pursuant to this subsection, a plan of care for that person must be approved by the Director of Adult Services Division and the Medical Director of the Department of Vermont Health Access.

(2) The facility shall submit the resident's assessment and plan of care for review by the Director of Adult Services Division and the Medical Director of the Department of Vermont Health Access whenever there is a significant change in the resident's condition, but in no case less frequently than every six months. This review shall form the basis for a determination that the payment of the special rate should be continued or revised pursuant to 14.1(e)(2).

#### (e) Calculation of the Special Rate:

- (1) A per diem rate shall be set by the Division based on the budgeted allowable costs for the individual's plan of care. The rate shall be exempt from the limits in section 7 of these rules.
- (2) From time to time the special rate may be revised to reflect significant changes in the resident's assessment, care plan, and costs of providing care. The Division may adjust the special rate retroactively based on the actual allowable costs of providing care to the resident.
- (3) Special rates set under this section shall not affect the facility's normal per diem rate. The case mix weight of any resident on whose behalf a special rate is paid shall not be included in the calculation of the facility's average case mix score pursuant to subsection 7.2(b), but the days of care shall be included in the facility's Medicaid days and total resident days. The provider shall track the total costs of providing care to the resident and shall self-disallow the incremental cost of such care on cost reports covering the period during which the facility receives Medicaid payments for services to the resident.

# 14.2 Special Rates for Certain Former Patients of the Vermont State Hospital

- (a) A special rate is available for nursing home services to patients transferred directly from the Vermont State Hospital or to such other similarly situated individuals as the Commissioner of Mental Health shall approve. The rate shall be prospective and shall be set before admission of the individual to the facility.
  - (1) The special rate payable for each individual shall consist of the current per diem rate for the receiving facility as calculated pursuant to Sections 5 to 9 of these rules and a monthly supplemental incentive payment. Three levels of supplemental payments are available for the care of residents meeting the eligibility criteria in this subsection based on the severity of the resident's condition and the resources needed to provide care.
  - (2) The supplemental payment will continue to be paid as long as the criteria in paragraph (c) are satisfied.
- (b) To be eligible for a special rate, the receiving facility must have in place a plan of care developed in conjunction with and approved by the Commissioner of Mental Health and the Division of Licensing and Protection.
- (c) Criteria for continuation of supplemental payments:
  - (i) The transferred person continues to reside at the receiving facility.
  - (ii) The facility documents to the satisfaction of the Division of Licensing and Protection that the transferred person continues to present significant behavior management problems by exhibiting behaviors that are significantly more challenging than those of the general nursing facility population.

(d) Any advance payments for days during which the transferred person is not resident or ceases to be eligible for the special transitional rate will be treated as overpayments and subject to refund by deductions from the provider's Medicaid payments.

# 14.3 Special Rates for Medicaid Eligible Furloughees of the Department of Corrections

A special rate equal to 150 percent of a nursing facility's ordinary Medicaid rate shall be paid for care provided to Medicaid eligible furloughees of the Department of Corrections.

### 15 ADMINISTRATIVE REVIEW AND APPEALS

#### **15.1 Draft Findings and Decisions**

- (a) Before issuing findings on any Desk Review, Audit of a Cost Report, or decision on any application for a rate adjustment, the Division shall serve a draft of such findings or decision on the affected provider. If the Division makes no adjustment to a facility's reported costs or application for a rate adjustment, the Division's findings shall be final and shall not be subject to appeal under this section.
- (b) The provider shall review the draft upon receipt. If it desires to review the Division's work papers, it shall file, within 10 days, a written Request for Work Papers on a form prescribed by the Director.

#### 15.2 Request for an Informal Conference on Draft Findings and Decisions

(a) Within 15 days of receipt of either the draft findings or decision or requested work papers, whichever is the later, a provider that is dissatisfied with the draft findings or decision issued pursuant to Subsection 15.1(a) may file a written Request for an Informal Conference with the Division's staff on a form prescribed by the Director.

- (b) Within 10 days of the receipt of the Request, the Division shall contact the provider to arrange a mutually convenient time for the informal conference, which shall be held within 45 days of the receipt of the Request at the Division. The informal conference may be held by telephone. At the conference, if necessary, a date certain shall be fixed by which the provider may file written submissions or other additional necessary information. Within 20 days thereafter, the Division shall issue its official agency action.
- (c) A Request for an Informal Conference must be pursued before a Request for Reconsideration can be filed pursuant to Subsection 15.3. Issues not raised in the Request for Informal Conference shall not be raised at the informal conference or in any subsequent proceeding arising from the same action of the Division, including appeals pursuant to 33 V.S.A. §909.
- (d) Should no timely Request for an Informal Conference be filed within the time period specified in Subsection 15.2(a), the Division's draft findings and/or decision are final and no longer subject to administrative review or judicial appeal.

#### 15.3 Request for Reconsideration

- (a) A provider that is aggrieved by an official action issued pursuant to Subsection 15.2(b) may file a Request for Reconsideration.
- (b) A Request for Reconsideration must be pursued before an appeal can be taken pursuant to 33 V.S.A. 909(a).
- (c) The Request for Reconsideration must be in writing, on a form prescribed by the Director, and filed within 30 days of the provider's receipt of the official action.
- (d) Within 10 days of the filing of a Request for Reconsideration, the provider must file the following:
- (1) A request for a hearing, if desired;

- (2) A clear statement of the alleged errors in the Division's action and of the remedy requested including: a description of the facts on which the Request is based, a memorandum stating the support for the requested relief in this rule, CMS 15, or other authority for the requested relief and the rationale for the requested remedy; and
- (3) If no hearing is requested, evidence necessary to bear the provider's burden of proof, including, if applicable, a proposed revision of the Division's calculations, with supporting work papers.
- (e) Issues not raised in the Request for Reconsideration shall not be raised later in this proceeding or in any subsequent proceeding arising from the same action of the Division, including appeals pursuant to 33 V.S.A. §909.
- (f) If a hearing is requested, within 10 days of the receipt of the Request for Reconsideration, the Division shall contact the provider to arrange a mutually agreeable time.
- (g) The hearing shall be conducted by the Director or her or his designee. The testimony shall be under eath and shall be recorded either stenographically or on tape. If the provider so requests, the Division staff involved in the official action appealed shall appear and testify. The Director, or her or his designee, may hold the record open to a date certain for the receipt of additional materials.
- (h) The Director shall issue a Final Order on Request for Reconsideration no later than 30 days after the record closes. Pending the issuance of a final order, the official action issued pursuant to subsection 15.2(b) shall be used as the basis for setting an interim rate from the first day of the calendar quarter following its issuance. Final orders shall be effective from the effective date of the official action.
- (i) Proceedings under this section are not subject to the requirements of 3 V.S.A. Chapter 25.

### 15.4 Appeals from Final Orders of the Division

- (a) Within 30 days of the date thereof, a nursing facility aggrieved by a Final Order of the Division may file an appeal pursuant to 33 V.S.A. §909(a) and Subsections 15.5, 15.6 and 15.7 of this rule.
- (b) Within 30 days of the date thereof, a ICF/MR aggrieved by a Final Order of the Division may file an appeal using the following procedures. Proceedings under this paragraph are not subject to the requirements of 3 V.S.A. Chapter 25.
  - (1) Request for Administrative Review by the Commissioner of Mental Health. The Commissioner or a designee shall review a final order of the Division of Rate Setting if a timely request is filed with the Director of the Division.
    - (i) Within 10 days of the receipt of the Request, the Director shall forward to the Commissioner a copy of the Request for Administrative Review and the materials that represent the documentary record of the Division's action.
    - (ii) The Commissioner or the designee shall review the record of the appeal and may request such additional materials as they shall deem appropriate, and shall, if requested by the provider, convene a hearing on no less than 10 days written notice to the provider and the Division. Within 45 days after the close of the record, the Commissioner or the designee shall issue a decision which shall be served on the provider and the Division.
  - (2) Appeal to the Secretary of Human Services. Within 20 days of the date of the date of issuance, an ICF/MR aggrieved by

- the Commissioner's decision, may appeal to the Secretary.
  - (i) The Notice of Appeal shall be filed with the Commissioner, who, within 10 days of the receipt of the Notice, shall forward to the Secretary a copy of the Notice and the record of the Administrative Review.
  - (ii) The Secretary or his designee shall review the record of the Administrative Review and may, within their sole discretion, hold a hearing, request more documentary information, or take such other steps to review the Commissioner's decision as shall seem appropriate.
  - (iii) Within 60 days of the filing of the Notice of Appeal or the closing of the record, whichever is the later, the Secretary or the designee shall issue a Final Determination.
- (3) Further review of the Final Determination is available only pursuant to Rule 75 of the Vermont Rules of Civil Procedure.

#### 15.5 Request for Administrative Review to the Secretary of Human Services pursuant to 33 V.S.A. §909(a)(3)

- (a) No appeal may be taken under this section when the remedy requested is retrospective relief from the operation of a provision of this rule or such other relief as may be outside the power of the Secretary to order. Such relief may be pursued by an appeal to the Vermont Supreme Court or Superior Court pursuant to 33 V.S.A. §909(a)(1) and (2), or prospectively by a request for rulemaking pursuant 3 V.S.A. §806.
- (b) Appeals under this section shall be governed by the relevant provisions of the Administrative Procedures Act, 3 V.S.A. §§809-815.
- (c) Proceedings under this section shall be initiated by filing two copies of a written

Request for Administrative Review with the Division, on forms prescribed therefor.

- (d) Within 5 days of receipt of the Request, the Director shall forward one copy to the Secretary. Within 10 days thereafter, the Secretary shall designate an independent appeals officer who shall be a registered or certified public accountant. The Letter of Designation shall be served on all parties to the appeal. All documents filed thereafter shall be filed directly with the independent appeals officer and copies served on all parties.
- (e) Within 10 days of the designation of an independent appeals officer, the Division shall forward to him or her those materials that represent the documentary record of the Division's action.
- (f) Within 30 days thereafter, the independent appeals officer shall, on reasonable notice to the parties, convene a prehearing conference (which may be held by telephone) to consider such matters as may aid in the efficient disposition of the case, including but not limited to:
- (1) the simplification of the issues,
- (2) the possibility of obtaining stipulations of fact and/or admissions of documents which will avoid unnecessary proof,
- (3) the appropriateness of prefiled testimony,
- (4) a schedule for the future conduct of the case.

The independent appeals officer shall make an order which recites the action taken at the conference, including any agreements made by the parties.

(g) The independent appeals officer shall hold a hearing, pursuant to 3 V.S.A. §809, on no less than 10 days written notice to the parties, according to the schedule determined at the prehearing conference. The

- independent appeals officer shall have the power to subpoena witnesses and documents and administer oaths. Testimony shall be under oath and shall be recorded either stenographically or on tape. Prefiled testimony, if admitted into evidence, shall be included in the transcript, if any, as though given orally at the hearing. Evidentiary matters shall be governed by 3 V.S.A. §810.
- (h) The independent appeals officer may allow or require each party to file Proposed Findings of Fact which shall contain a citation to the specific part or parts of the record containing the evidence upon which the proposed finding is based. The Proposed Findings shall be accompanied by a Memorandum of Law which shall address each matter at issue.
- (i) Within 60 days after the date of the hearing, or after the filing of Proposed Findings of Fact, whichever is the later, the independent appeals officer shall file with the Secretary a Recommendation for Decision, a copy of which shall be served on each of the parties. The Recommendation for Decision shall include numbered findings of fact and conclusions of law, separately stated, and a proposed order. If a party has submitted Proposed Findings of Fact, the Recommendation for Decision shall include a ruling upon each proposed finding. Each party's Proposed Findings and Memorandum of Law shall accompany the Recommendation.
- (j) At the time the independent appeals officer makes her or his Recommendation, she or he shall transmit the docket file to the Secretary. The Secretary shall retain the file for a period of at least one year from the date of the Final Determination in the docket. In the event of an appeal of the Secretary's Final Determination to the Vermont Supreme Court or to Superior Court, the Secretary shall make disposition of the file as required by the applicable rules of civil and appellate procedure.

(k) Any party aggrieved by the Recommendation for Decision may file Exceptions, Briefs, and if desired, a written Request for Oral Argument before the Secretary. These submissions shall be filed with the Secretary within 15 days of the date of the receipt of a copy of the Recommendation and copies served on all other parties.

(1) If oral argument is requested, within 20 days of the receipt of the Request for Oral Argument, the Secretary shall arrange with the parties a mutually convenient time for a hearing.

(m) Within 45 days of the receipt of the Recommendation or the hearing on oral argument, whichever is the later, the Secretary shall issue a Final Determination which shall be served on the parties.

(n) A party aggrieved by a Final Determination of the Secretary may obtain judicial review pursuant to 33 V.S.A. §909(a)(1) and (2) and Subsections 15.6 and 15.7 of this Rule.

# 15.6 Appeal to Vermont Supreme Court pursuant to 33 V.S.A. §909(a)(1)

Proceedings under this section shall be initiated, pursuant to the Vermont Rules of Appellate Procedure, as follows:

(a) by filing a Notice of Appeal from a Final Order with the Division; or

(b) by filing a Notice of Appeal from a Final Determination with the Secretary.

# 15.7 Appeal to Superior Court pursuant to 33 V.S.A. §909(a)(2)

De novo review is available in the Superior Court of the county where the nursing facility is located. Such proceedings shall be initiated, pursuant to Rule 74 of the Vermont Rules of Civil Procedure, as follows:

(a) by filing a Notice of Appeal from a Final Order with the Division; or

(b) by filing a Notice of Appeal from a Final Determination with the Secretary.

#### **15.8 Settlement Agreements**

The Director may agree to settle reviews and appeals taken pursuant to Subsections 15.3 and 15.5, and, with the approval of the Secretary, may agree to settle other appeals taken pursuant to 33 V.S.A. §909 and any other litigation involving the Division on such reasonable terms as she or he may deem appropriate to the circumstances of the case.

#### 16 DEFINITIONS AND TERMS

For the purposes of these rules the following definitions and terms are used:

Accrual Basis of Accounting: an accounting system in which revenues are reported in the period in which they are earned, regardless of when they are collected, and expenses are reported in the period in which they are incurred, regardless of when they are paid.

Agency: the Agency of Human Services.

AICPA: American Institute of Certified Public Accountants.

Allowable Costs or Expenses: costs or expenses that are recognized as reasonable and related to resident care in accordance with these rules.

Base Year: a calendar year for which the allowable costs are the basis for the case mix prospective per diem rate.

Case-Mix Weight: a relative evaluation of the nursing resources used in the care of a given class of residents.

Centers for Medicare and Medicaid Services(CMS) (formerly called the Health Care Financing Administration (HCFA)): Agency within the U.S. Department of Health and Human Services (HHS) responsible for developing and implementing policies governing the Medicare and Medicaid programs.

Certificate of Need (CON): certificate of approval for a new institutional health service, issued pursuant to 18 V.S.A. §2403.

Certified Rate: the rate certified by the Division of Rate Setting to the Department of Vermont Health Access.

Common Control: where an individual or organization has the power to influence or direct the actions or policies of both a provider and an organization or institution serving the provider, or to influence or direct the transactions between a provider and an organization serving the provider. The term includes direct or indirect control, whether or not it is legally enforceable.

Common Ownership: where an individual or organization owns or has equity in both a facility and an institution or organization providing services to the facility.

Companion Aide: a Licensed Nurse Aide (LNA) with specialized training in person centered dementia care.

Cost Finding: the process of segrogating direct costs by cost centers and allocating indirect costs to determine the cost of services provided.

Cost Report: a report prepared by a provider on forms prescribed by the Division.

**Direct Costs:** costs which are directly identifiable with a specific activity, service or product of the program.

Director: the Director of Rate Setting.

**Division:** the Division of Rate Setting, Agency of Human Services.

**Donated Asset:** an asset acquired without making any payment in the form of cash, property or services.

Facility or nursing facility: a nursing home facility licensed and certified for participation in the Medicaid Program by the State of Vermont.

Fair Market Value: the price an asset would bring by bona fide bargaining between well-informed buyers and sellers at the date of acquisition.

FASB: Financial Accounting Standards Board.

Final Order of the Division: an action of the Division which is not subject to change by the Division, for which no review or appeal is available from the Division, or for which the review or appeal period has passed.

Free standing facility: a facility that is not hospital affiliated.

Funded Depreciation: funds that are restricted by a facility's governing body for purposes of acquiring assets to be used in rendering resident care or servicing long term debt.

Fringe Benefits: shall include payroll taxes, workers' compensation, pension, group health, dental and life insurances, profit sharing, cafeteria plans and flexible spending plans, child care for employees, employee parties, and gifts shared by all staff. Fringe benefits may include tuition for college credit in a discipline related to the individual staff member's employment or costs of obtaining a GED.

Generally Accepted Accounting Principles (GAAP): those accounting principles with substantial authoritative support. In order of authority the following documents are considered GAAP: (1) FASB Standards and Interpretations, (2) APB Opinions and Interpretations, (3) CAP Accounting

Research Bulletins, (4) AICPA Statements of Position, (5) AICPA Industry Accounting and Auditing Guides, (6) FASB Technical Bulletins, (7) FASB Concepts Statements, (8) AICPA Issues Papers and Practice Bulletins, and other pronouncements of the AICPA or FASB.

Generally Accepted Auditing Standards (GAAS): the auditing standards that are most widely recognized in the public accounting profession.

Health Care Cost Service: publication, by Global Insight, Inc., of national forecasts of hospital, nursing home (NHMB), and home health agency market baskets and regional forecasts of CPI (All Urban) for food and commercial power and CPIU All Items.

Hold Day: a day for which the provider is paid to hold a bed open is counted as a resident day.

Hospital affiliated facility: a facility that is a distinct part of a hospital provider, located either at the hospital site or within a reasonable proximity to the hospital.

Incremental Cost: the added cost incurred in alternative choices.

Independent Public Accountant: a Certified Public Accountant or Registered Public Accountant not employed by the provider.

Indirect Costs: costs which cannot be directly identified with a particular activity, service or product of the program. Indirect costs are apportioned among the program's services using a rational statistical basis.

Inflation Factor: a factor that takes into account the actual or projected rate of inflation or deflation as expressed in indicators such as the New England Consumer Price Index.

Interim Rate: a prospective Case Mix rate paid to nursing facilities on a temporary basis.

Look-back: a review of a facility's actual costs for a previous period prescribed by the Division.

Medicaid Resident: a nursing home resident for whom the primary payor for room and board is the Medicaid program.

New England Consumer Price Index (NECPI-U): the New England consumer price index for all urban consumers as published by the Health Care Cost Service.

New Health Care Project: A project requiring a certificate of need (CON) pursuant to 18 V.S.A.§9434(a) or projects which would require a CON except that their costs are lower than those required for CON jurisdiction pursuant 18 V.S.A.§ 9434(a).

OBRA 1987: the Omnibus Budget Reconciliation Act of 1987.

Occupancy Level: the number of paid days, including hold days, as a percentage of the licensed bed capacity.

Paid feeding/dining assistants: persons (other than the facility's administrator, registered nurses, licensed practical nurses, certified or licensed nurse aides) who are qualified under state law pursuant to 42 C.F.R. §§483.35(h)(2), 483.160 and 488.301 and who are paid to assist in the feeding of residents.

Per Diem Cost: the cost for one day of resident care.

Prescription Drugs: drugs for which a physician's prescription is required by state or federal law.

Person-Centered Dementia Care: care that includes the following elements: an individualized approach to care planning that

uses the perspective of the person with dementia as the primary frame of reference; values the personhood of the individual with dementia; and provides a social environment that supports psychological needs.

Prospective Case Mix Reimbursement System: a method of paying health care providers rates that are established in advance. These rates take into account the fact that some residents are more costly to care for than others.

Provider Reimbursement Manual, CMS-15: a manual published by the U.S. Department of Health and Human Services, Centers for Medicare and Medicaid Services, used by the Medicare Program to determine allowable costs.

Rate year: the State's fiscal year ending June 30.

Related organization or related party: an individual or entity that is directly or indirectly under common ownership or control or is related by family or other business association with the provider. Related organizations include but are not restricted to entities in which an individual who directly or indirectly receives or expects to receive compensation in any form is also an owner, partner, officer, director, key employee, or lender, with respect to the provider, or is related by family to such persons.

Resident Assessment Form: Vermont version of a federal form, which captures data on a resident's condition and which is used to predict the resource use level needed to care for the resident.

Resident Day: any day of services for which the facility is paid. For example, a paid hold day is counted as a resident day.

Restricted Funds and Revenue: funds and investment income earned from funds restricted for specific purposes by donors,

excluding funds restricted or designated by an organization's governing body.

RUG IV: A systematic classification of residents in nursing facilities based upon a broad study of nursing care time required by groups of residents exhibiting similar needs.

Secretary: the Secretary of the Agency of Human Services.

Special hospital-based nursing facility: a facility that meets the following criteria: (a) is physically integrated as part of a hospital building with at least one common wall and a direct internal access between the hospital and the nursing home; (b) is part of a single corporation that governs both the hospital and the nursing facility; and (c) files one Medicare cost report for both the hospital and the nursing home.

Standardized Resident Days: Base Year resident days multiplied by the facility's average Case Mix score for the base year.

State nursing facilities: facilities owned and/or operated by the State of Vermont.

Swing-Bed: a hospital bed used to provide nursing facility services.

#### 17 TRANSITIONAL PROVISIONS

Notwithstanding any other provisions of these rules, the amendments to these rules effective March 6, 2015 shall be applied to payments for services rendered on or after March 1, 2015.

#### 17.1 Companion Aide Pilot Project

The Companion Aide Pilot Project will provide a per diem rate adjustment to selected facilities to develop additional knowledge and experience in the area of person-centered dementia care through the use of Companion Aides. Companion Aides will be Licensed Nurse Aides with

specialized training in person-centered dementia care to provide an individualized approach that uses the perspective of the person with dementia as the primary frame of reference.

The work of the Companion Aides funded by this pilot program must comply with the job description detailed in the Companion Aide application. The selected nursing facilities may have the Companion Aide work any shift.

The pilot project will be for 2.5 years beginning January 1, 2015 and ending on June 30, 2017.

#### (a) Selection Process

- (1) All Vermont nursing facilities participating in the Medicaid program are eligible to apply.
- (2) Five facilities will be selected from the pool of completed applications by the Commissioner of the Department of Disabilities Aging and Independent Living. One facility will be selected from each of five geographical areas of the State based on the county groupings in the Council on Aging service areas. These geographical areas will be Northwest Vermont (Addison, Chittenden, Franklin and Grand Isle counties); Northeast Vermont (Caledonia, Essex, and Orleans counties); Central Vermont (Lamoille, Orange, and Washington counties); Southwest Vermont (Rutland and Bennington counties); and Southeast Vermont (Windham and Windsor counties).
- (3) Within each geographical area, the applicants will be ranked by the proportion of their residents with a diagnosis of Alzheimer's or dementia compared to the number of total residents, and the facility with the highest proportion will be selected. This data will be reported on the Companion Aide application and must be from the Minimum Data Set (MDS)

information used for the June 15, 2014 picture date in the second quarter of 2014.

- (4) If no nursing facility applies from a given region, an additional nursing facility from the geographical area with the highest number of applicants will be selected. If there are two regions with no applicants, an additional facility then will be selected from the geographical area with the second highest number of applicants.
- (5) If there is a tie in the selection process, the facility with the highest percentage of Medicaid residents to total residents for State fiscal year 2014, based on census information reported to the Division of Rate Setting, will be selected.

# (b) Rate Adjustment Calculations and Procedures

- (1) The rate adjustment will include the salary and fringe benefit costs for the approved number of Companion Aides at the selected facilities. The hourly salaries and fringe benefit rates will be reported on the Companion Aide application and reviewed by the Division of Rate Setting.
- (2) The selected facilities will be funded at a ratio of five Companion Aides per 100 filled beds. The calculated number shall be rounded up or down to determine the number of Companion Aide Full Time Equivalents (2,080 hours/year). The resulting number of aides to be funded will vary with the number of filled beds at the selected facilities.
- (3) The number of total beds filled shall equal the total number of residents reported on the June 15, 2014 MDS picture date (Q2 2014) summary report supplied to the Division of Licensing and Protection.

#### (c) Inflation of Rate Adjustments

The original per diem adjustment for Companion Aides will be inflated on July 1, 2015 and July 1, 2016 using the same methodology as detailed in Subsection 5.8 of these rules.

# (d) End of Adjustment and Special Nursing Rebase Provisions

- (1) The adjustments in this Section will be terminated as of July 1, 2017 when Nursing Care costs are rebased to base year 2015. This will be the first year when the costs of the Companion Aides will be in the facility's base year costs.
- (2) For facilities with years ending earlier than December 31, the Division will annualize the cost of the Companion Aides so that a full year of these costs will be included in the selected facilities' 2015 base year costs.
- (3) The Companion Aide costs at the five selected facilities will be exempt from the cap on nursing costs in the July 1, 2017 rebase. In rebases after that time, the extant cap on Nursing Care Costs will apply.

#### (e) Ongoing Reporting Requirements

The selected facilities shall complete an annual Companion Aide Pilot Project Outcome Report. This report will be sent to the providers with the Companion Aide application so nursing facility staff will understand the data reporting requirement when they apply for the pilot. These reports will be due by November 10, 2015 and November 10, 2016. The Division may end the Companion Aide rate adjustment for a facility that does not comply with the ongoing reporting requirements.

# Methods, Standards, and Principles for Establishing Medicaid Payment Rates for Long-Term Care Facilities

## 5.101 Definitions

For the purposes of this rule, the term:

Accrual Basis of Accounting means an accounting system in which revenues are reported in the period in which they are earned, regardless of when they are collected, and expenses are reported in the period in which they are incurred, regardless of when they are paid.

**AICPA** means the American Institute of Certified Public Accountants.

Allowable Costs or Expenses means costs or expenses that are recognized as reasonable and related to resident care in accordance with these rules.

Base Year means a calendar year for which the allowable costs are the basis for the case-mix prospective per diem rate.

Case-Mix Weight means a relative evaluation of the nursing resources used in the care of a given class of residents.

Certificate of Need (CON) means certificate of approval for a new institutional health service, issued pursuant to 18 V.S.A. § 2403.

Certified Rate means the rate certified by the Division of Rate Setting to the Department of Vermont Health Access.

Common Control means where an individual or organization has the power to influence or direct the actions or policies of both a provider and an organization or institution serving the provider, or to influence or direct the transactions between a provider and an organization serving the provider. The term includes direct or indirect control, whether or not it is legally enforceable.

**Common Ownership** means where an individual or organization owns or has equity in both a facility and an institution or organization providing services to the facility.

Companion Aide means a Licensed Nurse Aide (LNA) with specialized training in person-centered dementia care.

Cost Finding means the process of segregating direct costs by cost centers and allocating indirect costs to determine the cost of services provided.

Cost Report means a report prepared by a provider on forms prescribed by the Division.

**Direct Costs** means costs which are directly identifiable with a specific activity, service or product of the program.

**Director** means the Director of Rate Setting.

**Division** means the Division of Rate Setting, Department of Vermont Health Access, Agency of Human Services.

Donated Asset means an asset acquired without making any payment in the form of cash, property or services.

**Facility or nursing facility** means a nursing home facility licensed and certified for participation in the Medicaid Program by the State of Vermont.

Fair Market Value means the price an asset would bring by bona fide bargaining between well-informed buyers and sellers at the date of acquisition.

**FASB** means Financial Accounting Standards Board.

**Final Order of the Division** means an action of the Division which is not subject to change by the Division, for which no review or appeal is available from the Division, or for which the review or appeal period has passed.

Free standing facility means a facility that is not hospital-affiliated.

Funded Depreciation means funds that are restricted by a facility's governing body for purposes of acquiring assets to be used in rendering resident care or servicing long term debt.

Fringe Benefits include benefits such as payroll taxes, workers' compensation, pension, group health, dental and life insurances, profit sharing, cafeteria plans and flexible spending plans, child care for employees, employee parties, and gifts shared by all staff. Fringe benefits may include tuition for college credit in a discipline related to the individual staff member's employment or costs of obtaining a GED.

Generally Accepted Accounting Principles (GAAP) means those accounting principles with substantial authoritative support. In order of authority the following documents are considered GAAP: (1) FASB Standards and Interpretations, (2) APB Opinions and Interpretations, (3) CAP Accounting Research Bulletins, (4) AICPA Statements of Position, (5) AICPA Industry Accounting and Auditing Guides, (6) FASB Technical Bulletins, (7) FASB Concepts Statements, (8) AICPA Issues Papers and Practice Bulletins, and other pronouncements of the AICPA or FASB.

Generally Accepted Auditing Standards (GAAS) means the auditing standards that are most widely recognized in the public accounting profession.

Health Care Cost Service means the publication, by Global Insight, Inc., of national forecasts of hospital, nursing home (NHMB), and home health agency market baskets and regional forecasts of CPI (All Urban) for food and commercial power and CPIU-All Items.

Hold Day means a day for which the provider is paid to hold a bed open is counted as a resident day.

Hospital-affiliated facility means a facility that is a distinct part of a hospital provider, located either at the hospital site or within a reasonable proximity to the hospital.

Incremental Cost means the added cost incurred in alternative choices.

**Independent Public Accountant** means a Certified Public Accountant or Registered Public Accountant not employed by the provider.

**Indirect Costs** means costs which cannot be directly identified with a particular activity, service or product of the program. Indirect costs are apportioned among the program's services using a rational statistical basis.

**Inflation Factor** means a factor that takes into account the actual or projected rate of inflation or deflation as expressed in indicators such as the New England Consumer Price Index.

Interim Rate means a prospective Case-Mix rate paid to nursing facilities on a temporary basis.

Look-back means a review of a facility's actual costs for a previous period prescribed by the Division.

Medicaid Resident means a nursing home resident for whom the primary payor for room and board is the Medicaid program.

New England Consumer Price Index (NECPI-U) means the New England consumer price index for all urban consumers as published by the Health Care Cost Service.

New Health Care Project means a project requiring a certificate of need (CON) pursuant to 18 V.S.A.§ 9434(a) or projects which would require a CON except that their costs are lower than those required for CON jurisdiction pursuant 18 V.S.A.§ 9434(a).

OBRA 1987 means the Omnibus Budget Reconciliation Act of 1987.

Occupancy Level means the number of paid days, including hold days, as a percentage of the licensed bed capacity.

Paid feeding/dining assistants means persons (other than the facility's administrator, registered nurses, licensed practical nurses, certified or licensed nurse aides) who are qualified under state law pursuant to 42 C.F.R. §§483.35(h)(2), 483.160 and 488.301 and who are paid to assist in the feeding of residents.

5.101

Per Diem Cost means the cost for one day of resident care.

**Prescription Drugs** means drugs for which a physician's prescription is required by state or federal law.

**Person-Centered Dementia Care** means care that includes the following elements: an individualized approach to care planning that uses the perspective of the person with dementia as the primary frame of reference; values the personhood of the individual with dementia; and provides a social environment that supports psychological needs.

**Prospective Case-Mix Reimbursement System** means a method of paying health care providers rates that are established in advance. These rates take into account the fact that some residents are more costly to care for than others.

**Provider Reimbursement Manual, CMS-15** means a manual published by the U.S. Department of Health and Human Services, Centers for Medicare and Medicaid Services, used by the Medicare Program to determine allowable costs.

Rate year means the State's fiscal year ending June 30.

Related organization or related party means an individual or entity that is directly or indirectly under common ownership or control or is related by family or other business association with the provider. Related organizations include but are not restricted to entities in which an individual who directly or indirectly receives or expects to receive compensation in any form is also an owner, partner, officer, director, key employee, or lender, with respect to the provider, or is related by family to such persons.

Resident Assessment Form means Vermont version of a federal form, which captures data on a resident's condition and which is used to predict the resource use level needed to care for the resident.

Resident Day means any day of services for which the facility is paid. For example, a paid hold day is counted as a resident day.

Restricted Funds and Revenue means funds and investment income earned from funds restricted for specific purposes by donors, excluding funds restricted or designated by an organization's governing body.

Secretary means the Secretary of the Agency of Human Services.

Special hospital-based nursing facility means a facility that meets the following criteria: (a) is physically integrated as part of a hospital building with at least one common wall and a direct internal access between the hospital and the nursing home; (b) is part of a single corporation that governs both the hospital and the nursing facility; and (c) files one Medicare cost report for both the hospital and the nursing home.

**Standardized Resident Days** means Base Year resident days multiplied by the facility's average Case-Mix score for the base year.

State nursing facilities means facilities owned and/or operated by the State of Vermont.

Swing-Bed means a hospital bed used to provide nursing facility services.

## 5.101.1 General Provisions

#### 5.101.1.1 Purpose

The purpose of these rules is to implement state and federal reimbursement policy with respect to nursing facilities providing services to Medicaid eligible persons. The methods, standards, and principles of rate setting established herein reflect the objectives set out in 33 V.S.A. §901 and balance the competing policy objectives of access, quality, cost containment and administrative feasibility. Rates set under this payment system are consistent with the efficiency, economy, and quality of care necessary to provide services in conformity with state and federal laws, regulations, quality and safety standards, and meet the requirements of 42 U.S.C. § 1396a(a)(13)(A).

## 5.101.1.2 <u>Scope</u>

These rules apply to all privately owned nursing facilities and state nursing facilities providing services to Medicaid residents. Long-term care services in swing-bed hospitals are reimbursed under different methods and standards. Swing-bed hospitals are reimbursed pursuant to 42 U.S.C. § 1396l(b)(1).

## 5.101.1.3 <u>Authority</u>

These rules are promulgated pursuant to 33 V.S.A. §§ 904(a) and 908(c) to meet the requirements of 33 V.S.A. Chapter 9, 42 U.S.C. §§ 1396a(a)(13)(A) and §1396a(a)(30).

# 5.101.1.4 General Description of the Rate Setting System

Vermont Medicaid shall employ a prospective case-mix payment system for nursing facilities in which the payment rate for services is set in advance of the actual provision of those services. A per diem rate is set for each facility based on the historic allowable costs of that facility. The costs are divided into certain designated cost categories, some of which are subject to limits. The basis for reimbursement within the Nursing Care cost category is a resident classification system that groups residents into classes according to their assessed conditions and the resources required to care for them. The costs in some categories are adjusted to reflect economic trends and conditions, and the payment rate for each facility is based on the per diem costs for each category.

# 5.101.1.5 Requirements for Participation in Medicaid Program

- (a) Nursing facilities must satisfy all of the following prerequisites in order to participate in the Medicaid program:
  - (1) be licensed by the Agency, pursuant to 33 V.S.A. §7103(b),
  - (2) be certified by the Secretary of Health and Human Services pursuant to 42 C.F.R. Part 442, Subpart C, and
  - (3) have executed a Provider Agreement with the Agency, as required by 42 C.F.R. Part 442, Subpart B.
- (b) To the extent economically and operationally feasible, providers are encouraged, but not required, to be certified for participation in the Medicare program, pursuant to 42 C.F.R. §488.3.
- (c) Medicaid payments shall not be made to any facility that fails to meet all the requirements of Section 5.101.1.5(a).

## 5.101.1.6 Responsibilities of Owners

Owners must prudently manage and operate a residential health care program of adequate quality to meet its residents' needs. Regardless of the per diem rate set by the Division, or any other orders made by the Director, Commissioner, or Secretary under these rules the owner of a nursing facility must comply with the requirements and standards of the Agency of Human Services.

## 5.101.1.6 Duties of the Owner

The owner of a nursing facility, or a duly authorized representative shall:

- (a) Comply with the provisions of these rules, the Nursing Facility Provider Manual, and all applicable state and federal laws and rules.
- (b) Submit cost reports in accordance with the provisions of sections 5.101.3.2 and 5.101.3.3 of these rules and the Nursing Facility Provider Manual.
- (c) Maintain adequate financial and statistical records and make them available at reasonable times for inspection by an authorized representative of the Division, the state, or the federal government.
- (d) Assure that an annual audit is performed in conformance with Generally Accepted Auditing Standards (GAAS).
- (e) Assure that the construction of buildings and the maintenance and operation of premises and programs comply with all applicable health and safety standards.

5.101

(f) Notwithstanding any other provision of these rules, any provider that fails to make a complete cost report filing within the time prescribed in section 5.101.3.3(a) or fails to file any other materials requested by the Division within the time prescribed shall receive no increase to its Medicaid rate until the first day of the calendar quarter after a complete cost report or the requested materials are filed, unless within an extension of time previously approved by the Division.

# 5.101.1.7 Powers and Duties of the Division and the Director

- (a) The Division shall establish and certify to the Department of Vermont Health Access per diem rates for payment to providers of nursing facility services on behalf of residents eligible for assistance under Title XIX of the Social Security Act.
- (b) The Division may request any nursing facility, related party, or similar individual or organization to file data, statistics, schedules, or information as the Division finds necessary to enable it to carry out its function.
- (c) The Division may examine the books and accounts of any nursing facility, related parties, or similar individuals or organizations, subpoena witnesses and documents, administer oaths to witnesses and examine them on all matters over which the Division has jurisdiction.
- (d) From time to time, the Director may issue notices of practices and procedures employed by the Division in carrying out its functions under these rules.
- (e) The Director shall prescribe the forms required by these rules and instructions for their completion.
- (f) The Director shall issue, amend, and enforce the Nursing Facility Manual.
- (g) These rules and the Nursing Facility Manual apply regardless whether the Division's final per diem rates or final orders fail to enforce their provisions. If the Division's final per diem rates or final orders fail to enforce a provision of these rules or the Manual, that does not waive these rules or the Manual. The Division shall continue to have the right and the obligation to enforce these rules and the Manual.

# 5.101.1.8 <u>Powers and Duties of the Department of Disabilities, Aging and Independent Living's Division of Licensing and Protection as Regards Reimbursement</u>

(a) The Division of Licensing and Protection of the Department of Disabilities, Aging and Independent Living shall receive from providers resident assessments on forms it specifies. The Department of Disabilities, Aging and Independent Living shall process this information and shall periodically, but no less frequently than quarterly, provide the Division of Rate Setting with the average case-mix scores of each facility

- based on Vermont Medicaid's chosen resident classification system. This score will be used in the quarterly determination of the Nursing Care portion of the rate.
- (b) The management of the resident assessment process used in the determination of casemix scores shall be the duty of the Division of Licensing and Protection of the Department of Disabilities, Aging and Independent Living. Any disagreements between the facility's assessment of a resident and the assessment of that same resident by the audit staff of Licensing and Protection shall be resolved with the Division of Licensing and Protection and shall not involve the Division of Rate Setting. As the final rates are prospective and adjusted on a quarterly basis to reflect the most current data, the Division of Rate Setting will not make retroactive rate adjustments as a result of audits or successfully appealed individual case-mix scores.

## 5.101.1.9 Computation of and Enlargement of Time, Filing and Service of Documents

- (a) When computing time under these rules or the Nursing Facility Manual, the day of the act or event that begins a period of time shall not be included in that period. The last day of the period of time shall be included, unless it is a Saturday, Sunday, or state or federal legal holiday, in which case the period runs until the next business day..
- (b) The addressee of any notice or document issued by the Division is rebuttably presumed to have received the notice or document three days after the date on the document.
- (c) The Division may extend a period of time set in these rules with or without motion or notice for good cause. This section shall not apply to the time limits for appeals to the Vermont Supreme Court or Superior Court from Final Orders of the Division or Final Determinations of the Secretary, which are governed by the Vermont Rules of Appellate Procedure and the Vermont Rules of Civil Procedure respectively.
- (d) Filing shall be deemed to have occurred when a document is received and date-stamped as received at the office of the Division or in the case of a document directed to be filed under this rule other than at the office of the Division, when it is received and stamped as received at the appropriate office. Filings with the Division may be made by telefacsimile (FAX), but the sender bears the risk of a communications failure from any cause. Filings with the Division may also be made electronically, but the sender bears the risk of a communications failure from any cause, including, but not limited to, filings blocked due to size. If a provider files a document by FAX or electronically, the provider need not file a hard copy of the document.
- (e) The Division shall serve any document required to be served by this rule or the Nursing Facility Provider Manual in accordance with the Nursing Facility Provider Manual.

# 5.101.1.10 Representation in All Matters before the Division

A facility may be represented in any matter under this rule as described in the Nursing Facility Provider Manual.

## 5.101.1.11 Severability

If any part of these rules or their application is held invalid, the invalidity does not affect other provisions or applications which can be given effect without the invalid provision or application, and to this end the provisions of these rules are severable.

## 5.101.1.12 Effective Date

- (a) These rules are effective from January 29, 1992 (as most recently amended July 1, 2024).
- (b) Application of Rule: Amended provisions of this rule shall apply to:
  - (1) all cost reports draft findings issued on or after the effective date of the most recent amendment, and
  - (2) all rates set on or after the effective date of the most recent amendment.
- (c) If these rules or the Nursing Facility Provider Manual are amended while an administrative proceeding is pending, the Director or Secretary may apply the prior version of the rule or manual if applying the current version would work an injustice or substantial inconvenience.

#### 5.101.2 Accounting Requirements

## 5.101.2.1 Accounting Principles

- (a) All financial and statistical reports shall be prepared in accordance with Generally Accepted Accounting Principles (GAAP), consistently applied, unless these rules or the Nursing Facility Provider Manual authorize specific variations in such principles.
- (b) The provider shall establish and maintain a financial management system which provides for adequate internal control assuring the accuracy of financial data, safeguarding of assets and operational efficiency.
- (c) Providers shall report on an accrual basis. Providers whose records are not maintained on an accrual basis shall develop accrual data for reports on the basis of an analysis of the available documentation. In such a case, the provider's accounting process shall provide sufficient information to compile data to satisfy the accrued expenditure reporting requirements and to demonstrate the link between the accrual data reports

5.101

and the non-accrual fiscal accounts. The provider shall retain all such documentation for audit purposes.

## 5.101.2.2 Procurement Standards

Providers shall establish a code of standards to govern the performance of employees that procure goods and services in accordance with the Nursing Facility Provider Manual.

# 5.101.2.3 Cost Allocation Plans and Changes in Accounting Principles

Providers may reasonably allocate costs to the nursing facility from related entities, and may reasonably allocate costs from the nursing facility to related entities. The Division shall review cost allocations in accordance with the Nursing Facility Provider Manual. The Division reserves the right not to recognize changes in accounting principles or methods or bases of cost allocation that are unreasonable or are made for the purpose of, or having the likely effect of, increasing a facility's Medicaid payments.

## 5.101.2.4 Substance over Form

The substance of a transaction shall prevail over the form. Accordingly, the Division may adjust the cost effect of a transaction that circumvents the intention of these rules or the Nursing Facility Provider Manual.

# 5.101.2.5 Record Keeping and Retention of Records

- (a) Each provider must maintain complete documentation of all records that substantiate the date the provider reports to the Division.
- (b) Each provider must make all records described in subsection (a) of this section available to the Division of Rate Setting, the federal Department of Health and Human Services, and any authorized representatives of those agencies.
- (c) Complete documentation means clear and compelling evidence of all of the financial transactions of the provider and affiliated entities, including but not limited to census data, ledgers, books, invoices, bank statements, canceled checks, payroll records, copies of governmental filings, time records, time cards, purchase requisitions, purchase orders, inventory records, basis of apportioning costs, matters of provider ownership and organization, resident service schedule and amounts of income received by service, or any other record which is necessary to provide the Director with the highest degree of confidence in the reliability of the claim for reimbursement. For purposes of this definition, affiliated entities shall extend to realty, management and other entities for which any reimbursement is directly or indirectly claimed whether or not they fall within the definition of related parties.
- (d) The provider shall maintain all such records for at least six years from the date of filing, or the date upon which the fiscal and statistical records were to be filed,

5.101

whichever is the later. The Division shall keep all cost reports, supporting documentation submitted by the provider, correspondence, workpapers and other analyses supporting Summaries of Findings for six years. In the event of litigation or appeal involving rates established under these regulations, the provider and Division shall retain all records which are in any way related to such legal proceeding until the proceeding has terminated and any applicable appeal period has lapsed.

(e) Pursuant to 33 V.S.A. § 908(a), all documents and other materials filed with the Division are public information, except for individually identifiable health information protected by law or the policies, practices, and procedures of the Agency of Human Services.

## 5.101.3 Financial Reporting

## 5.101.3.1 Repealed

# 5.101.3.2 Uniform Cost Reports

Each long-term care facility participating in the Vermont Medicaid program shall annually, or upon request, submit a uniform financial and statistical report (cost report) on forms prescribed by the Division and in accordance with the Nursing Facility Provider Manual.

# 5.101.3.3 Adequacy and Timeliness of Filing

- (a) Providers shall file acceptable cost reports on or before the last day of the fifth month following the close of the period covered by the report, subject to the following exceptions:
  - (1) Hospital-based nursing homes shall file their Medicaid cost-reports within five days after filing their Medicare cost report for the same cost reporting period with CMS.
  - (2) If a hospital-based Medicaid nursing home's cost report is not filed on or before June 30 following the end of the facility's fiscal year, the Division may require the facility to provide certain data or to file a draft cost report.
  - (3) The Division may grant an extension to any facility's filing deadline, as described in the Nursing Facility Provider Manual.
- (b) The Division may reject any filing which does not comply with these rules, the cost reporting instructions, or the Nursing Facility Provider Manual. If the Division rejects a cost report filing, the report shall be deemed not filed until the provider files an acceptable cost report that complies with these rules, the cost reporting instructions, and the Nursing Facility provider Manual.
- (c) Repealed.

(d) Notwithstanding any other provision of these rules, any provider that fails to make a complete cost report filing within the time prescribed in section 5.101.3.3(a) or within an extension of time approved by the Divisions shall be subject to the provisions of section 5.101.1.7(f).

# 5.101.3.4 Review of Cost Reports by Division

## (a) Uniform Desk Review

- (1) The Division shall perform a uniform desk review on each cost report submitted.
- (2) The uniform desk review is an analysis of the provider's cost report to determine the adequacy and completeness of the report, accuracy and reasonableness of the data recorded thereon, allowable costs and a summary of the results of the review for the purpose of either settling the cost report without an on-site audit or determining the extent to which an on-site audit verification is required.
- (3) Uniform desk reviews shall be completed within an average of 18 months after receipt of an acceptable cost report filing, except in unusual situations, including but not limited to, delays in obtaining necessary information from a provider. Notwithstanding this subdivision, the Division shall have an additional six months to complete its review or audits of facilities' base year cost reports.
- (4) Unless the Division schedules an on-site audit, it shall issue a written summary report of its findings and adjustments upon completion of the uniform desk review.

#### (b) On-site Audit

- (1) The Division will perform on-site audits, as considered appropriate, of the provider's financial and statistical records and systems in accordance with the relevant provisions of the *Medicare Intermediary Manual Audits-Reimbursement Program Administration*, CMS Publication 13-2 (CMS-13).
- (2) The Division will base its selection of a facility for an on-site audit on factors such as length of time since last audit, changes in facility ownership, management, or organizational structure, evidence or official complaints of financial irregularities, questions raised in the uniform desk review, failure to file a timely cost report without a satisfactory explanation, and prior experience.

- (3) The audit scope will be limited so as to avoid duplication of work performed by an independent public accountant, provided such work is adequate to meet the Division's audit requirements.
- (4) Upon completion of an audit, the Division shall review its draft findings and adjustments with the provider and issue a written summary report of such findings.

## 5.101.3.5 Settlement of Cost Reports

- (a) A cost report is settled if there is no request for reconsideration of the Division's findings or, if such request was made, the Division has issued a final order pursuant to section 5.101.15.3 of these rules.
- (b) The Division may correct or reopen a determination or order regarding a cost report, even when it is final, in accordance with the process laid out in the Nursing Facility Provider Manual.
- (c) Repealed.
- (d) Repealed.
- (e) Repealed.
- (f) Repealed.
- (g) Repealed.

#### 5.101.4 Determination of Allowable Costs for Nursing Facilities

## 5.101.4.1 Provider Reimbursement Manual and GAAP

In determining the allowability or reasonableness of costs or treatment of any reimbursement issue not addressed in these rules or the Nursing Facility Provider Manual, the Division shall apply the appropriate provisions of the Medicare Provider Reimbursement Manual (CMS-15, formerly known as HCFA or HIM-15). If neither these rules nor the Nursing Facility Provider Manual nor CMS-15 specifically addresses a particular issue, the determination of allowability will be made in accordance with Generally Accepted Accounting Principles (GAAP). The Division reserves the right, consistent with applicable law, to determine the allowability and reasonableness of costs in any case not specifically covered in the sources referenced in this section.

## 5.101.4.2 General Cost Principles

For rate setting purposes, a cost must satisfy criteria, including, but not limited to, the following:

- 5.101
- (a) The cost must be ordinary, reasonable, necessary, related to the care of residents, and actually incurred.
- (b) The cost adheres to the prudent buyer principle.
- (c) The cost is related to goods and/or services actually provided in the nursing facility.

#### 5.101.4.3 Non-Recurring Costs

Non-recurring costs shall be capitalized and amortized and carried as an on-going adjustment beginning with the first quarterly rate change after the settlement of the cost report for a period of three years as described in the Nursing Facility Provider Manual.

## 5.101.4.4 Interest Expense

- (a) Necessary and proper interest is an allowable cost.
- (b) The Nursing Facility Provider Manual shall define when interest expenses are necessary and proper, how providers must report interest expenses, and other reporting rules related to interest expenses.

## 5.101.4.5 Basis of Property, Plant and Equipment

The Division shall assess the basis of donated, owned, constructed, improved, or transferred assets in accordance with the Nursing Facility Provider Manual.

#### 5.101.4.6 Depreciation and Amortization of Property, Plant and Equipment

- (a) Costs for depreciation and amortization must be based on property records sufficient in detail to identify specific assets.
- (b) Providers must compute depreciation and amortization in accordance with the Nursing Facility Provider Manual.
- (c) Repealed.
- (d) The Division shall estimate the useful life of an asset in accordance with the Nursing Facility Provider Manual.

## 5.101.4.7 Change in Ownership of Depreciable Assets – Sales of Facilities

A facility may qualify for an adjustment in the basis of a depreciable asset after it has changed ownership. The Division's process for recognizing a change of ownership and the according adjustment to a depreciable asset's basis shall be provided in the Nursing Facility Provider Manual.

#### 5.101.4.8 Repealed

# 5.101.4.9 Leasing Arrangements for Property, Plant and Equipment

The Division will recognize costs associated with leasing arrangements for property, plant, and equipment in accordance with the Nursing Facility Provider Manual.

## 5.101.4.10 Funding of Depreciation

The Division strongly recommends that providers use depreciation to conserve funds to replace depreciable assets and that providers coordinate capital expenditure planning with community and state agencies. The Division shall recognize depreciation in accordance with the Nursing Facility Provider Manual.

# 5.101.4.11 Adjustments for Large Asset Acquisitions and Changes of Ownership

# (a) Large Asset Acquisitions

- (1) A provider may apply to the Division for an adjustment to the property and related component of the rate for *individual* capital expenditures determined to be necessary and reasonable. No application for a rate adjustment should be made if the change to the rate would be smaller than one half of one percent of the facility's rate in effect at the time the application is made. Interest expense related to these assets, provided it is necessary and reasonable, shall be included in calculating the adjustment.
- (2) In the event that approval is granted by the Division, the adjustment will be made effective from the first day of the quarter after the filing date of the written notice, following the date of the final order on the application, or following the date the asset is actually put into service, whichever is the latest.

## (b) Changes of Ownership

- (1) Application shall also be made under this section, no later than 30 days after the execution of a purchase and sale agreement or other binding contract, or the receipt of a Certificate of Need pursuant to 18 V.S.A. §9434, for changes in basis resulting from a change in ownership of depreciable assets recognized by the Division pursuant to section 5.101.4.7. The Division may make related adjustments to the Property and Related rate component.
- (2) Adjustments to the Property and Related rate component resulting from a change in ownership of depreciable assets shall be effective from the first day of the month following the date of sale.
- (c) Except in circumstances determined by the Division to constitute an emergency precluding a 60 day notice period, a provider applying for an adjustment pursuant to

this section is required to give 60 days written notice to the Division prior to the purchase of the asset. Such applications shall be exempt from the materiality test set out in section 5.101.8.7(b), but are subject to the other provisions of section 5.101.8.7. The burden is on the provider to document all information applicable to this adjustment and to demonstrate that any costs to be incurred are necessary and reasonable. When applicable, such documentation shall include the Certificate of Need application and all supporting financial information. The Division shall review the application and issue draft findings approving, denying, or proposing modifications to the adjustment applied for within 60 days of receipt of all information required. Providers may request review of the Division's decision in accordance with section 5.101.15 of these rules.

## 5.101.4.12 Repealed

## 5.101.4.13 Advertising Expenses

The Division shall recognize reasonable and necessary advertising expenses in accordance with the Nursing Facility Provider Manual.

## 5.101.4.14 Barber and Beauty Service Costs

The Division shall recognize costs related to barber and beauty services in accordance with the Nursing Facility Provider Manual.

#### 5.101.4.15 Bad Debt, Charity and Courtesy Allowances

Providers shall not include bad debts, charitable donations, or courtesy allowances as allowable costs.

## 5.101.4.16 Child Day Care

The Division may recognize reasonable and necessary costs related to providing child care services to employees in accordance with the Nursing Facility Provider Manual.

#### 5.101.4.17 Community Service Activities

The Division may recognize costs related to providing community service activities in accordance with the Nursing Facility Provider Manual.

## 5.101.4.18 Dental Services

The Division shall recognize costs related to dental services in accordance with the Nursing Facility Provider Manual.

## 5.101.4.19 <u>Legal Costs</u>

The Division shall recognize costs related to legal fees in accordance with the Nursing Facility Provider Manual.

## 5.101.4.20 Litigation and Settlement Costs

The Division shall recognize litigation and settlement costs, including costs related to challenges of the Division's decisions, in accordance with the Nursing Facility Provider Manual.

## 5.101.4.21 Motor Vehicle Allowance

The Division shall recognize costs to operate motor vehicles necessary to meet the needs of the facility in accordance with the Nursing Facility Provider Manual.

# 5.101.4.22 Non-Competition Agreement Costs

Amounts paid to the seller of an on-going facility by the purchaser for an agreement not to compete are considered capital expenditures. The amortized costs for such agreements are not allowable.

# 5.101.4.23 Compensation of Owners, Operators, or their Relatives

The Division shall recognize compensation for owners or operators of facilities, or their relatives, in accordance with the Nursing Facility Provider Manual..

# 5.101.4.24 Management Fees and Home Office Costs

- (a) Management fees, home office costs and other costs incurred by a nursing facility for similar services provided by other entities shall be included in the Indirect Cost category. These costs are subject to the provisions for allowable costs, allocation of costs and related party transactions contained in these rules and shall include property and related costs incurred for the management company. These costs are allowable only if such costs would be allowable if a nursing facility provided the services for itself.
- (b) Allowable costs shall be limited to five percent of the total net allowable costs less reported management fees, home office, or other costs, as defined in this section.

#### 5.101.4.25 Membership Dues

Reasonable and necessary membership dues, including any portions used for lobbying activities, shall be considered Medicaid allowable costs, provided the organization's function and purpose are directly related to providing resident care.

## 5.101.4.26 Post-Retirement Benefits

The Division may recognize costs related to certain retired personnel in accordance with the Nursing Facility Provider Manual.

## 5.101.4.27 Public Relations

Costs incurred for services, activities and events that are determined by the Division to be for public relations purposes will not be allowed.

## 5.101.4.28 Related Party

The Division shall disallow costs related to a related party expense in accordance with the Nursing Facility Provider Manual. The Division may request that the provider or the related party submit information, books, and records related to related party expenses.

## 5.101.4.29 Revenues

The Division shall disallow costs related to revenues the facility receives for selling goods or services in accordance with the Nursing Facility Provider Manual.

# 5.101.4.30 Travel/Entertainment Costs

The Division shall allow costs related for meals, lodging, transportation, and incidentals incurred for purposes related to resident care in accordance with the Nursing Facility Provider Manual.

# 5.101.4.31 <u>Transportation Costs</u>

- (a) Costs for ambulance services for emergency transportation or for transportation home from a nursing facility are covered pursuant to other rules promulgated by the Agency of Human Services and are not allowable under these rules.
- (b) The Division shall recognize costs of transportation that a facility incurs, other than costs described in subsection (a) of this section, in accordance with the Nursing Facility Provider Manual.

## 5.101.4.32 Services Directly Billable

Allowable costs shall not include the cost of services to individual residents which are ordinarily billable directly to Medicaid irrespective of whether such costs are payable by Medicaid.

# 5.101.5 Reimbursement Standards

## 5.101.5.1 Prospective Case-Mix Reimbursement System

- (a) The Division shall operate a prospective case-mix reimbursement system that accounts for some residents being more costly to care for than others. The Division shall require providers to assess and classify residents in accordance with the Nursing Facility Provider Manual. The Division shall weight the relative costs of caring for different classes of residents to determine an average case-mix score at each facility.
- (b) Repealed.
- (c) Per diem rates shall be prospectively determined for the rate year based on the allowable operating costs of a facility in a Base Year, plus property and related and ancillary costs from the most recently settled cost report, calculated as described in section 5.101.9.2.

## 5.101.5.2 Retroactive Adjustments to Prospective Rates

- (a) In general, a final rate may not be adjusted retroactively.
- (b) The Division may retroactively revise a final rate under the following conditions:
  - (1) as an adjustment pursuant to sections 5.101.8 and 5.101.10;
  - (2) in response to a decision by the Secretary pursuant to section 5.101.15.5 or to an order of a court of competent jurisdiction, whether or not that order is the result of a decision on the merits, or as the result of a settlement pursuant to section 5.101.15.8;
  - (3) for mechanical computation or typographical errors;
  - (4) for a terminating facility or a facility in receivership, pursuant to sections 5.101.5.10, 5.101.8.3, and 5.101.10.2;
  - (5) as a result of revised findings resulting from the reopening of a settled cost report pursuant to section 5.101.3.5;
  - (6) in those cases where a rate includes payment for Ancillary services and the provider subsequently arranges for another Medicaid provider to provide and bill directly for these services;
  - (7) recovery of overpayments, or other adjustments as required by law or duly promulgated regulation;
  - (8) when a special rate is revised pursuant to section 5.101.14.1(e)(2).

# 5.101.5.3 Lower of Rate or Charges

- (a) At no time shall a facility's Medicaid per diem rate exceed the provider's average customary charges to the general public for nursing facility services in semi-private rooms at the beginning of the calendar quarter. In this section, "charges" shall mean the amount actually required to be paid by or on behalf of a resident (other than by Medicaid, Medicare Part A or the Department of Veterans Affairs) and shall take into account any discounts or contractual allowances.
- (b) It is the duty of the provider to notify the Division within 10 days of any change in its charges.
- (c) Rates limited pursuant to paragraph (a) shall be revised to reflect changes in the provider's average customary charges to the general public effective on the latest of the following:
  - (1) the first day of the month in which the change to the provider's charges is made if the changes is effective on the first day of the month,
  - (2) the first day of the quarter after the effective date of the change to the provider's charges if the change to the provider's charges is not effective on the first day of the quarter, or
  - (3) the first day of the following quarter after the receipt by the Division of notification of the change pursuant to paragraph (b).

#### 5.101.5.4 Interim Rates

- (a) The Division may set interim rates for any or all facilities. The notice of an interim rate is not a final order of the Division and is not subject to review or appeal pursuant to any provision of these rules or 33 V.S.A. § 909.
- (b) Any overpayments or underpayments resulting from the difference between the interim and final rates will be either refunded by the provider or paid to the provider.

5.101.5.5 Repealed

Repealed.

5.101.5.6 Repealed

Repealed.

# 5.101.5.7 Occupancy Level

- (a) A facility should maintain an annual average level of occupancy at a target occupancy established in the Nursing Facility Provider Manual.
- (b) For facilities with less than the target occupancy amount, the number of total resident days at the target occupancy amount shall be used, pursuant to section 5.101.7, in determining the per diem rate for all categories except the Nursing Care and Ancillary categories.
- (b) The target occupancy amount provision in paragraph (b) shall be waived for facilities with 20 or fewer beds or terminating facilities pursuant to section 5.101.5.10, and when appropriate, for facilities operating under a receivership pursuant to section 5.101.8.3.
- (c) Decreasing the Number of Licensed Beds For any facility that operated at less than the target occupancy amount during the period used as the cost basis for any rate component subject to subsection (b) which subsequently reduces the number of licensed beds, the minimum occupancy shall be calculated based on the number of the facility's licensed beds on the first day of the quarter after the facility notifies the Division of such reduction.

# 5.101.5.8 Inflation Factors

The Director shall adjust each component of the rate by an inflation factor in accordance with a procedure established in the Nursing Facility Provider Manual.

## 5.101.5.9 Costs for New Facilities

- (a) For facilities that are newly constructed, newly operated as nursing facilities, or new to the Medicaid program, the prospective case-mix rate shall be determined based on budget cost reports submitted to the Division and the greater of the estimated resident days for the rate year or the resident days equal to the target occupancy amount established under section 5.101.5.7(a) of these rules of all beds used or intended to be used for resident care at any time within the budget cost reporting period. This rate shall remain in effect no longer than one year from the effective date of the new rate. The principles on allowability of costs and existing limits in sections 5.101.4 and 5.101.7 shall apply.
- (b) The costs reported in the budget cost report shall not exceed reasonable budget projections (adjusted for inflation and changes in interest rates as necessary) submitted in connection with the Certificate of Need.
- (c) Property and related costs included in the rate shall be consistent with the property and related costs in the approved Certificate of Need.

5.101

(d) At the end of the first year of operation, the prospective case-mix rate shall be revised based on the provider's actual allowable costs as reported in its annual cost report filed pursuant to section 5.101.3.2 for its first full fiscal year of operation.

# 5.101.5.10 Costs for Terminating Facilities

- (a) When a nursing facility plans to discontinue all or part of its operation, the Division may adjust its rate so as to ensure the protection of the residents of the facility.
- (b) A facility applying for an adjustment to its rate pursuant to this section must have a transfer plan approved by the Department of Disabilities, Aging and Independent Living, a copy of which shall be supplied to the Division.
- (c) An application under this section shall be made on a form prescribed by the Director and shall be accompanied by a financial plan demonstrating how the provider will meet its obligations set out in the approved transfer plan.
- (d) In approving such an application the Division may waive the minimum occupancy requirements in section 5.101.5.7, the limitations on costs in section 5.101.7, or make such other reasonable adjustments to the facility's reimbursement rate as shall be appropriate in the circumstances. The adjustments made under this section shall remain in effect for a period not to exceed six months.

## 5.101.6 Base Year Cost Categories for Nursing Facilities

## 5.101.6.1 General

In the case-mix system of reimbursement, allowable costs are grouped into cost categories. The accounts to be used for each cost category shall be prescribed by the Director. The Base Year costs shall be grouped into the following cost categories:

# 5.101.6.2 Nursing Care Costs

Providers shall allot appropriate costs to the Nursing Care component of the rate in accordance with the Nursing Facility Provider Manual.

## 5.101.6.3 Resident Care Costs

Providers shall allot appropriate costs to the Resident Care component of the rate in accordance with the Nursing Facility Provider Manual.

## 5.101.6.4 Indirect Costs

Providers shall allot appropriate costs to the Indirect component of the rate in accordance with the Nursing Facility Provider Manual.

## 5.101.6.5 Director of Nursing

Providers shall allot appropriate costs to the Director of Nursing component of the rate in accordance with the Nursing Facility Provider Manual.

# 5.101.6.6 Property and Related

Providers shall allot appropriate costs to the Property and Related component of the rate in accordance with the Nursing Facility Provider Manual.

## 5.101.6.7 Ancillaries

Providers shall allot appropriate costs to the Ancillaries component of the rate in accordance with the Nursing Facility Provider Manual.

# 5.101.7 Calculation of Costs, Limits and Rate Components for Nursing Facilities

The Division shall calculate base year costs, rates, and category limits pursuant to the Nursing Facility Provider Manual.

## 5.101.7.1 Repealed

Repealed.

## 5.101.7.2 Repealed

Repealed.

## 5.101.7.3 Resident Care Base Year Rate

The Division shall compute Resident Care Base Year rates in accordance with the Nursing Facility Provider Manual.

#### 5.101.7.4 Indirect Base Year Rate

The Division shall compute Indirect Base Year rates in accordance with the Nursing Facility Provider Manual.

# 5.101.7.5 <u>Director of Nursing Base Year Rate</u>

The Division shall compute the Director of Nursing Base Year per diem rates in accordance with the Nursing Facility Provider Manual.

# 5.101.7.6 Ancillary Services Rate

The Division shall compute the Ancillary per diem rate in accordance with the Nursing Facility Provider Manual.

## 5.101.7.7 Property and Related Per Diem

The Division shall compute The Property and Related per diem rate in accordance with the Nursing Facility Provider Manual.

## 5.101.7.8 <u>Limits Final</u>

Once a final order has been issued for all facilities' Base Year cost reports, notwithstanding any subsequent changes to the cost report findings, resulting from a reopening, appeal, or other reason, any caps on increases in the Nursing Care component, the Resident Care component, or the Indirect component set forth in the Nursing Facility Provider Manual will not change until nursing home costs are rebased pursuant to 5.6(b), except for annual adjustment by the inflation factors or a change in law necessitating such a change.

## 5.101.8 Adjustments to Rates

## 5.101.8.1 Change in Services

The Division, on application by a provider, may make an adjustment to the prospective case-mix rate for additional costs which are directly related to:

- (a) a new health care project previously approved under the provisions of 18 V.S.A. § 9434. Costs greater than those approved in the Certificate of Need (as adjusted for inflation) will not be considered when calculating such an adjustment,
- (b) a change in services, facility, or new health care project not covered under the provisions of 18 V.S.A. § 9434, if such a change has previously been approved by the Division, or
- (c) with the prior approval of the Division, a reduction in the number of licensed beds.

#### 5.101.8.2 Change in Law

The Division may make or a provider may apply for an adjustment to a facility's prospective case-mix rate for additional costs that are a necessary result of complying with changes in applicable federal and state laws, and regulations, or the orders of a State agency that specifically requires an increase in staff or other expenditures.

## 5.101.8.3 Facilities in Receivership

- (a) The Division, on application by a receiver appointed pursuant to state or federal law, may make an adjustment to the prospective case-mix rate of a facility in receivership for the reasonable and necessary additional costs to the facility incurred during the receivership.
- (b) On the termination of the receivership, the Division shall recalculate the prospective case-mix rate to eliminate this adjustment.

# 5.101.8.4 Efficiency Measures

The Division, on application by a provider, may make an adjustment to a prospective case-mix rate for additional costs which are directly related to the installation of energy conservation devices or the implementation of other efficiency measures, if they have been previously approved by the Division.

#### 5.101.8.5 Interest Rates

- (a) A provider may apply for an adjustment to the Property and Related rate, or the Division may initiate an adjustment if there are cumulative interest rate increases or decreases of more than one-half of one percentage point because of existing financing agreements with a balloon payment or a refinancing clause that forces a mortgage to be refinanced at a different interest rate, or because of a variable rate of adjustable rate mortgages.
- (b) A provider with an interest rate adjustment shall notify the Division of any change in the interest rate within 10 days of its receipt of notice of that change. The Division may rescind all interest rate adjustments of any facility failing to file a timely notification pursuant to this section for a period of up to two years.

#### 5.101.8.6 Emergencies and Unforeseeable Circumstances

- (a) The Division, on application by a provider, may make an adjustment to the prospective case-mix rate under emergencies and unforeseeable circumstances, such as damage from fire or flood.
- (b) Providers must carry sufficient insurance to address adequately such circumstances.

## 5.101.8.7 Procedures and Requirements for Rate Adjustments

- (a) Providers must apply for rate adjustments in accordance with this rule. The Director shall decide to grant, deny, or grant in part any application for a rate adjustment in their sole discretion.
- (b) Except for applications made pursuant to section 5.101.4.11, no application for a rate adjustment should be made if the change to the rate would be smaller than one percent of the rate in effect at the time.

- (a) Application for a Rate Adjustment shall be made on a form prescribed by the Director and filed with the Division and shall be accompanied by all documents and evidence determined necessary for the Division to make a decision.
- (d) The burden of proof is at all times on the provider to show that the costs for which the adjustment has been requested are reasonable, necessary and related to resident care.
- (e) The Division may grant or deny the Application, or make an adjustment modifying the provider's proposal. If the materials filed by the provider are inadequate to serve as a basis for a reasonable decision, the Division shall deny the Application, unless additional proofs are submitted.
- (f) The Division shall not be bound in considering other Applications, or in determining the allowability of reported costs, by any prior decision made on any Application under this section. Such decisions shall have no precedential value either for the applicant facility or for any other facility.
- (g) For adjustments requiring prior approval of the Division, such approval should be sought before the provider makes any commitment to expenditures. An Application for Prior Approval is subject to the same requirements as an Application for a Rate Adjustment under this section.
- (h) Rate adjustments made under this section shall be effective from the first day of the quarter following the date of the final order on the application or following the date the assets are actually put into service, whichever is the later, and may be continued, at the discretion of the Division, notwithstanding a general rebase of costs. Costs which are the basis for a continuing rate adjustment shall not be included in the cost categories used as the basis for the other rate components.
- (i) The Division may require an applicant for a rate adjustment under this section or under section 5.101.4.11 to file a budget cost report in support of its application.
- (j) When determined to be appropriate by the Division, a budget rate may be set for the facility according to the procedures in and subject to the provisions of section 5.101.5.9. Appropriate cases may include, but are not limited to, changes in the number of beds, major changes in the services delivered to residents, an addition to the facility, or the replacement of existing property.
- (k) In calculating an adjustment under this section and section 5.101.4.11, the Division may take into account the effect of such changes on all the cost categories of the facility.

- 5.101
- (l) A revision may be made prospectively to a rate adjustment under this section and section 5.101.4.11 based on a "look-back" which will be computed based on a provider's actual allowable costs.
- (m)In this section "additional costs" means the incremental costs of providing resident care directly and proximately caused by one of the events listed in this section or section 5.101.4.11. Increases in costs resulting from other causes will not be recognized. It is not intended that this section be used to effect a general rebase in a facility's costs.

## 5.101.8.8 Limitation on Availability of Rate Adjustments

Providers may not apply for a rate adjustment under this section for the sole reason that actual costs incurred by the facility exceed the rate of payment.

# 5.101.9 Private Nursing Facility and State Nursing Facility Rates

The Medicaid per diem payment rates for nursing home services are calculated according to the Nursing Facility Provider Manual.

5.101.9.1 Repealed

Repealed.

5.101.9.2 Repealed

Repealed.

5.101.9.3 <u>Repealed</u>

Repealed.

#### 5.101.9.4 State Nursing Facilities

- (a) Notwithstanding any other provisions of these rules, payment rates for state nursing facilities shall be determined retrospectively by the Division based on the reasonable and necessary costs of providing those services as determined using the cost reporting and cost finding principles set out in sections 5.101.3 and 5.101.4 of these rules.
- (b) Until such time as the cost report is settled, the Division shall set an interim rate based on an estimate of the facility's costs and census for the rate year.
- (c) After reviewing the facility's cost report, the Division shall set a final rate for the fiscal year based on the facility's allowable costs. If there has been an under payment for the period the difference shall be paid to the facility. If there has been an overpayment the excess payments shall be recouped.

# 5.101.9.5 Quality Incentives

The Division may make certain awards to facilities that provide a superior quality of care in an efficient and effective manner. The process for making these awards is described in the Nursing Facility Provider Manual.

# 5.101.10 Extraordinary Financial Relief

## 5.101.10.1 Objective

In order to protect Medicaid recipients from the closing of a nursing facility in which they reside, this section establishes a process by which nursing homes that are in immediate danger of failure may seek extraordinary financial relief. This process does not create any entitlement to rates in excess of those required by 33 V.S.A. Chapter 9 or to any other form of relief.

## 5.101.10.2 Nature of the Relief

- (a) Based on the individual circumstances of each case, the Director may recommend any of the following on such financial, managerial, quality, operational or other conditions as she or he shall find appropriate: a rate adjustment, an advance of Medicaid payments, other relief appropriate to the circumstances of the applicant, or no relief.
- (b) The Director's Recommendation shall be in writing and shall state the reasons for the Recommendation. The Recommendation shall be a public record.
- (c) The Recommendation shall be reviewed by the Secretary who shall make a Final Decision, which shall not be subject to administrative or judicial review.
- (d) In those cases where the Division determines that financial relief may be appropriate, such relief may be implemented on an interim basis pending a Final Decision by the Secretary. The interim financial relief shall be taken into account in the Division's Recommendation to the Secretary and in the Secretary's Final Decision.

#### 5.101.10.3 Criteria to be Considered by the Division

- (a) Before a provider may apply for extraordinary financial relief, its financial condition must be such that there is a substantial likelihood that it will be unable to continue in existence in the immediate future.
- (b) The following factors will be considered by the Director in making the Recommendation to the Secretary:
  - (1) the likelihood of the facility's closing without financial assistance,
  - (2) the inability of the applicant to pay bona fide debts,

- (3) the potential availability of funds from related parties, parent corporations, or any other source,
- (4) the ability to borrow funds on reasonable terms,
- (5) the existence of payments or transfers for less than adequate consideration,
- (6) the extent to which the applicant's financial distress is beyond the applicant's control,
- (7) the extent to which the applicant can demonstrate that assistance would prevent, not merely postpone the closing of the facility,
- (8) the extent to which the applicant's financial distress has been caused by a related party or organization,
- (9) the quality of care provided at the facility,
- (10) the continuing need for the facility's beds,
- (11) the age and condition of the facility,
- (12) other factors found by the Director to be material to the particular circumstances of the facility, and
- (13) the ratio of individuals receiving care in a nursing facility to individuals receiving home- and community-based services in the county in which the facility is located.

#### 5.101.10.4 Procedure for Application

- (a) An Application for Extraordinary Financial Relief shall be filed with the Division according to procedures to be prescribed by the Director.
- (b) The Application shall be in writing and shall be accompanied by such documentation and proofs as the Director may prescribe. The burden of proof is at all times on the provider. If the materials filed by the provider are inadequate to serve as a basis for a reasoned recommendation, the Division shall deny the Application, unless additional proofs are submitted.
- (c) The Secretary shall not be bound in considering other Applications by any prior decision made on any Application under this section. Such decisions shall have no precedential value either for the applicant facility or for any other facility.

## 5.101.11 Payment for Out-of-State Providers

# 5.101.11.1 Long-Term Care Facilities Other Than Rehabilitation Centers

Payment for services, other than Rehabilitation Center services, provided to Vermont Medicaid residents in long-term care facilities in another state shall be at the per diem rate established for Medicaid payment by the appropriate agency in that state. Payment of the per diem rate shall constitute full and final payment, and no retroactive settlements will be made.

# 5.101.11.2 Rehabilitation Centers

- (a) Payment for prior-authorized Rehabilitation Center services provided in nursing facilities located outside Vermont for the severely disabled, such as head injured or ventilator dependent people, will be made at the lowest of:
  - (1) the amount charged; or
  - (2) the Medicaid rate, including ancillaries as paid by at least one other state agency in CMS Region I.
- (a) Payment for Rehabilitation Center services which have not been prior authorized by the Commissioner of the Department of Vermont Health Access or a designee will be made according to section 5.101.11.1.

#### 5.101.11.3 Pediatric Care

No Medicaid payments will be made for services provided to Vermont pediatric residents in outof-state long-term care facilities without the prior authorization of the Commissioner of the Department of Vermont Health Access.

## 5.101.12 Rates for ICF/IIDs

Vermont does not currently license any Intermediate Care Facilities for the Intellectually Disabled (ICF/IDs). The Division shall reimburse out-of-state ICF/IDs according to the Medicaid rate established by the state in which the ICF/ID is located.

5.101.12.1 Repealed

Repealed.

5.101.12.2 Repealed

Repealed.

# 5.101.13 Rates for Swing Beds and Other Long-Term Care Services in Hospitals

Payment for swing-bed and other long-term care services provided by hospitals, pursuant to 42 U.S.C. § 1396l(a), shall be made at a rate equal to the average rate per diem during the previous calendar year under the State Plan to nursing facilities located in the State of Vermont. Supplemental payments made pursuant to section 5.101.14 and section 5.101.9.5 shall not be included in the calculation of swing-bed rates.

## 5.101.14 Special Rates for Certain Individual Residents

# 5.101.14.1 Availability of Special Rates for Individuals with Unique Physical Conditions

The Division may grant a special rate for the care of an individual with unique physical conditions whose physical conditions make it otherwise extremely difficult to obtain appropriate long-term care. The process for applying for, calculating, and receiving this special rate is stated in the Nursing Facility Provider Manual.

# 5.101.14.2 Special Rates for Certain Former Patients of the Vermont State Hospital

The Division may grant a special rate for the care of an individual who was transferred directly from the Vermont State Hospital or to a resident who has a documented history of severe behaviors that prevent them from being placed in a nursing home. The process for applying for, calculating, and receiving this special rate is stated in the Nursing Facility Provider Manual.

# 5.101.14.3 Special Rates for Medicaid Eligible Individuals in the Custody of the Department of Corrections

The Division may grant a special rate for the care of an individual who is transferred directly from the custody of the Department of Corrections. The process for applying for, calculating, and receiving this special rate is stated in the Nursing Facility Provider Manual.

## 5.101.15 Administrative Review and Appeals

## 5.101.15.1 Draft Findings and Decision

- (a) Before issuing findings on any Desk Review, Audit of a Cost Report, or decision on any application for a rate adjustment, the Division shall serve a draft of such findings or decision on the affected provider. If the Division makes no adjustment to a facility's reported costs or application for a rate adjustment, the Division's findings shall be final and shall not be subject to appeal under this section.
- (b) The provider shall review the draft upon receipt. If it desires to review the Division's work papers, it shall file, within 10 days, a written Request for Work Papers on a form prescribed by the Director.

# 5.101.15.2 Request for an Informal Conference on Draft Findings and Decisions

- (a) Within 15 days of receipt of either the draft findings or decision or requested work papers, whichever is the later, a provider that is dissatisfied with the draft findings or decision issued pursuant to section 5.101.15.1(a) may file a written Request for an Informal Conference with the Division's staff on a form prescribed by the Director.
- (b) Within 10 days of the receipt of the Request, the Division shall contact the provider to arrange a mutually convenient time for the informal conference, which shall be held within 45 days of the receipt of the Request at the Division. The informal conference may be held by telephone. At the conference, if necessary, a date certain shall be fixed by which the provider may file written submissions or other additional necessary information. Within 20 days thereafter, the Division shall issue its official agency action.
- (c) A Request for an Informal Conference must be pursued before a Request for Reconsideration can be filed pursuant to section 5.101.15.3. Issues not raised in the Request for Informal Conference shall not be raised at the informal conference or in any subsequent proceeding arising from the same action of the Division, including appeals pursuant to 33 V.S.A. §909.
- (d) Should no timely Request for an Informal Conference be filed within the time period specified in section 5.101.15.2(a), the Division's draft findings and/or decision are final and no longer subject to administrative review or judicial appeal.

# 5.101.15.3 Request for Reconsideration

- (a) A provider that is aggrieved by an official action issued pursuant to section 5.101.15.2(b) may file a Request for Reconsideration.
- (b) A Request for Reconsideration must be pursued before an appeal can be taken pursuant to 33 V.S.A. 909(a).
- (c) The Request for Reconsideration must be in writing, on a form prescribed by the Director, and filed within 30 days of the provider's receipt of the official action.
- (d) Within 10 days of the filing of a Request for Reconsideration, the provider must file the following:
  - (1) A request for a hearing, if desired;
  - (2) A clear statement of the alleged errors in the Division's action and of the remedy requested including: a description of the facts on which the Request is based, a memorandum stating the support for the requested relief in this rule, CMS-15, or other authority for the requested relief and the rationale for the requested remedy; and

- (3) If no hearing is requested, evidence necessary to bear the provider's burden of proof, including, if applicable, a proposed revision of the Division's calculations, with supporting work papers.
- (e) Issues not raised in the Request for Reconsideration shall not be raised later in this proceeding or in any subsequent proceeding arising from the same action of the Division, including appeals pursuant to 33 V.S.A. §909.
- (f) If a hearing is requested, within 10 days of the receipt of the Request for Reconsideration, the Division shall contact the provider to arrange a mutually agreeable time.
- (g) The hearing shall be conducted by the Director or her or his designee. The testimony shall be under oath and shall be recorded either stenographically or on tape. If the provider so requests, the Division staff involved in the official action appealed shall appear and testify. The Director, or her or his designee, may hold the record open to a date certain for the receipt of additional materials.
- (h) The Director shall issue a Final Order on Request for Reconsideration no later than 30 days after the record closes. Pending the issuance of a final order, the official action issued pursuant to section 5.101.15.2(b) shall be used as the basis for setting an interim rate from the first day of the calendar quarter following its issuance. Final orders shall be effective from the effective date of the official action.
- (i) Proceedings under this section are not subject to the requirements of 3 V.S.A. Chapter 25.

#### 5.101.15.4 Appeals from Final Orders of the Division

- (a) Within 30 days of the date thereof, a nursing facility aggrieved by a Final Order of the Division may file an appeal pursuant to 33 V.S.A. §909(a) and sections 5.101.15.5, 5.101.15.6 and 5.101.15.7 of this rule.
- (b) Within 30 days of the date thereof, a ICF/ MR aggrieved by a Final Order of the Division may file an appeal using the following procedures. Proceedings under this paragraph are not subject to the requirements of 3 V.S.A. Chapter 25.
  - (1) Request for Administrative Review by the Commissioner of Mental Health. The Commissioner or a designee shall review a final order of the Division of Rate Setting if a timely request is filed with the Director of the Division.
  - (2) Within 10 days of the receipt of the Request, the Director shall forward to the Commissioner a copy of the Request for Administrative Review and the materials that represent the documentary record of the Division's action.

- (3) The Commissioner or the designee shall review the record of the appeal and may request such additional materials as they shall deem appropriate, and shall, if requested by the provider, convene a hearing on no less than 10 days written notice to the provider and the Division. Within 45 days after the close of the record, the Commissioner or the designee shall issue a decision which shall be served on the provider and the Division.
- (4) Appeal to the Secretary of Human Services. Within 20 days of the date of the date of issuance, an ICF/MR aggrieved by the Commissioner's decision, may appeal to the Secretary.
- (c) The Notice of Appeal shall be filed with the Commissioner, who, within 10 days of the receipt of the Notice, shall forward to the Secretary a copy of the Notice and the record of the Administrative Review.
  - (1) The Secretary or his designee shall review the record of the Administrative Review and may, within their sole discretion, hold a hearing, request more documentary information, or take such other steps to review the Commissioner's decision as shall seem appropriate.
  - (2) Within 60 days of the filing of the Notice of Appeal or the closing of the record, whichever is the later, the Secretary or the designee shall issue a Final Determination.
- (d) Further review of the Final Determination is available only pursuant to Rule 75 of the Vermont Rules of Civil Procedure.

# 5.101.15.5 Request for Administrative Review to the Secretary of Human Services Pursuant to 33 V.S.A. § 909(a)(3)

- (a) No appeal may be taken under this section when the remedy requested is retrospective relief from the operation of a provision of this rule or such other relief as may be outside the power of the Secretary to order. Such relief may be pursued by an appeal to the Vermont Supreme Court or Superior Court pursuant to 33 V.S.A. §909(a)(1) and (2), or prospectively by a request for rulemaking pursuant 3 V.S.A. §806.
- (b) Appeals under this section shall be governed by the relevant provisions of the Administrative Procedures Act, 3 V.S.A. §§809-815.
- (c) Proceedings under this section shall be initiated by filing two copies of a written Request for Administrative Review with the Division, on forms prescribed therefor.
- (d) Within 5 days of receipt of the Request, the Director shall forward one copy to the Secretary. Within 10 days thereafter, the Secretary shall designate an independent appeals officer who shall be a registered or certified public accountant. The Letter of Designation shall be served on all parties to the appeal. All documents filed thereafter

shall be filed directly with the independent appeals officer and copies served on all parties.

- (e) Within 10 days of the designation of an independent appeals officer, the Division shall forward to him or her those materials that represent the documentary record of the Division's action.
- (f) Within 30 days thereafter, the independent appeals officer shall, on reasonable notice to the parties, convene a prehearing conference (which may be held by telephone) to consider such matters as may aid in the efficient disposition of the case, including but not limited to:
  - (1) the simplification of the issues,
  - (2) the possibility of obtaining stipulations of fact and/or admissions of documents which will avoid unnecessary proof,
  - (3) the appropriateness of prefiled testimony,
  - (4) a schedule for the future conduct of the case.

The independent appeals officer shall make an order which recites the action taken at the conference, including any agreements made by the parties.

- (g) The independent appeals officer shall hold a hearing, pursuant to 3 V.S.A. §809, on no less than 10 days written notice to the parties, according to the schedule determined at the prehearing conference. The independent appeals officer shall have the power to subpoena witnesses and documents and administer oaths. Testimony shall be under oath and shall be recorded either stenographically or on tape. Prefiled testimony, if admitted into evidence, shall be included in the transcript, if any, as though given orally at the hearing. Evidentiary matters shall be governed by 3 V.S.A. §810.
- (h) The independent appeals officer may allow or require each party to file Proposed Findings of Fact which shall contain a citation to the specific part or parts of the record containing the evidence upon which the proposed finding is based. The Proposed Findings shall be accompanied by a Memorandum of Law which shall address each matter at issue.
- (i) Within 60 days after the date of the hearing, or after the filing of Proposed Findings of Fact, whichever is the later, the independent appeals officer shall file with the Secretary a Recommendation for Decision, a copy of which shall be served on each of the parties. The Recommendation for Decision shall include numbered findings of fact and conclusions of law, separately stated, and a proposed order. If a party has submitted Proposed Findings of Fact, the Recommendation for Decision shall include a ruling upon each proposed finding. Each party's Proposed Findings and Memorandum of Law shall accompany the Recommendation.

- (j) At the time the independent appeals officer makes her or his Recommendation, she or he shall transmit the docket file to the Secretary. The Secretary shall retain the file for a period of at least one year from the date of the Final Determination in the docket. In the event of an appeal of the Secretary's Final Determination to the Vermont Supreme Court or to Superior Court, the Secretary shall make disposition of the file as required by the applicable rules of civil and appellate procedure.
- (k) Any party aggrieved by the Recommendation for Decision may file Exceptions, Briefs, and if desired, a written Request for Oral Argument before the Secretary. These submissions shall be filed with the Secretary within 15 days of the date of the receipt of a copy of the Recommendation and copies served on all other parties.
- (1) If oral argument is requested, within 20 days of the receipt of the Request for Oral Argument, the Secretary shall arrange with the parties a mutually convenient time for a hearing.
- (m) Within 45 days of the receipt of the Recommendation or the hearing on oral argument, whichever is the later, the Secretary shall issue a Final Determination which shall be served on the parties.
- (n) Parties may appeal the Final Determination of the Secretary pursuant to 33 V.S.A. §909(a)(1) and (2) and sections 5.101.15.6 and 5.101.15.7 of this Rule.

#### 5.101.15.6 Appeal to Vermont Supreme Court pursuant to 33 V.S.A. § 909(a)(1)

Proceedings under this section shall be initiated, pursuant to the Vermont Rules of Appellate Procedure, as follows:

- (a) by filing a Notice of Appeal from a Final Order with the Division; or
- (b) by filing a Notice of Appeal from a Final Determination with the Secretary.

#### 5.101.15.7 Appeal to Superior Court pursuant to 33 V.S.A. § 909(a)(2)

De novo review is available in the Superior Court of the county where the nursing facility is located. Such proceedings shall be initiated, pursuant to Rule 74 of the Vermont Rules of Civil Procedure, as follows:

- (a) by filing a Notice of Appeal from a Final Order with the Division; or
- (b) by filing a Notice of Appeal from a Final Determination with the Secretary.

#### 5.101.15.8 Settlement Agreements

The Commissioner of the Department of Vermont Health Access or their designee may agree to settle reviews and appeals taken pursuant to sections 5.101.15.3 and 5.101.15.5, and, with the approval of the Secretary, may agree to settle other appeals taken pursuant to 33 V.S.A. § 909 and any other litigation involving the Division on such reasonable terms as she or he may deem appropriate to the circumstances of the case.

# Ex. A Proposed Long-Term Care Reimbursement Manual

# [Cover page: Gainwell, GreenMountainCare, and DVHA branding; title: Nursing Facility Provider Manual]

#### [Table of contents]

#### Section 1: Introduction

This manual supplements existing federal and state law, primarily including the Division of Rate Setting's *Methods, Standards and Principles for Establishing Medicaid Payment Rates for Long-Term Care Facilities* (V.D.R.S.R.). This manual applies to nursing facilities licensed by the Department of Disabilities, Aging, and Independent Living that provide care to Medicaid residents and that receive per diem Medicaid rates from the Division under V.D.R.S.R. § 1.2.

This manual describes the main aspects of the rate-setting process, including how and when providers must file cost reports (Section 3), how the Division determines which costs are allowable (Section 4), any adjustments and caps that may apply to a facility (Section 5), how the Division allocates costs within categories (Section 6), how the Division calculates rate components (Section 7), how the Division calculates final per diem rates for private facilities (Section 8), and how providers can apply for special rates for specific residents (Section 9).

#### **Section 2: General Provisions**

#### Section 2.1: Representation in DRS Matters

Nursing facility providers can be complex organizations with multiple administrative staff, attorneys, accountants, or other personnel who may need or desire to receive notices of decisions that affect the provider. Accordingly, the V.D.R.S.R. Section 1.11 requires providers to notify the Division of all personnel who shall receive notices of Division decisions.

Providers may identify both general and special representatives to receive notices. A general representative is an individual who shall receive notice of all Division decisions with respect to that provider. A special representative is an individual who shall receive notice of all Division decisions with respect to one matter, such as an appeal of an adjustment or a request for a rate adjustment.

Providers may select more than one representative of either type. Providers may select only nursing home administrators, licensed attorneys, or certified public accountants as their representatives, and may not be represented by laypeople or clinicians.

Providers must use forms the Division has created to identify general and special representatives. To download copies of the forms, <u>visit the Division's website</u>. The forms may include additional requirements, such as listing a representative's address and title

and requiring the representative to affirm that they have the authority to receive notifications from the Division.

When a provider names a representative, the representative must select if they wish to be served documents by mail, fax, or email. If a provider chooses to be served documents by fax or email, they consent to the risks of electronic communications, including the risk that an email may be unsent because of file size limitations.

If a provider has multiple representatives that share a physical address and request notice by mail, the Division shall not send duplicate copies of documents to that physical address.

#### **Section 2.2: Procurement Standards**

Providers must establish and maintain a code of standards to assess the performance of employees who procure goods and services. The standards must provide, to the extent practicable, that the provider values open and free competition among multiple vendors. Providers should participate in group purchasing plans where feasible.

If a provider pays more than what the Division determines to be a competitive bid for a good or service, any amount over a lower bid that cannot be demonstrated to be a reasonable and necessary expenditure that satisfies the prudent buyer principle will not be an allowable cost under the Division's rules or this manual.

#### Section 2.3: Cost Allocation Plans and Changes in Accounting Principles

#### Section 2.3.1: In General

- (a) Any cost allocated to the nursing facility must be a reasonable cost.
- (b) The preferred statistical methods for allocating specific costs are as follows:
  - (1) Nursing supplies and salaries: direct costs
  - (2) Plant operations: square footage
  - (3) Utilities: square footage
  - (4) Laundry: pounds of laundry
  - (5) Dietary: number of resident days
  - (6) Administrative and general costs: accumulated costs
  - (7) Property and related costs: square footage
  - (8) Fringe benefits: direct allocation/gross salaries

#### Section 2.3.2: Cost Allocation From Related Entities

A provider's corporate parent may allocate costs from related entities to the provider. If this occurs, when a provider submits their cost report under Section 3 of this manual, the provider must include a summary of the allocated costs and a reconciliation of the allocated costs with the audited financial statements required by Section 3. If the related entity is the provider's home office or management company, the provider's cost report must include a Home Office Cost Statement.

#### Section 2.3.3: Recognizing Changes in Accounting Principles

The Division reserves the right not to recognize a change in accounting principles, methods, or bases of cost allocation if the Division finds that the change was intended to, or likely will, increase the provider's Medicaid payments.

#### Section 2.3.4: Medicare Intermediary Requirements

Providers must notify the Division if the Medicare intermediary requires them to change how they allocate costs or keep records.

#### **Section 2.3.5: Dietary Costs Calculation**

Food costs included in allocated dietary costs are calculated by dividing the facility's allocated dietary costs by the total organization dietary costs, both of which include allocated overhead, and multiplying the results by the total organization food costs.

#### Section 2.3.6: Utility Costs Calculation

Utility costs included in allocated plant operation and maintenance costs are calculated by dividing the facility's plant operation and maintenance costs by total organization plant operation and maintenance cost, both of which include allocated overhead, and multiplying the result by the total organization utility costs.

#### Section 2.3.7: Allocated Cost Categories

- (a) All administrative and general costs that are allocated to a facility, including home office and management company costs, shall be included in the Indirect cost category.
- (b) The capital component of goods or services purchased or allocated from a related or unrelated party, such as plant operation and maintenance, utilities, dietary, laundry, housekeeping, and all others, shall be considered as costs for that particular good or service and not classified as Property and Related costs of the nursing facility.

#### Section 3: Cost Reports

The Division's rules, V.D.R.S.R. § 3, require providers to file annual statistical and financial information in the form of a uniform report called a cost report. The cost report is the basis for all of the work done by the Division in setting a facility's rate. Providers should carefully ensure that they only report actual and allowable costs under this manual and the Division's rules. If the Division fails to correct an error on a provider's cost report that results in an increased rate, the provider may be liable for Medicaid waste, abuse, or fraud, or for filing a false claim.

The cost report template <u>is available on the Division's website</u>. Providers must file cost reports using this template. If a provider fails to file a cost report using this template, the

Division shall reject the cost report. Cost reports must be signed by the facility's owner or by the owner's authorized representative.

In addition to the template cost report, providers must submit additional supporting documentation to justify the reported costs. If a provider fails to file this supporting documentation, the Division shall reject the cost report. Providers must submit:

- Audited financial statements, except that the Director may waive this requirement in writing.
- The provider's most recently filed Medicare Cost Report, including the required supplemental data on CMS Form 339.
  - If a provider does not participate in Medicare, this requirement does not apply.
  - If a provider is a hospital-based nursing home, the provider shall submit the Medicare Cost Report for the same fiscal year as the cost report required by this manual and the Division's rules.
- An independent auditor's adjusting entries and reconciliation of the audited financial statements to the cost report, except that if the Director waived the requirement to submit an audited financial statement, this requirement does not apply.

The Division may also request other data, statistics, or information as necessary to carry out its functions. If the Division requests other information while reviewing a provider's cost report, but the provider fails to submit the requested information, the requested information shall not be admissible at any other stage of the rate setting process, including any subsequent appeal of a final decision of the Division under Section 15 of the Division's rules.

#### Section 3.1: Deadlines for filing a cost report

All providers must file a cost report at least annually for the 12-month period that covers the provider's fiscal year. The Division may also request that a provider file a special cost report or a budget cost report covering a shorter or greater period of time. Providers must file an acceptable annual cost report according to the deadlines prescribed in Section 3.3(a) of the Division's rules.

#### Section 3.2: Extensions for filing a cost report

The Division may grant an extension to a provider who is unable to file an acceptable annual cost report according to the Section 3.3(a) deadline. To receive an extension, providers must file a request in writing on a form prescribed by the Division. Forms are available on <a href="the Division's website">the Division's website</a>. The Division must receive the request before the deadline specified in Section 3.3(a). The request must clearly state the reason that the provider is requesting an extension and the date on which the Division will receive the cost report. If a request for an extension fails to meet these criteria, the Division shall reject the request for an extension.

The Division shall grant an extension only for good cause. Under this manual, "good cause" means a substantial reason that affords a legal excuse for a delay, an intervening action beyond the provider's control, or both. For example, the Division may find good cause exists for delay if a natural disaster prevents the cost report preparer from reporting to work in person or electronically, or a ransomware attack prevents the provider from accessing its records. The Division shall not find that good cause exists for extending a cost report deadline if the reason for the delay is ignorance of the rule, the inconvenience of preparing a cost report, or because the person who typically prepares the cost report is busy with other work.

#### Section 3.3: Reopening and correcting a cost report

The Division shall review cost reports under V.D.R.S.R. Section 3.4, and cost reports are settled under V.D.R.S.R. Section 3.5(a). After a cost report is settled, the Division may reopen and correct a cost report for certain narrow reasons.

#### Section 3.3.1: Reopening

Upon request. The Division may reopen a cost report if the provider requests that the Division reopen the report and submits new and material evidence concerning an element of the cost report, unless the Division requested that evidence under Section 3 of this manual during its review of the cost report and the provider failed to submit it.

The Division shall only reopen a cost report at a provider's request if doing so would have a material effect on the provider's Medicaid rate payments. Reopening would have a material effect if the provider's rate payments would be adjusted by one percent or more.

Providers must file a request to reopen a cost report within 3 years of the date of the Division's final determination with respect to a cost report.

Upon receiving an order. The Division shall reopen a cost report if a court of competent jurisdiction or the Secretary of the Agency of Human Services orders that a Division decision is inconsistent with applicable law or rules. The Division shall reopen the cost report whether or not doing so would have a material effect under this section.

The Division shall reopen a cost report in response to an order within 3 years of the date of the order, or within a time period as specified by the order.

For cause. The Division may reopen a cost report for cause, as established elsewhere in this manual.

#### Section 3.3.2: Correction

After the Division has reopened a cost report, the cost report may be corrected. A correction is a revision to a finding with respect to any aspect of an otherwise final cost report. The Division may request that a provider submit a corrected cost report, or the Division may correct a cost report itself in response to the information the provider

submitted upon requesting to reopen the cost report. If the cost report has been reopened in response to an order from a court or the Secretary, the Division shall correct the cost report itself.

The Division may require or allow a cost report to be corrected to address material errors in the cost report or to comply with applicable law or rules.

#### Section 4: Allowable Costs

After receiving a cost report, the Division shall determine the allowability and reasonableness of the costs a provider reports as described in V.D.R.S.R. Section 4.1. In general, if the Division's rules, this manual, or CMS's Medicare Provider Reimbursement Manual (CMS-15) do not address whether a cost is allowable, the Division shall review the cost in accordance with Generally Acceptable Accounting Practices (GAAP).

This manual addresses specific categories of costs and addresses whether they are allowable or unallowable.

#### **Section 4.1: Non-Recurring Costs**

Certain costs are non-recurring costs that shall be capitalized, amortized, and carried as an ongoing adjustment as described in V.D.R.S.R. Section 4.3.

In general. The Division may designate any reasonable, resident-related cost that exceeds \$10,000 and that is not expected to recur at least annually as a non-recurring cost. If the cost would otherwise be assigned to a cost category that is subject to a limit or cap, the cost will continue to be subject to that limit or cap.

Litigation expenses. If the Division recognizes a litigation expense under Section 4.14 of this manual of \$10,000 or more, the Division shall designate the expense a non-recurring cost.

Overseas recruitment. If a provider incurs more than \$2,000 in lump sum costs for recruitment costs, including legal fees, associated with hiring nurses from countries outside the United States, the Division shall designate the expense a non-recurring cost under certain conditions. The Division shall only designate the expense as non-recurring if providers pay recruited nurses at least the prevailing salary or wage, including benefits, that employed nurses of similar qualifications and experience in the geographic area in which the facility is located would be paid.

Nurse aide training expenses. If Vermont Medicaid is required to reimburse for nurse aide training expenses under 42 C.F.R. § 483.152(c)(2), the Division shall designate the expense a non-recurring cost.

#### Section 4.2: Interest Expense

V.D.R.S.R. Section 4.4(a) requires the Division to allow interest expenses that are necessary and proper.

#### Section 4.2.1: Necessary Interest

Interest shall only be treated as "necessary" under the Division's rules when interest is incurred on a loan to satisfy a provider's financial need and the provider had a legal obligation to pay the interest. The Division shall find that a provider did not have a financial need when the provider had 60 days or more cash, or cash equivalents, on hand to pay for expenses at the time the provider took out the loan.

#### Cash equivalents include:

- monetary investments, including unrestricted grants and gifts,
- non-monetary investments unrelated to resident care that can be readily converted into cash, net of any associated liabilities or fees,
- receivables from members or owners of the corporate entity that controls the nursing home, officers, managers, employees, or related parties of the person or entity that controls the nursing home, excluding education loans to employees, and
- receivables that result from transactions not related to resident care.

#### Cash equivalents exclude:

- funded depreciation that has been recognized by the Division under its rules or this manual, and
- restricted grants and gifts.

#### Section 4.2.2: Proper Interest

Interest expense shall only be treated as "proper" under the Division's rules when providers incur the interest at a rate not in excess what a prudent buyer would have had to pay in the money market existing at the time the loan was made.

Interest paid as part of a transaction with a related party or parties is not proper interest, unless:

- The interest expense relates to a first or second mortgage, or to assets leased from a related party where the costs to the related party are recognized in lieu of rent, or
- The interest rate is no higher than the rate charged by lending institutions at the inception of the loan.

Interest paid with respect to a capital expenditure in property, plant, or equipment that is related to resident care that requires approval from any governmental body, and for which the necessary approval was not granted, is not proper interest.

Interest on loans that do not include reasonable and ordinary principal repayments in the debt service payments is not proper interest, except to the extent that it would have been incurred pursuant to a standard amortization schedule for a term equivalent to the useful life of the asset.

#### Section 4.2.3: Offset

Interest expenses shall be reduced by realized investment income, except where that income is from funded depreciation that has been recognized by the Division under its rules or this manual, or where that income is from any grants or gifts. If the provider incurs interest expenses from working capital, the Division shall only offset these expenses using interest income derived from working capital.

#### Section 4.2.4: Cost Category

Interest may be included in a provider's property costs if the interest is necessary and proper under the Division's rule and this manual and the provider incurs the interest as a result of financing an acquisition of fixed assets related to resident care. The provider must put the asset in use within 60 days of financing the acquisition unless the provider receives a Certificate of Need for the acquisition from the Green Mountain Care Board or the Division approves the acquisition under Section 4.11 of the Division's rules. If the provider does not put an asset in use within 60 days of financing the asset's acquisition, the interest cost shall be included in the provider's indirect costs for the entire term of the loan.

Interest expense on any debt incurred for a purpose other than acquiring an asset shall be recognized as working capital interest expense and included in a provider's indirect costs.

#### Section 4.2.5 Limit on Borrowing

Borrowing to finance an additional asset or assets cannot a exceed certain amount, calculated as follows. The Division must determine the basis of the asset or assets and the principal amount of the loan to finance the asset or assets. The Division shall then determine the amount of cash or cash equivalents the provider has in excess of the amount the provider needs to pay expenses for the next 60 days. The Division shall subtract the amount of excess cash from the lower of the basis of the asset or the principal amount of the loan to finance the asset. This calculation shall serve as the limit on borrowings related to the asset.

When determining the basis of the asset under this section of the manual only, the Division may recognize other costs related to acquiring the asset, including bank finance charges, points and costs for legal and accounting fees, and discounts on debentures and letters of credit, and may add a proportional amount of those costs to the basis of the asset. When determining the basis of an asset in general, the Division shall follow Section 4.3 of this manual.

If borrowing to finance an additional asset exceeds the amount calculated under this section, the excessive amount is not allowable.

#### Section 4.2.6 Application of Principal Payments

For loans entered into before a facility's 1998 fiscal year, principal payments shall be applied first to loan balances on allowable borrowings and second to non-allowable loan balances.

For loans entered into during or after a facility's 1998 fiscal year, principal payments shall be applied to allowable and non-allowable loan balances on the ratio of each to the total amount of the loan.

#### Section 4.2.7 Refinancing

When refinancing debts, a provider must demonstrate that the costs of refinancing – including account fees, legal fees, and new debt acquisition costs – must be less than the allowable costs of the provider's current financing.

If the principal balance of a refinanced debt exceeds the principal balance of the previous debt plus accounting fees, legal fees, and debt acquisition costs, the Division shall consider the refinanced debt a working capital loan and must determine whether the loan is necessary under Section 4.2.1 of this manual. The provider may demonstrate the excess debt was incurred to acquire an asset under Sections 4.2.4 and 4.2.5 of this manual.

To the extent that a refinanced loan's principal includes material interest expense related to the original loan's unpaid interest charges, the refinanced loan's principal shall not be allowed.

#### Section 4.3: Determining the Basis of Property, Plant, and Equipment

The basis of a donated asset is the fair market value of the asset. For all other assets that a provider owns and uses in providing resident care, the basis of the asset is the lower of either the cost of the asset or the fair market value of the asset, unless another provision of this manual or the Division's rules specifies a different method for determining an asset's basis.

An asset's cost, under this section, includes the asset's purchase price, any applicable sales tax, and any costs to prepare the asset for its intended use, including but not limited to shipping, handling, installation, consulting, legal fees, and architectural fees.

#### Section 4.3.1: Basis of New Construction or Betterments and Improvements

Providers may construct new assets to provide resident care. The basis of a newly constructed asset's costs shall be determined from the costs of construction, which include:

 All direct costs, including but not limited to salaries and wages, related payroll taxes and fringe benefits, purchase price of materials, applicable sales taxes, shipping, handling, installation, permits, architectural fees, consulting fees, and legal fees,

- Indirect costs related to the construction of the asset,
- Interest costs related to capital indebtedness used to finance the construction of the asset and prepare it for its future use.

Providers may improve or better an asset. If an improvement or betterment extends the useful life of an asset two or more years, or significantly increases the productivity of an asset, the basis of the betterment or improvement shall be the costs of the improvement or betterment as if it was a new construction.

#### Section 4.3.2: Assets with Significant Basis and Useful Life

Providers must capitalize and depreciate any asset with a basis of \$2,000 or more and a useful life of two or more years in accordance with Section 4.4 of this manual. Providers must also capitalize and depreciate any groups of assets if the majority of the assets in the group have a basis of \$300 or more and a useful life of two or more years in accordance with Section 4.4 of this manual. Providers may choose to capitalize and depreciate any other assets if doing so would be reasonable.

#### Section 4.4: Requirement to Capitalize and Depreciate or Amortize Assets

Providers must compute depreciation and amortization on the straight-line method. The basis of each depreciated or amortized asset shall be the basis established under Section 4.3 and 4.5 of the manual, net of any salvage value of the asset.

In general, the Division estimates the useful life of an asset by referring to the most recent version of the Estimated Useful Lives of Depreciable Hospital Assets published by the American Hospital Association. If a provider has negotiated an arms-length lease of an asset, leasehold improvements may be amortized over the term of the lease if the term of the lease is shorter than the estimated useful life of the asset. The term of the lease includes any renewal period specifically stated in the lease.

#### Section 4.5: Change in Basis of Depreciable Asset After Change of Ownership

After a qualifying change in ownership, the Division may recognize a new basis for an asset.

#### Section 4.5.1: Qualifying Changes in Ownership

To benefit from this section, the change in ownership must meet each of the criteria (a) through (e):

- (a) The change of ownership was made for reasonable consideration,
- (b) The change of ownership was a bona fide transfer of all the powers and indicia of ownership,
- (c) All obligations to the State of Vermont that arise out of the transaction have been satisfied.
- (d) The change in ownership is in substance the sale of the assets or stock of the facility and not a method of financing, except that:

- (1) The transferor and transferee may enter into a financing agreement, but it must be constructed to effect a complete change of ownership. The Division shall monitor each party's compliance with the agreement and may refuse to recognize a change of ownership,
- (2) If the transferor forgives or reduces the debt of the transferee after the transaction is complete, the amount of the forgiveness or reduction shall be retroactively applied to the acquisition or basis of the asset, and
- (3) A change in ownership that is effected by transferring stock or shares of a publicly traded corporation shall not be recognized as a change in ownership under this section, and
- (e) The change in ownership did not occur between related parties or related individuals, except that the Division may approve a transaction between family members under the following conditions:
  - (1) The family members notify the Division at least two years before the sale with a description of the terms and conditions of the sale and a current appraisal of the facility being sold,
  - (2) The buyer demonstrates that they or their staff shall capably operate the facility according to state and federal standards,
  - (3) The seller shall not remain employed with the facility full-time, except for a transition period which shall not be longer than one year, and
  - (4) The seller may not have purchased the facility from any members of their family within the previous 12 years.

For the purposes of this section only, "family members" include spouses, parents, grandparents, children, grandchildren, brothers, sisters, aunts, uncles, nieces, and nephews, including by marriage, or such other familial relationships as the Director may reasonably determine.

#### Section 4.5.2: Change in Basis for Qualifying Transfers

When the Division recognizes a qualifying transfer under Section 4.5.1 of this manual, the basis of the assets for the new owner shall be determined as follows.

- (a) If the seller did not own the assets during the entire twelve-year period immediately preceding the change in ownership, or if the seller's facility did not receive Vermont Medicaid reimbursement during the entire twelve-year period immediately preceding the change in ownership, the depreciable cost basis of the transferred asset for the new owner shall be the lowest of:
  - (1) the fair market value of the assets,
  - (2) the acquisition cost of the asset to the buyer,
  - (3) the original basis of the asset to the seller as recognized by the Division, less accumulated depreciation.
- (b) If the seller owned the assets during the entire twelve year period immediately preceding the change in ownership and if the seller's facility received Vermont

Medicaid reimbursement during the entire twelve year period immediately preceding the change in ownership, the depreciable cost basis of the transferred fixed equipment and building improvements for individual assets having an original useful life of at least 20 years in agreement with the useful life assigned in the American Hospital Association guidelines, the depreciable cost basis of land improvements, the depreciable cost basis of buildings, and the cost basis of land for the new owner shall be the lowest of:

- (1) the fair market value of the assets,
- (2) the acquisition cost of the asset to the buyer,
- (3) the amount determined by the revaluation of the asset. An asset is revalued by increasing the original basis of the asset to the seller, as recognized by the Division, by an annual percentage rate. The annual percentage rate will be limited to the lower of:
  - (A) One-half the percentage increase in the Consumer Price Index (CPI) for All Urban consumers (United States City Average).
  - (B) One-half the percentage change in an appropriate construction cost index as determined by the Division of Rate Setting, which change shall not be greater than one-half of the percentage increase in the Dodge Construction index (or a reasonable proxy therefor) for the same period.
- (c) If the seller owned the assets during the entire twelve-year period immediately preceding the change in ownership and if the seller's facility received Vermont Medicaid reimbursement during the entire twelve year period immediately preceding the change in ownership, the depreciable cost basis of individual assets categorized as building improvements and fixed equipment with an original useful life of less than 20 years, in agreement with the useful life assigned in the American Hospital Association guidelines, shall be the seller's net book value and shall be depreciated over a useful life of seven years.
- (d) If the seller owned the assets during the entire twelve-year period immediately preceding the change in ownership and if the seller's facility received Vermont Medicaid reimbursement during the entire twelve-year period immediately preceding the change in ownership, the depreciable cost basis of moveable equipment and vehicles shall be the seller's net book value and shall be depreciated over a useful life of ten years.

#### Section 4.6: Leasing Arrangement for Property, Plant, and Equipment

In general, providers may not use rental or leasing arrangements to inflate their allowable costs above what their costs would have been had they purchased the same services at market price.

If a provider leases facilities or equipment from a related organization, the provider's rent expense shall be limited to Medicaid allowable interest, depreciation, insurance,

and taxes incurred for the year under review, or the price of comparable services or facilities purchased or leased elsewhere, whichever is lower.

Rent or lease charges, including sale and leaseback agreements for property, plant, and equipment that would otherwise be allowable, cannot exceed the amount which the provider would have included in allowable costs had it purchased or retained legal title to the asset, such as interest on mortgage, taxes, insurance, and depreciation.

#### Section 4.7: Depreciation Funding

In general, to incentivize providers to use depreciation funding to conserve assets, the Division shall not reduce allowable interest expense if a provider reports investment income on funded depreciation. Providers must maintain appropriate documentation to support the funded depreciation account and income earned on the account to be eligible for this relief.

However, if a provider uses funded depreciation for any purpose other than acquiring or replacing a nursing facility asset without Division approval, the Division shall offset investment income on funded depreciation for both the current cost report and may reopen settled cost reports for cause to reduce allowable interest expense and revise funded appreciation. Providers may seek Division approval to use funded depreciation to convert licensed nursing home beds to residential care or assisted living facility beds, or to construct new residential care or assisted living facility beds while concurrently reducing licensing nursing home beds.

If a provider deposits funds in a funded depreciation account without retaining sufficient working capital or resources to support ongoing operations, the Division shall not recognize the deposits as funded depreciation.

## Section 4.7.1: Depreciation Funding CMS-15 Exceptions for Replacement Reserves

Some lending institutions require funds to be set aside periodically to replace fixed assets. The periodic amounts set aside for this purpose are not allowable costs in the period when the provider expends those funds, but the Division will allow those costs when the provider withdraws and uses them either through depreciation or expense after the Division considers the usage of the funds. Because the replacement reserves are essentially the same as funded depreciation, the Division shall apply the same rules regarding interest income and expense.

If a provider leases a facility from an unrelated party and the ownership of the reserve rests with the lessor, then the replacement reserve payment becomes part of the lease payment, and the Division shall consider the replacement reserve payment as an allowable cost in the year the provider expends it. If the lessee is allowed to use a replacement reserve to replace the lessee's assets, the lessee shall not be allowed to depreciate the assets it purchases.

#### **Section 4.8: Advertising Expenses**

Reasonable and necessary expenses for advertisements to secure necessary employees are allowable costs. Providers may purchase advertisements in newspapers or other media circulated to the public for this purpose.

#### Section 4.9: Barber and Beauty Service Costs

Direct costs of barber and beauty services are not allowable costs. The fixed costs for space and equipment to provide barber and beauty services, as well as overhead associated with billing for barber and beauty services, are allowable costs.

#### Section 4.10: Child Day Care Costs

Reasonable and necessary costs that a provider incurs to provide day care services to children of employees performing resident-related functions are allowable costs. If a provider receives revenue for providing day care services to employee's children, their costs will be offset by this revenue. The direct and indirect expenses related to providing day care services to children of individuals who are not employees performing resident-related functions are not allowable costs.

For the Division to accurately review and allow costs under this section, providers must accumulate all day care costs in a separate cost center. Providers must identify revenues for providing day care for employees and non-employees in separate accounts.

#### **Section 4.11: Community Service Activities**

Providers may request permission from the Division to adjust direct identifiable incremental costs (food, direct labor, fringe benefits, and transportation) related to providing community service to local communities, such as providing meals to vulnerable adults, adult day care, certain respite care, etc. If a provider significantly expands a program as a result of providing community services, the Division may also adjust overhead costs. Providers must maintain auditable records for all direct identifiable incremental costs associated with providing community services under this section.

#### Section 4.12: Dental Services

Costs incurred for dental services for Medicaid-eligible individuals are covered under other Medicaid programs and are not allowable. However, the fixed costs for space and equipment related to providing dental services and overhead associated with billing for dental services are allowable.

#### Section 4.13: Legal Costs

Necessary, ordinary, and reasonable legal fees incurred for resident-related activities will be allowable.

#### Section 4.14: Litigation and Settlement Costs

All costs allowed under this section are non-recurring costs within the meaning of Section 4.1 of this manual.

In general, the Division will recognize attorneys' fees and other expenses incurred for litigation only to the extent that the costs are related to resident care, that the provider prevails, and that the costs are not covered by insurance. If a provider settles a matter before a jury or bench verdict (whether or not a lawsuit has been filed), the Division will recognize one-half of the costs, including attorneys' fees, settlement award, and other expenses to the extent that the costs are related to resident care and not covered by insurance.

Litigation and settlement costs incurred in response to criminal investigations and professional licensing matters are not related to resident care for the purposes of this section and are not allowable costs.

If a provider incurs attorneys' fees and other similar expenses when challenging a decision of the Division, the Division shall allow the costs to the extent that the provider prevails, as determined by a ratio of total dollars at issue in the case to the total dollars the provider is awarded.

#### Section 4.15: Motor Vehicle Allowance

The cost to operate a motor vehicle necessary to meet facility needs is allowable. If a vehicle is used for both personal and business purposes, the portion of vehicle costs associated with personal use is not allowable. If a provider fails to adequately document how much a vehicle is used for personal use and business use, the Division reserves the right to disallow all costs for the vehicle.

The Division shall allow motor vehicle costs under this section to the extent they reflect the cost of operating a similar size, mid-price vehicle in the same class.

#### Section 4.16: Compensating Owners, Operators, or their Relatives

The Division shall not allow costs related to compensating an owner, operator, or their relative who claims to provide administrative functions at any facilities that employ a full-time (40 hours per week or more) nursing home administrator or assistant administrator, except as authorized in this section.

The Division may allow compensation for an owner, operator, or their relative if the provider's cost report specifically reports the function the owner, operator, or their relative performed, the number of hours worked, and the number of employees supervised. The function that the owner, or operator, or their relative performed must be unique and unduplicated by another employee.

The maximum allowable salary for an owner, operator, or their relative who claims to provide administrative functions shall be 110% percent of the average of all reported

administrator salaries for Vermont nursing facilities participating in the Medicaid program for the reported fiscal year.

#### Section 4.17: Post-Retirement Benefits

If CMS-15 would allow costs for benefits to retired personnel, all such costs shall be included in fringe benefits and the Division shall allocate such costs accordingly.

#### Section 4.18: Related Party Expenses

If a provider pays otherwise allowable expenses to a related party, the Division shall disallow the costs, subject to the following exception. The Division may allow the costs if the provider identifies all related party expenses, the relationship the provider has with the related party, and all expenses attributable to the related party. The provider must also demonstrate that the related party expenses do not exceed the lower of the cost to the related party or the price of comparable services, facilities, or supplies that the provider could purchase elsewhere.

#### Section 4.19: Revenues

If a facility reports operating or non-operating revenues related to goods or services they provide, the costs to which those revenues correspond are not allowable. If the specific costs cannot be identified, the revenues shall be deducted from the most appropriate costs. If the revenues are more than the costs to which the revenues correspond, the deduction shall be equal to such costs.

#### Section 4.20: Travel/Entertainment Costs

Reasonable and necessary costs of meals, lodging, transportation, and incidentals incurred for purposes related to resident care are allowable. All costs that the Division determines are for the pleasure and convenience of the provider or the providers' representatives will not be allowed.

#### Section 4.21: Transportation Costs

Reasonable and necessary costs for transportation, other than ambulance services for emergency transportation or for transportation home from a nursing facility, that are related to the care of residents are allowable. Transportation costs shall include the depreciation of utility vehicles, mileage reimbursement to employees when employees use their privately owned vehicles to transport residents, and any contractual arrangements for providing transportation. Transportation costs shall not be separately billed for individual residents.

Transportation costs related to residents receiving kidney dialysis shall be reported in the Ancillary cost category. All other costs allowed under this section shall be reported in the Indirect cost category.

#### Section 5: Reimbursement Standards

#### Section 5.1: Resident Acuity Classification

The Division adopts this section of the Nursing Facility manual to comply with state law, 33 V.S.A. § 905(b)(1), which requires the Division to group residents into classes according to the similarity of their assessed conditions and needed services to incentivize facilities to admit residents that may be more costly to care for than others. To accomplish this, the Division shall operate a prospective case-mix reimbursement system.

#### Section 5.1.1: Classification Process

The Department of Aging and Independent Living, Division of Licensing and Protection, shall prescribe a form for assessing residents and classifying them into groups. Providers shall self-assess their residents according to this form.

#### Section 5.1.2: Categorization

Vermont Medicaid has adopted a variant of the Patient-Driven Payment Model (PDPM) adopted by the federal Centers for Medicare and Medicaid Services (CMS) for use in calculating Medicaid rates. Vermont Medicaid uses the nursing components of the PDPM model to categorize nursing home residents. The Division shall weight different categories of residents according to the table set out in Section 7.2 of this Manual.

#### Section 5.2: Base Years

A Base Year shall be a calendar year from January 1 through December 31. The Division originally rebased all costs on January 1, 2007, and has rebased all costs every four years thereafter. The Division rebased nursing costs every two years thereafter. The Division rebases ancillary costs and property and related costs every year.

When rebasing, the Director of the Division may require a facility to file a special cost report covering the calendar year that is the Base Year if the facility's fiscal year does not already run from January 1 through December 31. The Division may instead use the facility's fiscal year costs, adjusted for inflation in accordance with Section 5.4 of this Manual. If the Director of the Division requires a facility to prepare and file a special cost report, the cost of doing so is an allowable cost under the Division's rules and this Manual.

#### Section 5.3: Target Resident Occupancy

Vermont Medicaid has established a target occupancy rate of 80 percent. If a facility fails to maintain average resident days at or above this amount, the Division shall apply the provisions of V.D.R.S.R. § 5.7.

#### Section 5.4: Inflation Adjustment

The Division adopts this section of the Nursing Facility manual to comply with state law, 33 V.S.A. § 905, which requires the Division to adjust a facility's base year rates annually by reasonable and adequate inflation factors.

On June 1 of each year, the Director shall consult the most recent available publication of the Health Care Cost Service to calculate annual inflation adjustments and rebase inflation adjustments, if necessary.

The Division shall use different inflation factors to adjust different rate components and shall weigh subcomponents of each inflation factor in proportion to the percentage of actual allowable costs that Vermont facilities incur for each subcomponent. For example, if a cost in the Nursing Care cost component is 80 percent attributable to salaries and wages and 20 percent attributable to employee benefits, the Division shall weight these two subcomponents of the Nursing Care inflation factor at .8 and .2 respectively. The Division shall recalculate the weights for each inflation factor each time the relevant cost category is rebased.

#### Section 5.4.1: Nursing Care Component Inflation

The Division shall adjust the Nursing Care rate component by an inflation factor that uses two price indices to account for estimated economic trends with respect to two subcomponents of nursing care costs: (1) wages and salaries, and (2) benefits. The price indices for each subcomponent are the wages and salaries portion of the Health Care Cost Service NHMB and the employee benefits of the NHMB respectively. The Division shall additionally adjust the rate component by one percentage point for every 12 month period, prorated for fractions thereof, from the midpoint of the base year to the midpoint of the rate year.

#### Section 5.4.2: Resident Care Component Inflation

The Division shall adjust the Resident Care rate component by an inflation factor that uses four price indices to account for estimated economic trends with respect to four subcomponents of resident care costs: (1) wages and salaries, (2) benefits, (3) utilities, and (4) food and all other costs. The price indices for each subcomponent are the wages and salaries portion of the Health Care Cost Service NHMB, the employee benefits of the NHMB, the utilities portion of the NHMB, and the food portion of the NHMB respectively.

#### **Section 5.4.3: Indirect Component Inflation**

The Division shall adjust the Indirect rate component by an inflation factor that uses three price indices to account for estimated economic trends with respect to three subcomponents of indirect costs: (1) wages and salaries, (2) benefits, and (3) all other indirect costs. The price indices for each subcomponent are the wages and salaries

portion of the Health Care Cost Service NHMB, the employee benefits of the NHMB, the utilities portion of the NHMB, and the NECPI-U (all items) respectively.

#### **Section 5.4.4: Director of Nursing Component Inflation**

The Division shall adjust the Director of Nursing rate component by an inflation factor that uses two price indices to account for estimated economic trends with respect to two subcomponents of Director of Nursing costs: (1) wages and salaries, and (2) employee benefits. The price indices for each subcomponent are the wages and salaries portion of the Health Care Cost Service NHMB and the employee benefits of the NHMB respectively.

#### **Section 6: Base Year Cost Categories**

#### Section 6.1: In General

Providers should allot costs that they incur into appropriate accounts that relate to nursing care, resident care, indirect costs, director of nursing costs, property and related costs, and ancillary costs as described in this manual.

#### **Section 6.2: Nursing Care Costs**

Allowable costs in the Nursing Care component of the rate include the actual costs of licensed personnel providing resident care who are required to follow state and federal law. These personnel include (1) registered nurses, (2) licensed practical nurses, (3) certified or licensed nurse aides, including wages for initial and ongoing nurse training as required by the 1987 Omnibus Budget Reconciliation Act, (4) contract nursing, (5) the Minimum Data Set (MDS) coordinator, and (6) fringe benefits for the personnel listed in this section, including child day care as allowed by this manual.

The costs of unlicensed staff, including bedmakers, geriatric aides, transportation aides, paid feeding and dining assistants, ward clerks, and medical records librarians, are not Nursing Care costs. The salary and related benefits of the Director of Nursing shall be excluded from the Nursing Care costs and reimbursed separately.

#### Section 6.3: Resident Care Costs

Allowable costs in the Resident Care component of the rate include reasonable costs associated with expenses related to direct care, including:

- (a) Food, vitamins, and food supplements,
- (b) Utilities, including heat, electricity, sewer and water, garbage, liquid propane gas, or other required fuels,
- (c) Activities personnel, including recreational therapy and direct activity supplies,
- (d) Medical Director, Pharmacy Consultant, Geriatric Consultant, and Psychological or Psychiatric Consultant costs,
- (e) Counseling personnel, chaplains, art therapists, and volunteer stipends,

- (f) Social service workers,
- (g) Employee physicals,
- (h) Wages for paid feeding/dining assistants, for those hours that assistants are actually engaged in assisting residents with eating,
- (i) Fringe benefits, including child day care, for any personnel identified above, and
- (j) Any other items that the Director may prescribe by issuing a practice and procedure issued pursuant to subsection 1.8(d).

#### Section 6.4: Indirect Costs

Allowable costs in the Indirect component include reasonable costs reported in the following functional cost centers on the facility's cost report, including those extracted from a facility's cost report or the cost report of an affiliated hospital or institution:

- (a) Fiscal services,
- (b) Administrative services or professional fees,
- (c) Plant operation and maintenance,
- (d) Grounds,
- (e) Security,
- (f) Laundry and linen,
- (g) Housekeeping,
- (h) Medical records,
- (i) Cafeteria,
- (j) Seminars, conferences, or other in-service training, except that tuition for college credit in a discipline related to a staff member's employment or costs of obtaining a GED are treated as fringe benefits for that staff member,
- (k) Dietary, excluding food,
- (I) Motor vehicle costs,
- (m)Clerical costs, including ward clerks,
- (n) Transportation (excluding depreciation),
- (o) Insurance for director and officer liability, comprehensive liability, bond indemnity, malpractice, premises liability, motor vehicles, or any other costs of insurance incurred or required in the care of residents that has not been specifically addressed elsewhere,
- (p) Office supplies and telephone costs,
- (q) Conventions and meetings,
- (r) EDP bookkeeping and payroll,
- (s) Fringe benefits for staff employed in the roles listed in this section, including child day care, and
- (t) Any expense not specified for inclusion in another cost category, except that:
  - (1) The Director may specify that costs be reported in a different cost category in the instructions to the cost report, the chart of accounts, or by issuing a practice and procedure under V.D.R.S.R. § 1.8, and

(2) Vermont Medicaid reimburses prescription drug costs through other programs; therefore, these costs are not allowable and shall not be included in any cost category.

#### Section 6.5: Director of Nursing Costs

Allowable costs in the Director of Nursing component include a reasonable salary for one position and associated fringe benefits, including child day care.

#### Section 6.6: Property and Related Costs

- (a) Allowable costs in the Property and Related component include:
  - (1) Depreciation on buildings and fixed equipment, major movable equipment, minor equipment, computers, motor vehicles, land improvements, and amortization of leasehold improvements and capital leases,
  - (2) Interest on capital indebtedness,
  - (3) Real estate leases and rents,
  - (4) Real estate taxes and property taxes,
  - (5) All equipment, whether it is capitalized, expensed, or rented,
  - (6) Fire and casualty insurance, and
  - (7) Amortization of mortgage acquisition costs.
- (b) For any proposed change in services or facility, or for a new health care project, with projected property and related costs of \$250,000 or more, providers must give the Division written notice of the project no less than 60 days before commencing work on the project. The notice must include a detailed description of the project and a detailed estimate of all costs. If a provider fails to give the Division notice of a change or project as required by this section, the Division may refuse to allow the associated costs.

#### Section 6.7: Ancillaries

Allowable costs in the Ancillary component include:

- (a) All physical, speech, occupational, respiratory, and IV therapy services and therapy supplies, excluding oxygen. Medicaid allowable costs shall be based on the cost-tocharge ratio for these services. These therapy services shall not be allowable unless:
  - (1) The services are provided pursuant to a physician's order,
  - (2) The services are provided by a licensed therapist or other State certified or registered therapy assistant, a qualified IV professional, or other therapy aides,
  - (3) The services are not reimbursable by Medicare, and
  - (4) The provider records charges by payor class for all units of these services.
- (b) Medical supplies, whether or not the provider customarily records charges. For purposes of this section,
  - (1) "Medical supplies" includes, but is not limited to, oxygen, disposable catheters, catheters, colostomy bags, drainage equipment, trays, and tubing,

- (2) "Medical supplies" does not include rented or purchased equipment, with the exception of rented or purchased oxygen concentrators, which are medical supplies.
- (c) Over the counter drugs. All drug costs may be disallowed if a provider commingles the costs of prescription drugs with over-the-counter drugs.
- (d) Incontinence supplies and personal care items. These items include adult diapers, chux and other disposable pads, and personal care items such as toothpaste, shampoo, baby powder, combs, brushes, and similar products.
- (e) Dialysis transportation. The costs of transportation for Medicaid residents receiving kidney dialysis shall be included in the ancillary cost category. Allowable costs may include contract or other costs, but shall not include employee salaries or wages or costs associated with the use of provider-owned vehicles.
- (f) Overhead costs related to ancillary services and supplies.

#### Section 7: Calculating Costs, Limits, and Rate Components

After determining which costs are allowable, the Division calculates per diem costs for each cost category, sets base year rates, and sets limits on the amount that the base year rates can rise.

#### Section 7.1: Calculating Per Diem Costs

The Division calculates per diem costs for each cost category, except the Nursing Care and Ancillary cost categories, by dividing allowable costs for each case mix category by the greater of actual bed days of service rendered, including revenue generating hold/reserve days, or the number of resident days that the facility would have provided had the facility operated at the minimum occupancy standard set in Section 5.3 of this manual during the cost period under review.

Section 7.2: Nursing Care Component

Section 7.2.1: Case Mix Weights

There are **[XX]** case-mix resident classes. Each case-mix class has a specific case-mix weight as follows:

Group Code	Case-Mix Weight	Description	

#### Section 7.2.2: Average Case Mix Score

The Department of Disabilities, Aging and Independent Living's Division of Licensing and Protection shall compute each facility's average case-mix score. After each base year, the Division of Licensing and Protection shall certify the average case-mix score for all residents of each facility to the Division of Rate Setting. In between these base year certifications, the Division of Licensing and Protection shall certify the average case-mix score of all residents at each facility for whom Medicaid pays their room and board no less than quarterly.

#### Section 7.2.3: Calculating Nursing Care Costs Per Case Mix Point

The Division shall calculate each facility's Nursing Care cost per case mix point as follows:

- (a) The Division shall determine each provider's allowable Nursing Care costs using each facility's base year cost report.
- (b) The Division shall compute each facility's Standardized Resident Days by multiplying total resident days from the most recent base year by that facility's average case mix score for all residents for the four quarters of the cost reporting period under review.
- (c) The Division shall compute the per diem nursing care cost per case mix point by dividing total allowable Nursing Care costs by the Base Year Standardized Resident Days for that base year.

#### Section 7.2.4: Nursing Care Cost Limits

The Division shall array all nursing care facilities' allowable base year per diem Nursing Care costs per case mix point, excluding those for state nursing facilities and any nursing facilities that have stopped participating in the Medicaid program at the time the limits are set, from lowest to highest. These costs shall be limited to the cost at the ninety-fifth percentile calculated using the percentile spreadsheet function. Each facility's base year Nursing Care rate per case mix point shall be the lesser of the limit in this section or the facility's actual allowable Nursing Care costs per case mix point.

#### Section 7.3: Resident Care Base Year Rate

The Division shall compute Resident Care base year rates as follows:

- (a) The Division shall determine each provider's allowable Resident Care costs using each facility's base year cost report.
- (b) The Division shall calculate each facility's base year per diem allowable Resident Care costs by dividing the base year total allowable Resident Care costs by total base year resident days.
- (c) The Division shall array all nursing care facilities' allowable base year per diem Resident Care costs, excluding those for state nursing facilities and any nursing facilities that have stopped participating in the Medicaid program at the time the limits are set, from lowest to highest, then identify the median.

- (d) The per diem limit shall be the median plus twenty-five percent.
- (e) Each facility's base year Resident Care per diem rate shall be the lesser of the limit set in subsection (d) of this section or the facility's actual base year per diem allowable Resident Care costs.

#### Section 7.4: Indirect Base Year Rate

The Division shall compute Indirect base year rates as follows:

- (a) The Division shall determine each provider's allowable Indirect costs using each facility's base year cost report.
- (b) The Division shall calculate the base year per diem allowable Indirect costs for each facility by dividing the base year total allowable Indirect costs by total base year resident days.
- (c) The Division shall array all nursing care facilities' allowable base year per diem Indirect costs, excluding those for state nursing facilities and any nursing facilities that have stopped participating in the Medicaid program at the time the limits are set, from lowest to highest, then identify the median.
- (d) The Division shall set the per diem limit as follows:
  - (1) For special hospital-based nursing facilities, the limit shall be 137 percent of the median.
  - (2) For all other privately owned nursing facilities, the limit shall be 115 percent of the median.
- (e) Each provider's base year Indirect per diem rate shall be the lesser of the limit in subsection (d) of this section or the facility's actual base year per diem allowable costs.

#### Section 7.5: Director of Nursing Base Year Rate

The Division shall compute each facility's Director of Nursing base year per diem rates as follows:

- (a) The Division shall determine each provider's allowable Director of Nursing costs using each facility's base year cost report.
- (b) The Division shall calculate each facility's base year per diem allowable Director of Nursing costs by dividing the base year total allowable Director of Nursing costs by total base year resident days.
- (c) There shall be no limit on Director of Nursing per diem costs besides the basic requirement that these costs must be allowable.

#### Section 7.6: Ancillary Services Rate

The Division shall compute each facility's Ancillary per diem rates as follows:

- (a) The Division shall determine each facility's Medicaid ancillary costs.
- (b) The Division shall calculate each facility's Ancillary rate using each facility's most recently settled cost report as follows:

- (1) The Division shall calculate costs for therapy services per diem, including IV therapy, by dividing allowable Vermont Medicaid costs by the number of related Vermont Medicaid resident days less Vermont Medicaid hold days.
- (2) The Division shall calculate dialysis transportation costs per diem by dividing the allowable costs for Vermont Medicaid residents by the number of Vermont Medicaid resident days less Vermont Medicaid hold days.
- (3) The Division shall calculate costs for medical supplies, over the counter drugs, incontinent supplies, and personal care items per diem by dividing allowable costs for those services by total resident days, less hold days.
- (c) If the Division determines that a facility's Ancillary per diem rate should change as a result of its analysis, the Division shall implement that change at the time of the first quarterly case mix rate recalculation after the cost report is settled.

#### Section 7.7: Property and Related Per Diem Costs

The Division shall compute each facility's Property and Related per diem rates as follows:

- (a) Using each facility's most recently settled annual cost report, the Division shall determine each facility's total allowable Property and Related costs.
- (b) The Division shall calculate the facility's per diem Property and Related rate by dividing allowable Property and Related costs by total resident days.
- (c) If the Division determines that a facility's Property and Related per diem rate should change as a result of its analysis, the Division shall implement that change at the time of the first quarterly case mix rate recalculation after the cost report is settled.

#### Section 7.8: Total Per Diem Rate

#### Section 7.8.1: Nursing Facility Rate Components

The total Nursing Facility per diem rate of reimbursement consists of the following rate components:

- (a) Nursing Care,
- (b) Resident Care,
- (c) Indirect,
- (d) Director of Nursing,
- (e) Property and Related,
- (f) Ancillary, and
- (g) Adjustments (if any).

#### Section 7.8.2: Calculating the Total Rate

The Division shall identify all the rate components listed in Section 7.8.1 and adjust the components in accordance with Inflation Factors identified in this manual. The Division shall add all of the rate components together to arrive at the total per diem rate.

#### Section 7.8.3: Updating Rates for Change in Case-Mix Score

- (a) The Division shall update the Nursing Care rate component, including any rate adjustment that reimburses for Nursing Care costs, of each facility's rate quarterly, on the first day of January, April, July, and October, to reflect changes in the average case-mix score of the facility's Medicaid residents.
- (b) The Division shall calculate the updated Nursing Care rate component as follows:
  - (1) The Division shall divide the Nursing Care rate component, including any rate adjustment that reimburses for Nursing Care costs, by the average case-mix score used to determine the current Nursing Care rate component. This quotient is the current Nursing Care rate per case mix point.
  - (2) The current Nursing Care rate per case mix point is multiplied by the new average case mix score. This product is the new Nursing Care rate component.

#### Section 7.9: Quality Awards Program

The Division may make awards to facilities that provide superior care in a cost-effective manner under this section.

#### Section 7.9.1: Standards

The Division shall base any awards under this section on objective standards of:

- (a) quality, including resident satisfaction surveys, to be determined by the Department of Disabilities, Aging, and Independent Living, and
- (b) cost efficiency, to be determined by the Division in accordance with a practice and procedure that the Division shall issue.

#### Section 7.9.2: Purpose of Payments

Providers must use supplemental payments under the Quality Awards program to enhance the quality of care they provide to Medicaid-eligible residents. In determining how best to accomplish this goal, providers must consult with the facility's Resident Council. If a provider fails to comply with this section, the Division may recoup the supplemental payments when setting future per diem rates.

#### Section 7.9.3: Methodology for Distribution

- (a) Vermont Medicaid shall make quality incentive payments from a pool. The annual size of the pool shall be based on the amount of \$25,000, times the number of facilities meeting the award criteria, up to a maximum of 5 facilities and \$125,000.
- (b) The Division shall distribute the pool among the qualifying facilities, awarding each qualifying facility a share of the pool based on the ratio of the facility's Medicaid days to the total Medicaid days for all the qualifying facilities.

#### Section 7.9.4: Award Process and Criteria

(a) The Division shall apply the award criteria to facility data up to March 31 each year to determine eligibility for the awards, which the Division shall present before the end

- of the rate year. Facilities must participate in Vermont Medicaid and meet all of the award criteria.
- (b) The Department of Disabilities, Aging, and Independent Living shall rank all eligible facilities according to their quality of care. The five facilities with the highest quality of care will receive an award. If, based on the basic criteria, there are ties that would cause more than five facilities to be equally qualified, the tied facilities will be ranked according to the efficiency criteria to determine those facilities that will receive an award.
- (c) Basic criteria. The basic criteria are as follows:
  - (1) The most recent health survey report resulted in a score of five or less, no deficiency with a scope and severity greater than "D" level, and may not receive a deficiency of "D" level in each of the general categories of Quality of Care, Quality of Life, or Resident Rights.
  - (2) DAIL has not substantiated a complaint in the two most recent surveys related to quality of care, quality of life, or residents' rights.
  - (3) The facility must participate in a statewide quality improvement campaign approved by DAIL.
  - (4) Resident satisfaction surveys must record a result above the statewide average.
  - (5) The facility must receive a fire safety deficiency score of 5 or less with scope and severity less than "E" level in the most recent full survey.
- (d) Efficiency criteria. To resolve a tie under subsection (b) of this section, the Division shall determine each facility's allowable cost per day using each facility's most recently settled Medicaid cost report. The Division shall calculate the facility's cost per day using actual resident days for the same fiscal period. The Division shall resolve the tie in favor of the facility that had a lower cost per day.

#### **Section 8: Special Rates**

In rare and exceptional circumstances, an individual may be extremely difficult to place in appropriate long-term care settings. In these rare and exceptional circumstances, providers may apply for a special rate for that individual in accordance with this section.

#### Section 8.1: Individuals with Unique Physical Conditions

A special rate under this section is available subject to the conditions in the following subsections.

#### Section 8.1.1: Required Findings

Before Vermont Medicaid approves a special rate under Section 8.1, the following findings must be made.

(a) The Commissioner of the Department of Vermont Health Access, in consultation with that Department's Medical Director and the Director of DAIL's Adult Services Division or the ASD Director's designee, must make a written finding that the individual's care

- needs meet the requirements of this section and that the proposed placement is appropriate for the individual's needs,
- (b) The Division of Rate Setting, in consultation with the Commissioner of DVHA and the Commissioner of DAIL, must determine that the special rate that the Division has calculated under Section 8.1.3 is reasonable for the services provided.

#### Section 8.1.2: Plan of Care

- (a) Providers must submit a plan of care for the individual to the DVHA Medical Director and the DAIL Adult Services Director or the ASD Director's designee before they accept the individual for placement.
- (b) Providers must submit an updated plan of care for the individual at least every six months, or more frequently if there is a significant change in the resident's physical condition.
- (c) The DVHA Medical Director and the DAIL Adult Services Division Director must approve the plan of care and any updated plan of care.
- (d) The Division shall use the DVHA Medical Director and DAIL Adult Service Division Director's approval as a basis for determining whether a special rate should be granted, continued, or revised.

#### Section 8.1.3: Calculating the Rate

- (a) The Division shall set a per diem rate based on the budgeted allowable cost for the individual's plan of care. The per diem rate shall be exempt from any limits imposed elsewhere in this manual.
- (b) The Division may, from time to time, revise the special rate to reflect significant changes in the resident's assessment, care plan, and costs of providing care. The Division may revise the special rate prospective, retroactively, or both based on the actual allowable costs of providing care to the resident.
- (c) Special rates set under Section 8.1 shall not affect the facility's normal per diem rate.
  - (1) Any resident who receives a special rate under Section 8.1 shall not be included when the Division calculates a facility's average case-mix score.
  - (2) The Division shall include the resident's days of care in a facility's Medicaid days and total resident days when determining a facility's per diem rates and limits.
  - (3) Providers receiving a special rate under Section 8.1 shall track the total costs of providing care to the resident and shall self-disallow the incremental cost of providing that care on cost reports that cover the period during which the facility receives Medicaid payments for services to the resident.

#### Section 8.2: Individuals with Unique Mental and Emotional Conditions

A special rate is available under this section is available subject to the conditions in the following subsections. The special rate is available on a prospective basis only.

#### Section 8.2.1: Required Findings

The Commissioner of the Department of Mental Health or their designee must determine that a resident or prospective resident has a documented history of severe behaviors that prevent them from being placed in a nursing home. The resident must exhibit behaviors that would be significantly more challenging than those of the general nursing facility population.

#### Section 8.2.2: Plan of Care

- (a) Providers must submit a plan of care for the individual to the Department for Mental Health and the DAIL Adult Services Division before they accept the individual for placement.
- (b) The DMH Commissioner and DAIL's Division of Licensing and Protection must approve the plan of care and any updated plan of care before the individual is placed.
- (c) After the individual is placed at the facility, the facility must document that the transferred person continues to present significant behavior management problems by exhibiting behaviors that are significantly more challenging than those of the general nursing facility population, or that the transferred person's behavior has abated because of supports provided by the nursing home. The facility must submit this documentation to DAIL's Division of Licensing and Protection no less than annually.

#### Section 8.2.3: Available Rates

The special rate that the facility shall receive shall consist of the current per diem rate for the facility as calculated under the Division's rules and this manual, plus a monthly supplemental incentive payment. Vermont Medicaid makes three levels of supplemental payments available for the care of residents that meet the eligibility criteria in Section 8.2. DAIL's Commissioner shall determine the three levels of payments that are available. Facilities shall receive one of the three levels based on the severity of the resident's condition and the resources needed to provide care.

#### Section 8.3: Rates for Individuals in the Custody of the Department of Corrections

Vermont Medicaid may grant an incentive payment for a facility to care for an individual who is transferred directly from the custody of the Department of Corrections, whether serving as an inmate at an institution or on probation or parole. In general, the special rate shall be 150 percent of a nursing facility's ordinary Medicaid rate. Facilities must apply for this special rate on forms prescribed by the Division. Facilities must annually recertify that the individuals for which they receive an incentive payment under this section continue to be in the custody of the Department of Corrections.

#### The Vermont Statutes Online

The Vermont Statutes Online have been updated to include the actions of the 2023 session of the General Assembly.

NOTE: The Vermont Statutes Online is an unofficial copy of the Vermont Statutes Annotated that is provided as a convenience.

Title 33: Human Services

**Chapter 019: Medical Assistance** 

Subchapter 001: Medicaid

(Cite as: 33 V.S.A. § 1901)

#### § 1901. Administration of program

- (a)(1) The Secretary of Human Services or designee shall take appropriate action, including making of rules, required to administer a medical assistance program under Title XIX (Medicaid) and Title XXI (SCHIP) of the Social Security Act.
- (2) The Secretary or designee shall seek approval from the General Assembly prior to applying for and implementing a waiver of Title XIX or Title XXI of the Social Security Act, an amendment to an existing waiver, or a new state option that would restrict eligibility or benefits pursuant to the Deficit Reduction Act of 2005. Approval by the General Assembly under this subdivision constitutes approval only for the changes that are scheduled for implementation.
  - (3) [Repealed.]
- (4) A manufacturer of pharmaceuticals purchased by individuals receiving State pharmaceutical assistance in programs administered under this chapter shall pay to the Department of Vermont Health Access, as the Secretary's designee, a rebate on all pharmaceutical claims for which State-only funds are expended in an amount that is in proportion to the State share of the total cost of the claim, as calculated annually on an aggregate basis, and based on the full Medicaid rebate amount as provided for in Section 1927(a) through (c) of the federal Social Security Act, 42 U.S.C. § 1396r-8.
  - (b) [Repealed.]
- (c) The Secretary may charge a monthly premium, in amounts set by the General Assembly, per family for pregnant women and children eligible for medical assistance under Sections 1902(a)(10)(A)(i)(III), (IV), (VI), and (VII) of Title XIX of the Social Security Act, whose family income exceeds 195 percent of the federal poverty level, as permitted under section 1902(r)(2) of that act. Fees collected under this subsection shall be credited to the State Health Care Resources Fund established in section 1901d of this title and shall be available to the Agency to offset the costs of providing Medicaid services. Any co-payments, coinsurance, or other cost sharing to be charged shall also be authorized and set by the General Assembly.

- (d)(1) To enable the State to manage public resources effectively while preserving and enhancing access to health care services in the State, the Department of Vermont Health Access is authorized to serve as a publicly operated managed care organization (MCO).
- (2) To the extent permitted under federal law, the Department of Vermont Health Access shall be exempt from any health maintenance organization (HMO) or MCO statutes in Vermont law and shall not be considered to be an HMO or MCO for purposes of State regulatory and reporting requirements. The MCO shall comply with the federal rules governing managed care organizations in 42 C.F.R. Part 438. The Vermont rules on the primary care case management in the Medicaid program shall be amended to apply to the MCO except to the extent that the rules conflict with the federal rules.
- (3) The Agency of Human Services and Department of Vermont Health Access shall report to the Health Care Oversight Committee about implementation of Global Commitment in a manner and at a frequency to be determined by the Committee. Reporting shall, at a minimum, enable the tracking of expenditures by eligibility category, the type of care received, and to the extent possible allow historical comparison with expenditures under the previous Medicaid appropriation model (by department and program) and, if appropriate, with the amounts transferred by another department to the Department of Vermont Health Access. Reporting shall include spending in comparison to any applicable budget neutrality standards.

#### (e) [Repealed.]

- (f) The Secretary shall not impose a prescription co-payment for individuals under age 21 enrolled in Medicaid or Dr. Dynasaur.
- (g) The Department of Vermont Health Access shall post prominently on its website the total per-member per-month cost for each of its Medicaid and Medicaid waiver programs and the amount of the State's share and the beneficiary's share of such cost.
- (h) To the extent required to avoid federal antitrust violations, the Department of Vermont Health Access shall facilitate and supervise the participation of health care professionals and health care facilities in the planning and implementation of payment reform in the Medicaid and SCHIP programs. The Department shall ensure that the process and implementation include sufficient State supervision over these entities to comply with federal antitrust provisions and shall refer to the Attorney General for appropriate action the activities of any individual or entity that the Department determines, after notice and an opportunity to be heard, violate State or federal antitrust laws without a countervailing benefit of improving patient care, improving access to health care, increasing efficiency, or reducing costs by modifying payment methods. (Added 1967, No. 147, § 6; amended 1997, No. 155 (Adj. Sess.), § 21; 2005, No. 159 (Adj. Sess.), § 2; 2005, No. 215 (Adj. Sess.), § 308, eff. May 31, 2006; 2007, No. 74, § 3, eff. June 6, 2007; 2009, No. 156 (Adj. Sess.), § E.309.15, eff. June 3, 2010; 2009, No. 156 (Adj. Sess.), § I.43; 2011, No. 48, § 16a, eff. Jan. 1, 2012; 2011, No. 139 (Adj. Sess.), § 51, eff. May 14, 2012; 2011, No. 162 (Adj. Sess.), § E.307.6; 2011, No. 171 (Adj. Sess.), § 41c; 2013, No. 79, § 23, eff. Jan. 1, 2014; 2013, No. 79, § 46; 2013, No. 131 (Adj. Sess.), § 39, eff. May 20, 2014; 2013, No. 142 (Adj. Sess.), § 98; 2017, No. 210 (Adj. Sess.), § 3, eff. June 1, 2018.)

#### The Vermont Statutes Online

The Vermont Statutes Online have been updated to include the actions of the 2023 session of the General Assembly.

NOTE: The Vermont Statutes Online is an unofficial copy of the Vermont Statutes Annotated that is provided as a convenience.

Title 33: Human Services

**Chapter 009: Division of Rate Setting** 

(Cite as: 33 V.S.A. § 904)

#### § 904. Rate setting

- (a) The Director shall establish by rule procedures for determining payment rates for care of State-assisted persons to nursing homes and to such other providers as the Secretary shall direct. The Secretary shall have the authority to establish rates that the Secretary deems sufficient to ensure that the quality standards prescribed by section 7117 of this title are maintained, subject to the provisions of section 906 of this title. Beginning in State fiscal year 2003, the Medicaid budget for care of State-assisted persons in nursing homes shall employ an annual inflation factor that is reasonable and that adequately reflects economic conditions, in accordance with the provisions of Section 5.8 of the rules adopted by the Division of Rate Setting (Methods, Standards, and Principles for Establishing Medicaid Payment Rates for Long-Term Care Facilities).
- (b) No payment shall be made to any nursing home, on account of any State-assisted person, unless the nursing home is certified to participate in the State/federal medical assistance program and has in effect a provider agreement. (Added 1977, No. 204 (Adj. Sess.), § 1; amended 1981, No. 224 (Adj. Sess.), § 1, eff. May 4, 1982; 1989, No. 267 (Adj. Sess.), § 2, eff. July 1, 1991; 1995, No. 160 (Adj. Sess.), § 12; 1997, No. 61, § 270a; 2001, No. 63, § 99; 2013, No. 131 (Adj. Sess.), § 19, eff. May 20, 2014; 2021, No. 20, § 280.)



# Proposed Rules Postings A Service of the Office of the Secretary of State

- Vermont.gov
- Feedback
- SOS Website
- APA Site
- Code of Vermont Rules
- Recent
- Search Rules
- Calendar
- Subscribe
- APA Contact Info

#### Search Rules

#### **Deadline For Public Comment**

Deadline: Feb 15, 2024

The deadline for public comment has expired. Contact the agency or primary contact person listed below for assistance.

#### **Rule Details**

Rule Number:

24P006

Title:

Nursing Home Reimbursement Rule Simplification.

Type:

Standard

Status:

Proposed

Agency:

Agency of Human Services

Legal Authority:

33 V.S.A. § 1901(a), 33 V.S.A. § 904(a)

These rules strike all existing Division of Rate Setting rules and replace them in the Agency of Human Services' Health Care Administrative Rules. Some material, particularly language regarding which costs are allowable and how the

Summary:

language regarding which costs are allowable and how the Division applies various bonuses or penalties, is moved into a new manual. The remaining material primarily sets out the

administrative process for applying for, receiving, and

appealing per diem rates set by the Division.

Nursing home residents; Nursing home providers, including

Persons Affected: their trade association, the Vermont Health Care

Association; Other providers of residential care or facilities

for the aging and disabled, who may be indirectly impacted by Vermont Medicaid reimbursement policy for nursing homes; The Department of Disabilities, Aging, and Independent Living, which manages the Choices for Care budget and licenses and inspects nursing homes; and the Division of Rate Setting, which calculates and adjusts each nursing home's Medicaid rate and recommends changes to the rate-setting system.

The proposed change will not have a fiscal impact. Vermont Medicaid is simultaneously proposing a new manual that increases caps on cost growth, lowers minimum occupancy requirements, and transitions to a new resident classification system that will moderately impact the Vermont Medicaid

Choices for Care budget managed by the Department of Disabilities, Aging, and Independent Living. The state is

seeking legislative approval for these impacts.

Posting date: Jan 24,2024

#### **Hearing Information**

**Economic Impact:** 

There are not Hearings scheduled for this Rule

#### **Contact Information**

#### **Information for Primary Contact**

PRIMARY CONTACT PERSON - A PERSON WHO IS ABLE TO ANSWER QUESTIONS A

Level: Primary

Name: James LaRock

Agency: Department of Vermont Health Access, Agency of Human Services

Address: NOB 1 South, 280 State Drive

City: Waterbury

State: VT

Zip: 05671

Telephone: 802-241-0251 Fax: 802-241-0260

Email: james.larock@vermont.gov

SEND A COMMENT

Website https://humanservices.vermont.gov/rules-policies/health-care-rules/health-care-adminis

Address: VIEW WEBSITE

#### **Information for Secondary Contact**

SECONDARY CONTACT PERSON - A SPECIFIC PERSON FROM WHOM COPIES OF FIL MAY ANSWER QUESTIONS ABOUT FORMS SUBMITTED FOR FILING IF DIFFERENT FI

Level: Secondary

Name: Jaime Mooney

Agency: Department of Vermont Health Access, Agency of Human Services

Address: NO

NOB 1 South, 280 State Drive

City:

Waterbury

State: Zip:

VT 05671

Telephone: 802-798-2144

Fax:

802-241-0260

Email:

jaime.mooney@vermont.gov

SEND A COMMENT

### **Keyword Information**

Keywords:

Nursing home Medicaid Per diem rate Residential

Back

Accessbility Policy | Privacy Policy

A Vermont Government Website Copyright 2024 State of Vermont



# OFFICE OF THE SECRETARY OF STATE VERMONT STATE ARCHIVES & RECORDS ADMINISTRATION (VSARA)

(802) 828-2863

то:	Seven Days Legals ( <u>legals@sevendaysvt.com</u> )	Tel: (802) 865-1020 x110.	
	The Caledonian Record Julie Poutré (adv@caledonian-record.com)	Tel: 748-8121 FAX: 748-1613	
	Times Argus / Rutland Herald Melody Hudson ( <u>classified.ads@rutlandherald.com</u> ) Elizabeth Marrier <u>elizabeth.marrier@rutlandherald.com</u> )	Tel: 802-747-6121 ext 2238 FAX: 802-776-5600	
	The Valley News (advertising@vnews.com)	Tel: 603-298-8711 FAX: 603-298-0212	
	The Addison Independent (legals@addisonindependent.com)	Tel: 388-4944 FAX: 388-3100 Attn: Display Advertising	
	The Bennington Banner / Brattleboro Reformer  Lylah Wright ( <u>lwright@reformer.com</u> )	Tel: 254-2311 ext. 132 FAX: 447-2028 Attn: Lylah Wright	
	The Chronicle (ads@bartonchronicle.com)	Tel: 525-3531 FAX: 880-1040	
	Herald of Randolph (ads@ourherald.com)	Tel: 728-3232 FAX: 728-9275 Attn: Brandi Comette	
	Newport Daily Express (ilafoe@newportvermontdailyexpress.com)	Tel: 334-6568 FAX: 334-6891 Attn: Jon Lafoe	
	News & Citizen ( <u>mike@stowereporter.com</u> ) Irene Nuzzo (irene@newsandcitizen.com and ads@stowereporter .com removed from distribution list per Lisa Stearns.	Tel: 888-2212 FAX: 888-2173 Attn: Bryan	
	St. Albans Messenger Legals (legals@samessenger.com; cfoley@orourkemediagroup.com)	Tel: 524-9771 ext. 117 FAX: 527- 1948 Attn: Legals	
	The Islander (islander@vermontislander.com)	Tel: 802-372-5600 FAX: 802-372-3025	
	Vermont Lawyer (hunter.press.vermont@gmail.com)	Attn: Will Hunter	

**FROM:** APA Coordinator, VSARA **Date of Fax:** January 25, 2024 **RE:** The "Proposed State Rules" ad copy to run on **February 1, 2024** 

PAGES INCLUDING THIS COVER MEMO: 4

\*NOTE\* 8-pt font in body. 12-pt font max. for headings - single space body. Please include dashed lines where they appear in ad copy. Otherwise minimize the use of white space. Exceptions require written approval.

If you have questions, or if the printing schedule of your paper is disrupted by holiday etc. please contact VSARA at 802-828-3700, or E-Mail <a href="mailto:sos.statutoryfilings@vermont.gov">sos.statutoryfilings@vermont.gov</a>, Thanks.

#### PROPOSED STATE RULES

By law, public notice of proposed rules must be given by publication in newspapers of record. The purpose of these notices is to give the public a chance to respond to the proposals. The public notices for administrative rules are now also available online at <a href="https://secure.vermont.gov/SOS/rules/">https://secure.vermont.gov/SOS/rules/</a>. The law requires an agency to hold a public hearing on a proposed rule, if requested to do so in writing by 25 persons or an association having at least 25 members.

To make special arrangements for individuals with disabilities or special needs please call or write the contact person listed below as soon as possible.

To obtain further information concerning any scheduled hearing(s), obtain copies of proposed rule(s) or submit comments regarding proposed rule(s), please call or write the contact person listed below. You may also submit comments in writing to the Legislative Committee on Administrative Rules, State House, Montpelier, Vermont 05602 (802-828-2231).

VPharm Coverage.

Vermont Proposed Rule: 24P003

AGENCY: Agency of Human Services

CONCISE SUMMARY: This proposed rulemaking amends VPharm rule 5450 titled "Coverage" which establishes coverage for the VPharm program. This VPharm Coverage rule was last amended effective February 25, 2012. This amendment aligns with federal and state guidance and law, improves clarity, and makes technical corrections. Certain content, such as 5450.1 Rebate or Price Discount, has been removed as it was redundant with language that exists in state statute. Substantive revisions include: expanding drug coverage available under VPharm 2 and VPharm 3 to be equivalent to the drug coverage available under VPharm 1 and the Medicaid program, as authorized through Vermont's Global Commitment to Health 1115 Demonstration waiver effective July 1, 2022.

FOR FURTHER INFORMATION, CONTACT: Ashley Berliner, Department of Vermont Health Access, 280 State Drive, Waterbury, VT 05671-1000 Tel: 802-578-9305 Fax: 802-241-0450 E-Mail: <a href="mailto:ahs.medicaidpolicy@vermont.gov">ahs.medicaidpolicy@vermont.gov</a> URL: <a href="mailto:https://humanservices.vermont.gov/rules-policies/health-care-rules">https://humanservices.vermont.gov/rules-policies/health-care-rules</a>.

FOR COPIES: Danielle Fuoco, Department of Vermont Health Access, 280 State Drive, Waterbury, VT 05671-1000 Tel: 802-585-4265 Fax: 802-241-0450 E-Mail: danielle.fuoco@vermont.gov.

Rules Governing the Importation of Domestic Animals, Including Livestock and Poultry.

Vermont Proposed Rule: 24P004

AGENCY: Agriculture, Food & Markets

CONCISE SUMMARY: This is an update to the existing importation rules for livestock and poultry. The rule outlines the documentation and disease testing requirements to import cattle/bison, equine, swine, sheep, goats, poultry, ratites, psittacine birds, camelids, and cervids from the US 50 States and Canada. Disease epidemiology and testing have changed, this update addresses the significant diseases of concerns and the tests required to reasonably demonstrate free status. This update aligns the rule with USDA disease programs, modern technology, and the disease traceability standards set by Vermont statutes. Examples of the changes: current USDA brucellosis program standards are eliminating the Class A-C language for describing state status

and has increased the age recommendation for swine testing from four months to six months of age.

FOR FURTHER INFORMATION, CONTACT: Kaitlynn Levine, Vermont Agency of Agriculture, Food & Markets, 116 State Street, Montpelier, VT 05620, Tel: 802-636-7144 Email: AGR.FSCPRule@vermont.gov URL: <a href="https://agriculture.vermont.gov/rule-governing-importation-livestock-and-poultry-rule-98074">https://agriculture.vermont.gov/rule-governing-importation-livestock-and-poultry-rule-98074</a>.

FOR COPIES: Kristin Haas, Vermont Agency of Agriculture, Food & Markets, 116 State Street, Montpelier, VT 05620, Tel: 802-522-7326 Email: AGR.FSCPRule@vermont.gov.

\_\_\_\_\_\_

Private Nonmedical Institution Rules Simplification.

Vermont Proposed Rule: 24P005

AGENCY: Agency of Human Services

CONCISE SUMMARY: These rules strike all existing Division of Rate Setting rules and replace them in the Agency of Human Services' Health Care Administrative Rules. Some material, particularly language regarding which costs are allowable and how the Division applies various bonuses or penalties, is moved into a new manual. The remaining material primarily sets out the administrative process for applying for, receiving, and appealing per diem rates set by the Division.

FOR FURTHER INFORMATION, CONTACT: James LaRock, Department of Vermont Health Access, NOB 1 South, 280 State Drive, Waterbury, VT 05671 Tel: 802-241-0251 Fax: 802-241-0260 Email: james.larock@vermont.gov URL:https://humanservices.vermont.gov/rules-policies/health-care-rules/health-care-administrative-rules-hcar.

Nursing Home Reimbursement Rule Simplification.

Vermont Proposed Rule: 24P006

**AGENCY: Agency of Human Services** 

CONCISE SUMMARY: These rules strike all existing Division of Rate Setting rules and replace them in the Agency of Human Services' Health Care Administrative Rules. Some material, particularly language regarding which costs are allowable and how the Division applies various bonuses or penalties, is moved into a new manual. The remaining material primarily sets out the administrative process for applying for, receiving, and appealing per diem rates set by the Division.

FOR FURTHER INFORMATION, CONTACT: James LaRock, Department of Vermont Health Access, NOB 1 South, 280 State Drive, Waterbury, VT 05671 Tel: 802-241-0251 Fax: 802-241-0260 Email: james.larock@vermont.gov URL: https://humanservices.vermont.gov/rules-policies/health-care-rules/health-care-administrative-rules-hcar.

FOR COPIES: Jaime Mooney, Department of Vermont Health Access NOB 1 South, 280 State Drive, Waterbury, VT 05671 Tel: 802-798-2144 Fax: 802-241-0260 Email: jaime.mooney@vermont.gov.

General Assistance Temporary Housing Assistance.

Vermont Proposed Rule: 24P007

AGENCY: Agency of Human Services, Department for Children and Families

CONCISE SUMMARY: The proposed rule contains four amendments to the General Assistance program rules: (1) language was added to rule 2650 authorizing DCF to withhold payments to hotels/motels in violation of

lodging licensing rules;(2) the rule expands categorical eligibility for 28 days of housing under rule 2652.3 to include families with children who are 19 years old or younger; (3) the rule updates the basic needs standard chart in rule 2652.4 to align with the current Reach Up basic needs dollar amounts; and (4) the methodology for calculating the 30% income contribution in rule 2652.4 was changed from using the least expensive daily motel rate to either the current daily rate at the motel in which the temporary housing applicant is staying or if the applicant is not currently housed in a motel, the average daily rate.

FOR FURTHER INFORMATION, CONTACT: Heidi Moreau, Agency of Human Services, Department for Children and Families 280 State Drive, NOB 1 North, Waterbury, VT 05671 Tel: 802-595-9639 Email: <a href="mailto:heidi.moreau@vermont.gov">heidi.moreau@vermont.gov</a> URL: <a href="https://dcf.vermont.gov/esd/laws-rules/current">https://dcf.vermont.gov/esd/laws-rules/current</a>.

FOR COPIES: Amanda Beliveau, Agency of Human Services, Department for Children and Families 280 State Drive, HC 1 South, Waterbury, VT 05671 Tel: 802-241-0641 Email: <a href="mailto:amanda.beliveau@vermont.gov">amanda.beliveau@vermont.gov</a>.

\_\_\_\_\_